UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Ή	ark	On	e)

✓ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.
 For the quarterly period ended December 28, 2007.

OR

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the transition period from ______ to _____

Commission File Number (0-21767)

ViaSat, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 33-0174996 (I.R.S. Employer Identification No.)

6155 El Camino Real Carlsbad, California 92009 (760) 476-2200 (Address of principal executive offices and telephone number)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer, or a smaller reporting company. See definition of "accelerated filer" and "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large Accelerated filer o Accelerated filer Image Accelerated filer o Smaller reporting company o

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

The number of shares outstanding of the registrant's Common Stock, \$0.0001 par value, as of February 1, 2008 was 30,465,339.

VIASAT, INC.

INDEX

	Page
PART I — FINANCIAL INFORMATION	
Item 1. Financial Statements (Unaudited)	
Condensed Consolidated Balance Sheets at December 28, 2007 and March 30, 2007	3
Condensed Consolidated Statements of Operations for the three and nine months ended December 28, 2007 and December 29, 2006	4
Condensed Consolidated Statements of Cash Flows for the nine months ended December 28, 2007 and December 29, 2006	5
Condensed Consolidated Statement of Stockholders' Equity for the nine months ended December 28, 2007	6
Notes to Condensed Consolidated Financial Statements	7
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3. Quantitative and Qualitative Disclosures About Market Risk	36
Item 4. Controls and Procedures	37
PART II — OTHER INFORMATION	
<u>Item 1. Legal Proceedings</u>	37
Item 1A. Risk Factors	37
<u>Item 4. Submission of Matters to a Vote of Security Holders</u>	39
<u>Item 5. Other Information</u>	39
<u>Item 6. Exhibits</u>	39
<u>Signatures</u>	40
EXHIBIT 10.1	
EXHIBIT 10.2	
EXHIBIT 10.3	
EXHIBIT 31.1	
EXHIBIT 31.2	
EXHIBIT 32.1	
2	

PART I — FINANCIAL INFORMATION

Item 1. Financial Statements

VIASAT, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED) (In thousands)

	As of December 28, 2007	As of March 30, 2007
Assets		
Current assets:		
Cash and cash equivalents	\$ 121,073	\$ 103,345
Short-term investments	11,953	47
Accounts receivable, net	147,356	139,789
Inventories	56,371	46,034
Deferred income taxes	15,821	9,721
Prepaid expenses and other current assets	20,907	9,218
Total current assets	373,481	308,154
Goodwill	67,210	65,988
Other intangible assets, net	28,726	33,601
Property and equipment, net	54,293	51,463
Other assets	19,744	24,733
Total assets	\$ 543,454	\$483,939
Liabilities and Stockholders' Equity		
Current liabilities:		
Accounts payable	\$ 52,853	\$ 43,516
Accrued liabilities	79,510	62,470
Payables to former stockholders of acquired businesses	1,101	14,762
Total current liabilities	133,464	120,748
Other liabilities	17,364	13,273
Total liabilities	150,828	134,021
Commitments and contingencies (Note 8)		
Minority interest in consolidated subsidiary	2,234	1,123
Stockholders' equity:		
Common stock	3	3
Paid in capital	252,618	232,693
Retained earnings	138,618	115,969
Common stock held in treasury	(1,034)	_
Accumulated other comprehensive income	187	130
Total stockholders' equity	390,392	348,795
Total liabilities and stockholders' equity	\$ 543,454	\$483,939

See accompanying notes to condensed consolidated financial statements.

VIASAT, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED)

(In thousands, except per share amounts)

		Three months ended			Nine months ended				
		mber 28, 2007	De	cember 29, 2006	De	cember 28, 2007	De	ecember 29, 2006	
Revenues	\$ 1	152,053	\$	124,336	\$	427,240	\$	384,538	
Operating expenses:									
Cost of revenues	1	105,842		90,383		306,751		285,942	
Selling, general and administrative		20,920		17,692		59,074		50,326	
Independent research and development		8,405		5,557		24,215		15,181	
Amortization of intangible assets		2,389		2,521		7,173		7,202	
Income from operations		14,497		8,183		30,027		25,887	
Other income (expense):									
Interest income		1,578		553		4,462		1,302	
Interest expense		(269)		(92)		(606)		(383)	
Income before income taxes		15,806		8,644		33,883		26,806	
Provision (benefit) for income taxes		4,803		(1,095)		9,863		5,076	
Minority interest in net earnings of subsidiary, net of tax		778		49		1,029		140	
Net income	\$	10,225	\$	9,690	\$	22,991	\$	21,590	
Basic net income per share	\$.34	\$.34	\$.76	\$.76	
Diluted net income per share	\$.32	\$.31	\$.71	\$.71	
Shares used in basic net income per share computation		30,338		28,687		30,164		28,352	
Shares used in diluted net income per share computation		32,458		30,773		32,309		30,422	

See accompanying notes to condensed consolidated financial statements.

VIASAT, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (In thousands)

	Nine months ended			i	
		nber 28, 007	December 29, 2006		
Cash flows from operating activities:		<u>007</u>		2000	
Net income	\$	22,991	\$	21,590	
Adjustments to reconcile net income to net cash provided by operating activities:					
Depreciation		11,660		10,457	
Amortization of intangible assets and capitalized software		9,053		9,702	
Deferred income taxes		1,440		(3,593)	
Incremental tax benefits from stock-based compensation		(934)		(1,202)	
Non-cash stock-based compensation		5,550		3,603	
Other non-cash adjustments		2,367		891	
Increase (decrease) in cash resulting from changes in operating assets and liabilities, net of the effects of acquisitions:					
Accounts receivable, net		(7,417)		(6,252)	
Inventories		(9,772)		445	
Other assets		(9,888)		(5,853)	
Accounts payable		7,491		(7,409)	
Accrued liabilities		12,085		18,439	
Other liabilities		1,461		1,518	
Net cash provided by operating activities		46,087		42,336	
Cash flows from investing activities:					
Payments related to acquisitions of businesses, net of cash acquired		(9,826)		(281)	
Purchases of short-term investments held-to-maturity	(11,835)		_	
Purchases of property and equipment	(13,584)		(12,062)	
Net cash used in investing activities	(35,245)		(12,343)	
Cash flows from financing activities:					
Proceeds from issuance of common stock, net of issuance costs		6,763		8,509	
Purchase of common stock in treasury		(1,034)		_	
Incremental tax benefits from stock-based compensation		934		1,202	
Net cash provided by financing activities		6,663		9,711	
Effect of exchange rate changes on cash		223		192	
Net increase in cash and cash equivalents		17,728		39,896	
Cash and cash equivalents at beginning of period	10	03,345		36,723	
Cash and cash equivalents at end of period	\$ 13	21,073	\$	76,619	
Non-cash investing and financing activities:	<u></u>				
Issuance of stock in connection with acquisition (see Note 12)	\$	452	\$	16,350	
Issuance of payables to former stockholders of acquired businesses (see Note 1)	\$	800	\$	9,000	
Issuance of stock in satisfaction of a payable to former stockholders of an acquired business (see Note 1)	\$	5,631	\$	_	

See accompanying notes to condensed consolidated financial statements.

VIASAT, INC. CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (UNAUDITED)

(In thousands, except share data)

	Common S	Stock	Paid in Retained		In Treasury O		In Treasury		ımulated Other orehensive		Comprehensive
	Shares	Amount	Capital	Earnings	Shares	Amount		ne (Loss)	<u>Total</u>	Income (Loss)	
Balance at March 30, 2007	29,733,396	\$ 3	\$232,693	\$ 115,969	_	\$ —	\$	130	\$348,795		
Cumulative effect of adopting FIN 48	_	_	_	(342)	_	_		_	(342)		
Exercise of stock options	367,592	_	5,494	<u> </u>	_	_		_	5,494		
Tax benefit from exercise of stock options and release of restricted stock unit (RSU) awards	,		1,529						1,529		
Issuance of stock under Employee Stock	_	_	1,323	_	_				1,323		
Purchase Plan	50,211	_	1,269	_	_	_		_	1,269		
Stock-based compensation expense	_	_	5,550	_	_	_		_	5,550		
Value of stock issued in connection with acquisition of a business	14,424		452						452		
Value of stock issued as additional consideration in connection with acquisition of a business, net of											
issuance costs	170,763 94,165	_	5,631	_	_				5,631		
RSU awards vesting Purchase of treasury shares pursuant to vesting of certain	94,105	_	_	_	_	_		_	_		
RSU agreements	_	_	_	_	(33,238)	(1,034)		_	(1,034)		
Net income Hedging transactions,	_	_	_	22,991	_	-		— (11)	22,991	\$ 22,991	
net of tax Foreign currency	_	_	_	_	_	_		(11)	(11)	(11)	
translation, net of tax	-	_	_	-	_	_		68	68	<u>68</u>	
Comprehensive income										\$ 23,048	
Balance at December 28, 2007	30,430,551	\$ 3	\$252,618	\$ 138,618	(33,238)	\$(1,034)	\$	187	\$390,392		

See accompanying notes to condensed consolidated financial statements.

Note 1 — Basis of Presentation

The accompanying condensed consolidated balance sheet at December 28, 2007, the condensed consolidated statements of operations for the three and nine months ended December 28, 2007 and December 29, 2006, the condensed consolidated statements of cash flows for the nine months ended December 28, 2007 and December 29, 2006, and the condensed consolidated statement of stockholders' equity for the nine months ended December 28, 2007 have been prepared by the management of ViaSat, Inc. (the "Company"), and have not been audited. These financial statements have been prepared on the same basis as the audited consolidated financial statements for the year ended March 30, 2007 and, in the opinion of management, include all adjustments (consisting only of normal recurring adjustments) necessary for a fair statement of the financial position, results of operations and cash flows for all periods presented. These financial statements should be read in conjunction with the financial statements and notes thereto for the year ended March 30, 2007 included in the Company's 2007 Annual Report on Form 10-K. Interim operating results are not necessarily indicative of operating results for the full year. The year-end condensed balance sheet data were derived from audited financial statements, but do not include all disclosures required by accounting principles generally accepted in the United States of America.

The Company's consolidated financial statements include the assets, liabilities and results of operations of TrellisWare Technologies, Inc., a majority owned subsidiary of the Company. All significant intercompany amounts have been eliminated.

The Company's fiscal year is the 52 or 53 weeks ending on the Friday closest to March 31 of the specified year. For example, references to fiscal year 2008 refer to the fiscal year ending on March 28, 2008. The Company's quarters for fiscal year 2008 end on June 29, 2007, September 28, 2007, December 28, 2007 and March 28, 2008.

During fiscal year 2007, the Company completed the acquisitions of Enerdyne Technologies, Inc. (Enerdyne) and Intelligent Compression Technologies, Inc. (ICT). During the Company's second quarter of fiscal year 2008, the Company completed the acquisition of JAST, S.A., a Switzerland based, privately-held company (see Note 12). The acquisitions were accounted for as purchases and accordingly, the condensed consolidated financial statements include the operating results of Enerdyne, ICT and JAST from the dates of acquisition.

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and reported amounts of revenues and expenses during the reporting period. Estimates have been prepared on the basis of the most current and best available information and actual results could differ from those estimates. Significant estimates made by management include revenue recognition, stock-based compensation, self-insurance reserves, allowance for doubtful accounts, warranty accrual, valuation of goodwill and other intangible assets, valuation of derivatives, long-lived assets and valuation allowance on deferred tax assets.

Marketable Securities

The Company accounts for marketable securities in accordance with Statement of Financial Accounting Standards (SFAS) No. 115 (SFAS 115), "Accounting for Certain Investments in Debt and Equity Securities." The Company determines the appropriate classification of all marketable securities as held-to-maturity, available-for-sale or trading at the time of purchase and re-evaluates such classification as of each balance sheet date. At December 28, 2007, marketable securities consisted primarily of commercial paper with original maturities greater than 90 days at the date of purchase but less than one year. At December 28, 2007 and March 30, 2007, all marketable securities were classified as held-to-maturity and recorded at amortized cost as the Company had the intent and ability to hold the securities to maturity. The amortized cost of debt securities is adjusted for amortization of premiums and accretion of discounts from the date of purchase to maturity. Such amortization is included in interest income (expense) as an addition to or deduction from the coupon interest earned on the investments. The amortized cost of the Company's marketable securities approximated fair value at December 28, 2007 and March 30, 2007.

The Company regularly monitors and evaluates the realizable value of its marketable securities. When assessing marketable securities for other-than-temporary declines in value, the Company considers factors including: how significant the decline in value is as a percentage of the original cost, how long the market value of the investment has been less than its original cost, the performance of the investee's stock price in relation to the stock price of its competitors within the industry, expected market volatility and the

market in general, any news or financial information that has been released specific to the investee and the outlook for the overall industry in which the investee operates. If events and circumstances indicate that a decline in the value of these assets has occurred and is other-than-temporary, the Company records a charge to interest income (expense). No such charges were incurred in the nine months ended December 28, 2007.

Derivatives

The Company enters into foreign currency forward and option contracts to hedge certain forecasted foreign currency transactions. Gains and losses arising from foreign currency forward and option contracts not designated as hedging instruments are recorded in interest income (expense) as gains (losses) on derivative instruments. Gains and losses arising from the effective portion of foreign currency forward and option contracts that are designated as cash-flow hedging instruments are recorded in accumulated other comprehensive income (loss) as unrealized gains (losses) on derivative instruments until the underlying transaction affects the Company's earnings at which time they are then recorded in the same income statement line as the underlying transaction.

Payables to Former Stockholders of Acquired Businesses

On May 23, 2006, in connection with the Company's Efficient Channel Coding, Inc. (ECC) acquisition, the Company agreed under the terms of the ECC acquisition agreement to pay the maximum additional consideration amount to the former ECC stockholders in the amount of \$9.0 million which was accrued as of March 30, 2007. The \$9.0 million was payable in cash or stock, at the Company's option, in May 2007. Accordingly, on May 30, 2007, the Company paid approximately \$9.0 million of additional cash consideration to the former stockholders of ECC. The additional purchase price consideration of \$9.0 million was recorded as additional goodwill in the Satellite Networks product group in the commercial segment in the first quarter of fiscal year 2007.

As of March 30, 2007, in connection with the Company's Enerdyne acquisition and under the terms of the Enerdyne acquisition agreement, the Company owed an additional consideration amount to the former Enerdyne stockholders in the amount of \$5.9 million which was accrued and recorded as additional goodwill in the government segment as of March 30, 2007. The \$5.9 million was payable in cash and stock in accordance with certain terms of the agreement, in May 2007. Accordingly, on May 3, 2007, the Company paid \$5.9 million of additional consideration to the former stockholders of Enerdyne, which was comprised of 170,763 shares of common stock and \$260,000 in cash.

On August 2, 2007, in connection with the terms of the Company's JAST acquisition, the Company has an obligation to pay the remaining portion of the initial purchase price of approximately \$800,000 on the first anniversary of the closing date of which \$483,000 will be paid in cash and \$317,000 will be paid in stock or cash, at the Company's election. The \$800,000 payable was accrued in current liabilities as of December 28, 2007.

Land Held-for-Sale

In January 2006, the Company purchased approximately 10 acres of land adjacent to a leased facility for approximately \$3.1 million. During the first quarter of fiscal year 2007, the Company signed a property listing agreement with the intention to sell the property. As of December 28, 2007, the Company reported the property in accordance with SFAS No. 144 (SFAS 144), "Accounting for the Impairment or Disposal of Long-Lived Assets," as an asset held-for-sale at the lower of carrying value or fair value, less estimated costs to sell, which is estimated to be \$3.1 million.

Self-Insurance Liabilities

The Company has a self-insurance plan to retain a portion of the exposure for losses related to employee medical benefits. The Company also has a self-insurance plan for a portion of the exposure for losses related to workers' compensation costs. The self-insured policies provide for both specific and aggregate stop-loss limits. The Company utilizes internal actuarial methods, as well as an independent third-party actuary for the purpose of estimating ultimate costs for a particular policy year. Based on these actuarial methods along with currently available information and insurance industry statistics, the Company recorded self-insurance liabilities as of December 28, 2007 and March 30, 2007 of \$1.1 million and \$883,000, respectively. The Company's estimate, which is subject to inherent variability, is based on average claims experience in the Company's industry and its own experience in terms of frequency and severity of claims, including asserted and unasserted claims incurred but not reported, with no explicit provision for adverse fluctuation from year to year. This variability may lead to ultimate payments being either greater or less than the amounts presented above. Self-insurance liabilities have been classified as current in accordance with the estimated timing of the projected payments.

Secured Borrowings

Occasionally, the Company enters into secured borrowing arrangements in connection with customer financing in order to provide additional sources of funding. As of December 28, 2007 and March 30, 2007, the Company had one secured borrowing arrangement, under which the Company pledged a note receivable from a customer to serve as collateral for the obligation under the borrowing arrangement. The arrangement includes recourse to certain other assets of the Company in the event of customer default on the note receivable. No significant guarantees beyond the recourse provision exist. Payments under the arrangement consist of semi-annual principal payments of \$590,000 plus accrued interest for five years with the first semi-annual payment being interest only. The interest rate resets semi-annually to the current LIBOR rate plus a margin of 2.5%. This secured borrowing arrangement does not qualify as a sale of assets under SFAS No. 140 (SFAS 140), "Accounting for Transfers and Servicing of Financial Assets and Extinguishment of Liabilities," as the Company has continued involvement related to the recourse provision.

During the third quarter of fiscal year 2008, due to a payment default, the Company wrote-down the note receivable by approximately \$5.3 million related to the principal and interest accrued to date. Pursuant to a notes receivable insurance arrangement which provides for the recovery of principal and certain interest amounts of the note, the Company recorded a current asset of approximately \$4.5 million as of December 28, 2007. As of December 28, 2007, the Company had \$5.1 million of the secured borrowing recorded under accrued liabilities with a carrying value approximating the balance of the secured borrowing. As of March 30, 2007, the Company had \$590,000 of the secured borrowing recorded under accrued liabilities and \$4.1 million recorded under other long-term liabilities with a carrying value approximating the balance of the secured borrowing.

Indemnification Provisions

In the ordinary course of business, the Company includes indemnification provisions within certain of its contracts, generally relating to parties with which the Company has commercial relations. Pursuant to these agreements, the Company will indemnify, hold harmless and agree to reimburse the indemnified party for losses suffered or incurred by the indemnified party, including but not limited to intellectual property indemnity. Historically, to date, there have not been any costs incurred in connection with such indemnification clauses. The Company's insurance policies do not necessarily cover the cost of defending indemnification claims or providing indemnification, so if a claim was filed against the Company by any party the Company indemnifies, the Company could incur substantial legal costs and damages. A claim would be accrued when a loss is considered probable and the amount can be reasonably estimated. At December 28, 2007 and March 30, 2007, no such amounts were accrued.

Treasury Stock

During the third quarter of fiscal year 2008, the Company delivered 94,165 shares of common stock based on the vesting terms of certain restricted stock unit agreements. In order for employees to satisfy minimum statutory employee tax withholding requirements related to the delivery of common stock underlying these restricted stock unit agreements, the Company repurchased 33,238 shares of common stock with a total value of \$1.0 million in the first nine months of fiscal year 2008. There was no common stock held in treasury as of March 30, 2007.

Stock-Based Compensation

The Company records compensation expense associated with stock options, restricted stock unit awards and other equity based compensation in accordance with SFAS No. 123 (revised 2004), "Share-Based Payment" (SFAS 123R), which the Company adopted on April 1, 2006. The Company recognizes these compensation costs on a straight-line basis over the requisite service period of the award. The Company recognized \$1.9 million and \$5.6 million of stock-based compensation expense related to the adoption of SFAS 123R for the three and nine months ended December 28, 2007, respectively, and \$1.6 million and \$2.4 million for the three and nine months ended December 29, 2006, respectively.

The Company recorded incremental tax benefits from stock options exercised and restricted stock unit award vesting of \$934,000 and \$1.2 million for the nine months ended December 28, 2007 and December 29, 2006, respectively, which is classified as part of cash flows from financing activities in the condensed consolidated statements of cash flows. At December 28, 2007, the total unrecognized estimated compensation cost net of estimated forfeitures related to unvested stock options and restricted stock units, and the employee stock purchase plan was approximately \$9.4 million, \$6.8 million and \$0, respectively. These costs are expected to be recognized over a weighted average period of 2.7 years, 2.8 years and 0 years, respectively.

Review of Stock Option Grant Procedures

In August 2006, the Company commenced and completed a voluntary internal investigation, assisted by the Company's outside legal counsel, of its historical stock option granting practices, stock option documentation and related accounting during the period from its initial public offering in December 1996 through June 30, 2006. At the conclusion of its investigation, the Company's outside legal counsel and the Company determined that there was no evidence of a pattern of intentionally misdating stock option grants to achieve an accounting result, or that any officer, director, or senior executive at the Company willfully or knowingly engaged in stock options misdating, or had knowledge of others doing so.

During the investigation, the Company identified certain accounting errors associated with stock options granted primarily to certain non-executive new hire employees during the ten-year period from December 1996 to June 30, 2006. Based on the results of the investigation, the Company identified that certain stock options to non-executive new hires had incorrectly been accounted for using an accounting measurement date prior to the date that the new hires commenced employment. The Company concluded, with the concurrence of the Audit Committee, that the financial impact of these errors was not material to its consolidated financial statements for any annual period in which the errors related. In accordance with Accounting Principles Board Opinion No. 28, "Interim Financial Reporting," paragraph 29, the Company recorded a cumulative adjustment to compensation expense in the first quarter of fiscal year 2007 of \$703,000, net of tax, because the effect of the correcting adjustment was not material to the Company's fiscal year 2007 net income. This non-cash compensation expense adjustment will have no impact on future periods. There is no impact on revenue or net cash provided by operating activities as a result of recording the compensation expense adjustment.

Income Taxes

Current income tax expense is the amount of income taxes expected to be payable for the current year. A deferred income tax asset or liability is established for the expected future tax consequences resulting from differences in the financial reporting and tax bases of assets and liabilities and for the expected future tax benefit to be derived from tax credit and loss carryforwards. Deferred tax assets are reduced by a valuation allowance when, in the opinion of management, it is more likely than not that some portion or all of the deferred tax assets will not be realized. Deferred income tax expense (benefit) is the net change during the year in the deferred income tax asset or liability.

Recent Accounting Pronouncements

In September 2006, the FASB issued Statement No. 157 (SFAS 157), "Fair Value Measurements." SFAS 157 defines fair value, establishes a framework and gives guidance regarding the methods used for measuring fair value, and expands disclosures about fair value measurements. SFAS 157 is effective for financial statements issued for fiscal years beginning after November 15, 2007 (fiscal year 2009 for the Company), and interim periods within those fiscal years. The Company is currently assessing the impact SFAS 157 will have on its results of operations and financial position.

In February 2007, the FASB issued Statement No. 159 (SFAS 159), "The Fair Value Option for Financial Assets and Financial Liabilities," which permits entities to choose to measure many financial instruments and certain other items at fair value that are not currently required to be measured at fair value. SFAS 159 will be effective for the Company in fiscal year 2009. The Company is currently evaluating the impact of adopting SFAS 159 on its financial position, cash flows, and results of operations.

In June 2007, the FASB ratified Emerging Issues Task Force (EITF) Issue No. 07-3 (EITF 07-3), "Accounting for Nonrefundable Advance Payments for Goods or Services to Be Used in Future Research and Development Activities." This issue provides that nonrefundable advance payments for goods or services that will be used or rendered for future research and development activities should be deferred and capitalized. Such amounts should be recognized as an expense as the related goods are delivered or the related services are performed. EITF 07-3 is effective for the Company's fiscal year 2009. The adoption of EITF Issue No. 07-3 is not expected to have a material impact on the Company's consolidated financial position, results of operations and cash flows.

In December 2007, the FASB issued SFAS No. 141 (revised 2007), "Business Combinations" (SFAS 141R). The purpose of issuing the statement is to replace current guidance in SFAS 141 to better represent the economic value of a business combination transaction. The changes to be effected with SFAS 141R from the current guidance include, but are not limited to: (1) acquisition costs will be recognized as expenses separately from the acquisition; (2) known contractual contingencies at the time of the acquisition will be considered part of the liabilities acquired measured at their fair value; all other contingencies will be part of the liabilities acquired measured at their fair value only if it is more likely than not that they meet the definition of a liability; (3) contingent consideration

based on the outcome of future events will be recognized and measured at the time of the acquisition; (4) business combinations achieved in stages (step acquisitions) will need to recognize the identifiable assets and liabilities, as well as non-controlling interests, in the acquiree, at the full amounts of their fair values; and (5) a bargain purchase (defined as a business combination in which the total acquisition-date fair value of the identifiable net assets acquired exceeds the fair value of the consideration transferred plus any non-controlling interest in the acquiree) will require that excess to be recognized as a gain attributable to the acquirer. SFAS 141R will be effective for the Company in fiscal year 2010. The Company is currently evaluating the impact of SFAS 141R.

In December 2007, the FASB issued SFAS No. 160 (SFAS 160), "Noncontrolling Interests in Consolidated Financial Statements — an amendment of ARB No. 51." SFAS 160, which changes the accounting and reporting for business acquisitions and non-controlling interests in subsidiaries, was issued to improve the relevance, comparability, and transparency of financial information provided to investors. Moreover, SFAS 160 eliminates the diversity that currently exists in accounting for transactions between an entity and non-controlling interests by requiring they be treated as equity transactions. SFAS 160 will be effective for the Company in fiscal year 2010. The Company is currently evaluating the impact that SFAS 160 will have on its financial statements and disclosures.

Note 2 — Revenue Recognition

A substantial portion of the Company's revenues are derived from long-term contracts requiring development and delivery of complex equipment built to customer specifications. Sales related to long-term contracts are accounted for under the percentage-of-completion method of accounting under the American Institute of Certified Public Accountants' Statement of Position 81-1 (SOP 81-1), "Accounting for Performance of Construction-Type and Certain Production-Type Contracts." Sales and earnings under these contracts are recorded either based on the ratio of actual costs incurred to total estimated costs expected to be incurred related to the contract or as products are shipped under the units-of-delivery method. Anticipated losses on contracts are recognized in full in the period in which losses become probable and estimable. Changes in estimates of profit or loss on contracts are included in earnings on a cumulative basis in the period the estimate is changed. During the three months ended December 28, 2007 and December 29, 2006, the Company recorded losses of approximately \$3.0 million and \$1.1 million, respectively, related to loss contracts. During the nine months ended December 28, 2007 and December 29, 2006, the Company recorded losses of approximately \$6.5 million and \$2.4 million, respectively, related to loss contracts.

The Company also has contracts and purchase orders where revenue is recorded on delivery of products in accordance with SAB 104, "Staff Accounting Bulletin No. 104: Revenue Recognition." In this situation, contracts and customer purchase orders are used to determine the existence of an arrangement. Shipping documents and customer acceptance, when applicable, are used to verify delivery. The Company assesses whether the sales price is fixed or determinable based on the payment terms associated with the transaction and whether the sales price is subject to refund or adjustment, and assesses collectibility based primarily on the creditworthiness of the customer as determined by credit checks and analysis, as well as the customer's payment history.

When a sale involves multiple elements, such as sales of products that include services, the entire fee from the arrangement is allocated to each respective element based on its relative fair value in accordance with EITF 00-21, "Accounting for Multiple Element Revenue Arrangements" and recognized when the applicable revenue recognition criteria for each element are met. The amount of product and service revenue recognized is impacted by the Company's judgments as to whether an arrangement includes multiple elements and, if so, whether sufficient objective and reliable evidence of fair value exists for those elements. Changes to the elements in an arrangement and the Company's ability to establish evidence for those elements could affect the timing of the revenue recognition.

In accordance with EITF 00-10, "Accounting for Shipping and Handling Fees and Costs," the Company records shipping and handling costs billed to customers as a component of revenues, and shipping and handling costs incurred by the Company for inbound and outbound freight are recorded as a component of cost of revenues.

Collections in excess of revenues represent cash collected from customers in advance of revenue recognition.

Contract costs on U.S. government contracts, including indirect costs, are subject to audit and negotiations with U.S. government representatives. These audits have been completed and agreed upon through fiscal year 2002. Contract revenues and accounts receivable are stated at amounts which are expected to be realized upon final settlement.

VIASAT, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

Note 3 — Earnings Per Share

Potential common stock of 2,119,581 and 2,085,677 shares for the three months ended December 28, 2007 and December 29, 2006, respectively, and 2,145,112 and 2,069,757 shares for the nine months ended December 28, 2007 and December 29, 2006, respectively, were included in the calculation of diluted earnings per share. Antidilutive shares excluded from the calculation were 995,948 and 810,974 shares for the three months ended December 28, 2007 and December 29, 2006, respectively and 931,453 and 398,407 shares for the nine months ended December 28, 2007 and December 29, 2006, respectively. Potential common stock includes options granted and restricted stock units awarded under the Company's equity compensation plan which are included in the earnings per share calculations using the treasury stock method and common shares expected to be issued under the Company's employee stock purchase plan.

Note 4 — Composition of Certain Balance Sheet Captions (In thousands)

	<u>December 28, 2007</u>			rch 30, 2007
Accounts receivable, net: Billed	ď	02.711	¢	90 G 4E
Unbilled	\$	92,711 55,844	\$	89,645 51,358
Allowance for doubtful accounts		(1,199)		(1,214)
Anowalice for dodottul accounts	\$	147,356	\$	139,789
	φ	147,550	φ <u></u>	133,703
Inventories:	d.	24 020	ď	10.040
Raw materials Work in process	\$	21,939 8,680	\$	19,840 7,963
Finished goods		25,752		18,231
Fillished goods	\$	56,371	\$	46,034
Prepaid expenses and other current assets:	Ψ	30,371	<u> </u>	40,054
Prepaid expenses	\$	12,064	\$	8,339
Income tax receivable	Ψ	2,978	Ψ	— —
Other		5,865		879
	\$	20,907	\$	9,218
Other intensible access not	Ψ	20,507	Ψ	3,210
Other intangible assets, net: Technology	\$	44,392	\$	43,270
Contracts and relationships	Ф	19,758	Ф	18,766
Non-compete agreement		9,076		8,920
Other intangibles		9,323		9,295
one mangere	<u></u>	82,549		80,251
Less accumulated amortization		(53,823)		(46,650)
200 detailmented universation	\$	28,726	\$	33,601
Property and equipment, net:	Ψ	20,720	Ψ	55,001
Machinery and equipment	\$	48,906	\$	48,439
Computer equipment and software	Ф	41,754	Ф	36,936
Furniture and fixtures		8,220		7,552
Leasehold improvements		12,885		12,983
Land held-for-sale		3,124		3,124
Construction in progress		3,422		2,440
		118,311		111,474
Less accumulated depreciation and amortization		(64,018)		(60,011)
	\$	54,293	\$	51,463
Other assets:	Ť	0 1,200	Ě	02,100
Capitalized software costs, net	\$	1,713	\$	3,576
Deferred income taxes	Ψ	11,298	Ψ	13,328
Other		6,733		7,829
	\$	19,744	\$	24,733
Accrued liabilities:	<u> </u>	13,7 11	Ψ	21,755
Secured borrowings and accrued interest	\$	5,111	\$	590
Current portion of warranty reserve	Ψ	6,012	Ψ	5,007
Accrued vacation		8,474		7,958
Accrued wages and performance compensation		9,866		10,678
Collections in excess of revenues		39,409		28,030
Other		10,638		10,207
	\$	79,510	\$	62,470
Other liabilities:	_ `		— —	
Unrecognized tax position liabilities	\$	6,760	\$	
Accrued warranty	Ψ	5,641	Ψ	4,856
Deferred rent, long-term portion		3,874		3,514
Secured borrowing, long-term portion				4,130
Other		1,089		773
	\$	17,364	\$	13,273
			Ė	

Note 5 — Accounting for Goodwill and Intangible Assets

The Company accounts for its goodwill under SFAS No. 142 (SFAS 142), "Goodwill and Other Intangible Assets." The SFAS 142 goodwill impairment model is a two-step process. First, it requires a comparison of the book value of net assets to the fair value of the reporting units that have goodwill assigned to them. Reporting units within the Company's government and commercial segments have goodwill assigned to them. The Company estimates the fair values of the reporting units using discounted cash flows. The cash flow forecasts are adjusted by an appropriate discount rate. If the fair value is determined to be less than book value, a second step is performed to compute the amount of the impairment. In this process, a fair value for goodwill is estimated, based in part on the fair value of the operations used in the first step, and is compared to its carrying value. The shortfall of the fair value below carrying value represents the amount of goodwill impairment.

The Company will continue to make assessments of impairment on an annual basis in the fourth quarter of its fiscal year or more frequently if specific triggering events occur. In assessing the value of goodwill, the Company must make assumptions regarding estimated future cash flows and other factors to determine the fair value of the reporting units. If these estimates or their related assumptions change in the future, the Company may be required to record impairment charges that would negatively impact operating results.

The intangible assets are amortized using the straight-line method over their estimated useful lives of eight months to ten years. The technology intangible asset has several components with estimated useful lives of five to nine years, the contracts and relationships intangible asset has several components with estimated useful lives of three to ten years, the non-compete agreements have useful lives of three to five years and other amortizable assets have several components with original estimated useful lives of eight months to ten years. The amortization expense was \$2.4 million and \$2.5 million for the three months ended December 28, 2007 and December 29, 2006, respectively, and \$7.2 million for the nine months ended December 28, 2007 and December 29, 2006.

The current and expected amortization expense for each of the following periods is as follows (in thousands):

	Am	ortization
For the nine months ended December 28, 2007	\$	7,173
Expected for the remainder of fiscal year 2008	\$	2,389
Expected for fiscal year 2009		9,020
Expected for fiscal year 2010		5,787
Expected for fiscal year 2011		5,024
Expected for fiscal year 2012		3,798
Thereafter		2,708
	\$	28,726

Note 6 — Notes Payable and Line of Credit

On January 31, 2005, the Company entered into a three-year, \$60 million revolving credit facility (the "Facility") in the form of a Second Amended and Restated Revolving Loan Agreement. On January 25, 2008, the Company amended the Second Amended and Restated Revolving Loan Agreement extending the Facility's current terms and conditions to April 30, 2008.

Borrowings under the Facility are permitted up to a maximum amount of \$60 million, including up to \$15 million of letters of credit. Borrowings under the Facility bear interest, at the Company's option, at either the lender's prime rate or at LIBOR (London Interbank Offered Rate) plus, in each case, an applicable margin based on the ratio of the Company's total funded debt to EBITDA (income from operations plus depreciation and amortization). The Facility is collateralized by substantially all of the Company's personal property assets. At December 28, 2007, the Company had approximately \$8.2 million outstanding under standby letters of credit leaving borrowing availability under the Facility of \$51.8 million.

The Facility contains financial covenants that set a minimum EBITDA limit for the twelve-month period ending on the last day of any fiscal quarter at \$30.0 million, a minimum tangible net worth as of the last day of any fiscal quarter at \$135.0 million and a minimum quick ratio (sum of cash and cash equivalents, accounts receivable and marketable securities, divided by current liabilities) as of the last day of any fiscal quarter at 1.50 to 1.00. The Company was in compliance with its loan covenants at December 28, 2007.

Note 7 — Product Warranty

The Company provides limited warranties on most of its products for periods of up to five years. The Company records a liability for its warranty obligations when products are shipped based upon an estimate of expected warranty costs. Amounts expected to be incurred within twelve months are classified as a current liability. For mature products, the warranty cost estimates are based on historical experience with the particular product. For newer products that do not have a history of warranty costs, the Company bases its estimates on its experience with the technology involved and the types of failures that may occur. It is possible that the Company's underlying assumptions will not reflect the actual experience and in that case, future adjustments will be made to the recorded warranty obligation. The following table reflects the change in the Company's warranty accrual during the nine months ended December 28, 2007 and December 29, 2006 (in thousands).

		For the nine months ended				
	Decem	ber 28, 2007	December 29, 2006			
Balance, beginning of period	\$	9,863	\$	8,369		
Change in liability for warranties issued in period		5,900		5,207		
Settlements made (in cash or in kind) during the period		(4,110)		(3,408)		
Balance, end of period	\$	11,653	\$	10,168		

Note 8 - Commitments, Contingencies

The Company is a party to various claims and legal actions arising in the normal course of business. Although the ultimate outcome of such matters is not presently determinable, the Company believes that the resolution of all such matters, net of amounts accrued, will not have a material adverse effect on its financial position or liquidity; however, there can be no assurance that the ultimate resolution of these matters will not have a material impact on its results of operations in any period.

Note 9 — Derivatives

During the three months ended December 28, 2007, the Company settled certain foreign exchange contracts recognizing a gain of approximately \$205,000, recorded as cost of revenues based on the nature of the underlying transaction. During the nine months ended December 28, 2007, the Company entered into a foreign currency exchange contract intended to reduce the foreign currency risk for amounts payable to vendors in Euros which have a maturity of less than six months. The fair value of the outstanding foreign currency contract was approximately \$11,000 and was recorded as a liability as of December 28, 2007. The Company had \$1.4 million and \$0 of notional value of foreign currency forward contracts outstanding as of December 28, 2007 and March 30, 2007, respectively. During the three months ended December 29, 2006, the Company settled certain foreign exchange contracts recognizing a gain of \$7,000 recorded as cost of revenues based on the nature of the underlying transaction. During the nine months ended December 29, 2006, the Company settled certain foreign contracts recognizing a loss of \$130,000 recorded as a cost of revenues based on the nature of the underlying transaction.

Note 10 — Income Taxes

The effective income tax rate for the three and nine months ended December 28, 2007 was 30.4% and 29.1%, respectively, compared to the 18.2% annual effective tax rate for the fiscal year ended March 30, 2007, reflecting the December 31, 2007 expiration of the federal research and development tax credit. The estimated tax rate is different from the expected statutory rate due primarily to research and development tax credits and the manufacturing deduction.

The Company's estimated effective tax rate of 28.4% for fiscal year 2008 reflects the expiration of the federal research and development tax credit at December 31, 2007. If the federal research and development tax credit is reinstated, the Company will have a lower effective tax rate. In the event the federal research and development tax credit is reinstated, the amount of the reduction in the Company's tax rate will depend on the effective date and terms of the reinstatement, as well as the amount of eligible research and development expenses in the reinstated period.

In 2006, the FASB issued Interpretation No. 48 (FIN 48), "Accounting for Uncertainty in Income Taxes-an interpretation of FASB Statement No. 109," which clarifies the accounting for uncertainty in tax positions. The Company adopted FIN 48 on March 31, 2007 and recognized a cumulative-effect adjustment of \$342,000, decreasing beginning retained earnings. The Company also recorded unrecognized tax position liabilities of approximately \$6.8 million that were previously offset against deferred tax assets.

As of the beginning of fiscal year 2008, the total gross amount of unrecognized tax benefits was \$28.8 million. Of that amount, \$19.9 million, if recognized, would affect the Company's effective tax rate. For the three and nine months ended December 28, 2007, the Company's gross unrecognized tax benefits increased by \$1.2 million and \$2.6 million, respectively. The Company's policy is to recognize interest and penalties related to unrecognized tax benefits as a component of income tax expense. Prior to the adoption of FIN 48, the Company had recognized interest expense related to income tax matters as a component of interest expense. The gross amount of interest and penalties accrued as of the beginning of fiscal year 2008 was \$892,000.

The Company is subject to periodic audits by domestic and foreign tax authorities. The Internal Revenue Service ("IRS") examination of the Company's U.S. federal tax returns for fiscal years 2001-2004 was completed in the fourth quarter of fiscal year 2006 and agreement was reached with the IRS on the proposed adjustments. There was no material impact on income taxes or interest resulting from these audits and the Company considers those fiscal years to be effectively settled under FIN 48. By statute, the Company's U.S. federal returns are subject to examination by the IRS for fiscal years 2005 through 2007. Additionally, tax credit carryovers that were generated in prior years and utilized in these years may also be subject to examination by the IRS. In July 2007, the Company was notified by the IRS of its intention to examine the Company's fiscal year 2006 federal income tax return. With few exceptions, the fiscal years 2003 to 2007 remain open to examination by state and foreign taxing jurisdictions. The Company believes that it has appropriate support for the income tax positions taken and to be taken on its tax returns and that its accruals for tax liabilities are adequate for all open years based on an assessment of many factors, including past experience and interpretations of tax law applied to the facts of each matter. In the next twelve months it is reasonably possible that the amount of unrecognized tax benefits will decrease by \$972,000 as a result of the expiration of the statute of limitations for previously filed tax returns.

Note 11 — Segment Information

The Company's commercial and government segments are primarily distinguished by the type of customer and the related contractual requirements. The more regulated government environment is subject to unique contractual requirements and possesses economic characteristics which differ from the commercial segment. Therefore, the Company is organized primarily on the basis of products with commercial and government (defense) communication applications. These product groups are distinguished from one another based upon their underlying technologies. During the third quarter of fiscal year 2008, the Company made management and organization structure changes due to a shift in product marketing and development strategies and consequently realigned the way management organizes and evaluates financial information internally for making operating decisions and assessing performance in the Satellite Networks and Government reportable segments. Reporting segments are determined consistent with the way management currently organizes and evaluates financial information internally for making operating decisions and assessing performance. The following segment information, including prior periods, recasts this new organizational and reporting structure:

	Three mon	ths ended December 29.	Nine mont	nonths ended December 29,		
(in thousands)	2007	2006	2007	2006		
Revenues						
Government	\$ 84,943	\$ 69,045	\$ 235,425	\$ 207,340		
Commercial						
Satellite Networks	52,234	46,821	152,420	147,948		
Antenna Systems	14,876	8,470	39,395	29,250		
	67,110	55,291	191,815	177,198		
Elimination of intersegment revenues	_	_	_	_		
Total revenues	152,053	124,336	427,240	384,538		
Operating profits (losses)						
Government	15,610	8,786	33,656	33,691		
Commercial						
Satellite Networks	(85)	1,980	771	(54)		
Antenna Systems	1,345	(41)	3,013	(607)		
	1,260	1,939	3,784	(661)		
Elimination of intersegment operating profits	56	_	56	· —		
Segment operating profit before corporate and amortization	16,926	10,725	37,496	33,030		
Corporate	(40)	(21)	(296)	59		
Amortization of intangible assets	(2,389)	(2,521)	(7,173)	(7,202)		
Income from operations	\$ 14,497	\$ 8,183	\$ 30,027	\$ 25,887		

Amortization of intangibles by segment for the three and nine months ended December 28, 2007 and December 29, 2006 is as follows:

	Three months ended					Nine months ended				
(in thousands)		December 28, 2007		December 29, 2006		December 28, 2007		ember 29, 2006		
Government	\$	272	\$	703	\$	815	\$	1,497		
Commercial										
Satellite Networks		1,799		1,654		5,609		5,213		
Antenna Systems		318		164		749		492		
Total amortization of intangibles	\$	2,389	\$	2,521	\$	7,173	\$	7,202		

Assets identifiable to segments include: accounts receivable, unbilled accounts receivable, inventory, intangible assets and goodwill. Segment assets as of December 28, 2007 and March 30, 2007 are as follows:

(in thousands)	December 28, 2007		March 30, 2007	
Segment assets				
Government	\$	133,398	\$	114,625
Commercial				
Satellite Networks		141,739		148,526
Antenna Systems		24,707		22,704
		166,446		171,230
Corporate assets		243,610		198,084
Total	\$	543,454	\$	483,939

The management and organization structure changes in the third quarter of fiscal year 2008 resulted in reclassifications of approximately \$5.9 million of goodwill for December 28, 2007 and March 30, 2007, and \$299,000 and \$345,000 of net intangible assets for December 28, 2007 and March 30, 2007, respectively, from the Satellite Networks segment to the Government segment. Net intangible assets and goodwill included in segment assets as of December 28, 2007 and March 30, 2007 are as follows:

	Net intangi	Net intangible assets		Goodwill		
(in thousands)	December 28, 2007	March 30, 2007	December 28, 2007	March 30, 2007		
Government	\$ 4,152	\$ 4,967	\$ 21,993	\$ 21,993		
Commercial						
Satellite Networks	21,666	27,275	40,367	40,367		
Antenna Systems	2,908	1,359	4,850	3,628		
Total	\$ 28,726	\$ 33,601	\$ 67,210	\$ 65,988		

Revenue information by geographic area for the three and nine month periods ended December 28, 2007 and December 29, 2006 is as follows:

	Three m	onths ended	Nine n	onths ended
(in thousands)	December 28, 2007	December 29, 2006	December 28, 2007	December 29, 2006
United States	\$ 125,164	\$ 102,885	\$ 349,196	\$ 326,554
Asia Pacific	5,548	3,316	21,074	14,818
Europe/Africa	11,197	8,919	30,184	27,047
North America other than United States	9,067	3,568	22,901	9,179
Latin America	1,077	5,648	3,885	6,940
	\$ 152,053	\$ 124,336	\$ 427,240	\$ 384,538

The Company distinguishes revenues from external customers by geographic areas based on customer location.

The net book value of long-lived assets located outside the United States was \$399,000 at December 28, 2007 and \$313,000 at March 30, 2007.

Note 12 — Acquisition

On August 2, 2007, the Company completed the acquisition of all of the outstanding capital stock of JAST, S.A., a Switzerland based, privately-held developer of microwave circuits and antennas for terrestrial and satellite applications, specializing in small, low-profile antennas for mobile satellite communications. The initial purchase price of approximately \$2.1 million was comprised

primarily of \$452,000 related to the fair value of 14,424 shares of the Company's common stock issued at the closing date, \$748,000 in cash consideration, the issuance of an \$800,000 payable, and approximately \$125,000 in direct acquisition costs. The \$748,000 in cash consideration paid to the former JAST stockholders plus approximately \$125,000 in direct acquisition costs less cash acquired of \$22,000 resulted in a net cash outlay of approximately \$851,000 as of September 28, 2007. The remaining \$800,000 is payable on the first anniversary of the closing date, of which \$483,000 will be paid in cash and \$317,000 will be paid in stock or cash, at the Company's election. Under the terms of the purchase agreement, up to an additional \$4.5 million in consideration is payable in stock and/or cash, at the Company's option, based on JAST achieving certain earnings performance and technology development targets during the two years following closing. No portion of this additional consideration is guaranteed. The additional consideration, if earned, is payable after JAST achieves specified earnings performance and technology development targets and will be recorded as additional purchase price.

The preliminary allocation of purchase price of the acquired assets and assumed liabilities based on the estimated fair values is as follows:

	Augu	st 2, 2007
	(In th	nousands)
Current assets	\$	169
Identifiable intangible assets		2,298
Goodwill		1,222
Total assets acquired		3,689
Liabilities assumed		(1,564)
Total purchase price	\$	2,125

Amounts assigned to other intangible assets are being amortized on a straight-line basis over their estimated useful lives ranging from two to five years and are as follows:

	(In th	nousands)
Technology (3 year weighted average life)	\$	1,122
Customer relationships (5 year weighted average life)		992
Non-compete agreements (5 year weighted average life)		156
Backlog (2 year weighted average life)		28
Total identifiable intangible assets	\$	2,298

The acquisition of JAST is beneficial to the Company because it adds complementary technologies and provides additional business opportunities, namely microwave circuits and antennas for terrestrial and satellite applications and small, low-profile antennas for mobile satellite communications. The benefit of these products can be offered to many of the Company's consumer, enterprise or government customers. These benefits and additional opportunities were among the factors that contributed to a purchase price resulting in the recognition of goodwill which has been recorded within Antenna Systems product group in the commercial segment. The intangible assets and goodwill recognized will not be deductible for federal income tax purposes. The purchase price is preliminary due to resolution of certain tax and contractual matters.

The condensed consolidated financial statements include the operating results of JAST from the date of acquisition in the Company's Antenna Systems product group in the commercial segment. Pro forma results of operations have not been presented because the effect of the acquisition was insignificant to the financial statements for all periods presented.

Note 13 - Subsequent Event

In January 2008, the Company entered into several agreements with Space Systems/Loral (SS/L), Loral Space & Communications (Loral) and Telesat Canada (Telesat) related to the Company's high capacity satellite system. Under the satellite construction contract with SS/L, the Company will purchase a Ka-band satellite (ViaSat-1) for approximately \$209.1 million, subject to purchase price adjustments based on satellite performance. The Company does not believe the purchase price paid by ViaSat to SS/L for the ViaSat-1 satellite will materially change. In addition, the Company entered into a beam sharing agreement with Loral ("Beam Sharing Agreement"), whereby Loral is responsible for contributing 15% of the total costs (estimated at approximately \$60 million) associated with the ViaSat-1 satellite project. As part of this arrangement, Loral executed a separate contract with SS/L whereby Loral is purchasing the Canadian beams on the ViaSat-1 satellite (Loral Beams) for approximately \$36.9 million (15% of the total satellite cost of \$246.0 million). In addition, Loral remains responsible under the Beam Sharing Agreement to reimburse ViaSat for costs associated with

VIASAT, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

launch, launch and in-orbit insurance, and operating the satellite. The reimbursed costs to ViaSat from Loral for launch and launch and the first year of in-orbit insurance are expected to be approximately \$23.1 million. Further, under the terms of the Beam Sharing Agreement, ViaSat is appointed the sole and exclusive supplier of end-user broadband terminals and hub equipment for broadband service operated over the Loral Beams. The Company also entered into an agreement with Telesat, whereby Telesat has agreed to transfer certain orbital slot license rights to the Company for ViaSat-1 satellite operation.

Michael Targoff, a director of the Company since February 2003, currently serves as the Chief Executive Officer and the Vice Chairman of the board of directors of Loral Space & Communications, Inc., the parent of Space Systems/Loral, Inc. John Stenbit, a director of the Company since August 2004, also currently serves on the board of directors of Loral Space & Communications, Inc. The purchase of the ViaSat-1 satellite by the Company from SS/L was approved by the disinterested members of the Company's Board of Directors.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following information should be read in conjunction with the condensed consolidated financial statements and the notes thereto included in Item 1 of this Quarterly Report and the audited consolidated financial statements and notes thereto and Management's Discussion and Analysis of Financial Condition and Results of Operations included in our Annual Report on Form 10-K for the fiscal year ended March 30, 2007, filed with the Securities and Exchange Commission.

Except for the historical information contained herein, the following discussion contains forward-looking statements that are subject to risks and uncertainties. Actual results may differ substantially from those referred to herein due to a number of factors, including but not limited to risks described in the section entitled "Risk Factors" and elsewhere in this Quarterly Report.

General

We are a leading provider of advanced digital satellite communications and other wireless and secure networking and signal processing equipment and services to the government and commercial markets. Our internal growth to date has historically been driven largely by our success in meeting the need for advanced communications products for our government and commercial customers. By developing cost-effective communications solutions incorporating our advanced technologies, we have continued to grow the markets for our products and services. Our goal is to leverage our advanced technology and capabilities to capture a considerable share of the global satellite communications equipment and services segment for both government and commercial customers. ViaSat was incorporated in 1986 and completed its initial public offering in 1996.

Our company is organized principally in two segments: government and commercial. Our government business encompasses specialized products principally serving defense customers and includes:

- Data links, including multifunction information distribution system (MIDS) terminals, joint tactical radio systems (JTRS) development and unmanned vehicle technologies,
- Information security and assurance products and services, which enable military and government users to communicate secure information over secure and non-secure networks, and
- Government satellite communication systems and products, including UHF DAMA satellite communications products consisting of modems, terminals and network control systems, and innovative broadband solutions to government customers to increase available bandwidth using existing satellite capacity.

Serving government customers with cost-effective products and solutions continues to be a critical and core element of our overall business strategy.

In recent years approximately one-half of our revenues has been generated from satellite based communications products and systems solutions to address commercial market needs. Our commercial business accounted for approximately 44% of our revenues in the three months ended December 28, 2007 and December 29, 2006, respectively, 45% and 46% of our revenues in the nine months ended December 28, 2007 and December 29, 2006, respectively, and 46% of our revenues in fiscal year 2007 and 51% of our revenues in fiscal year 2006.

The commercial segment comprises two business product groups: satellite networks and antenna systems. Our commercial business offers an end-to-end capability to provide customers with a broad range of satellite communication and other wireless communications equipment solutions including:

- Consumer broadband products and solutions to customers based on DOCSIS and DVB-RCS based technologies,
- Mobile broadband products and systems for in-flight, maritime and ground mobile broadband applications,
- Enterprise VSAT networks products and services,
- Satellite networking systems design and technology development, and
- Antenna systems for commercial and defense applications and customers.

With expertise in commercial satellite network engineering, gateway construction, and remote terminal manufacturing for all types of interactive communications services, we have the ability to take overall responsibility for designing, building, initially operating, and then handing over a fully operational, customized satellite network serving a variety of markets and applications.

There are a number of large new business opportunities we are pursuing in fiscal year 2008. In the government segment, the opportunities include future MIDS LVT production orders, international MIDS LVT orders, new MIDS joint tactical radio system development and pre-production contracts, additional funding for current information assurance projects, new information assurance contracts using our HAIPE technology, and orders for our KG-250 and KG-255 products. In our commercial segment, the opportunities include development and production agreements for next generation consumer broadband systems, production orders for existing consumer and mobile broadband systems and equipment, and further penetration in the North American market for enterprise VSAT and antenna systems. The probability and timing of these orders is not entirely predictable, so our revenue may vary somewhat from quarter-to-quarter or even year-to-year.

To date, our ability to grow and maintain our revenues has depended on our ability to identify and target markets where the customer places a high priority on the technology solution, and obtaining additional sizable contract awards from these customers. Due to the nature of this process, it is difficult to predict the probability and timing of obtaining awards in these markets.

We expect that our capital needs for fiscal year 2008 will be similar to fiscal year 2007. In fiscal years 2006 and 2007, we initiated and completed facility expansion and modernization projects in Carlsbad, California and Gilbert, Arizona, as well as expanded our production test equipment and lab development equipment and information technology to meet customer program requirements and growth forecasts. In fiscal year 2008, we have additional facility projects under way in Carlsbad, California, as well as production test equipment and information technology projects to support our growth needs. Our facility needs have normally been met with long-term lease agreements, but we do anticipate additional tenant improvements over the next two fiscal years associated with our expansion. Additionally, as our employee base increases, the need for additional computers and other equipment will also increase.

On August 2, 2007, we completed the acquisition of all of the outstanding capital stock of JAST, S.A., a Switzerland based, privately-held developer of microwave circuits and antennas for terrestrial and satellite applications, specializing in small, low-profile antennas for mobile satellite communications. The initial purchase price of approximately \$2.1 million was comprised primarily of \$452,000 related to the fair value of 14,424 shares of our common stock issued at the closing date, \$748,000 in cash consideration, the issuance of an \$800,000 payable, and approximately \$125,000 in direct acquisition costs. The \$748,000 in cash consideration paid to the former JAST stockholders plus approximately \$125,000 in direct acquisition costs less cash acquired of \$22,000 resulted in a net cash outlay of approximately \$851,000. The remaining \$800,000 is payable on the first anniversary of the closing date, of which \$483,000 will be paid in cash and \$317,000 will be paid in stock or cash, at our election. Under the terms of the purchase agreement, up to an additional \$4.5 million in consideration is payable in stock and/or cash, at our option, based on JAST achieving certain earnings performance and technology development targets during the two years following closing. No portion of this additional consideration is guaranteed. The additional consideration, if earned, is payable after JAST achieves specified earnings performance and technology development targets and will be recorded as additional purchase price.

Subsequent Event

In January 2008, the Company entered into several agreements with Space Systems/Loral (SS/L), Loral Space & Communications (Loral) and Telesat Canada (Telesat) related to the Company's high capacity satellite system. Under the satellite construction contract with SS/L, the Company will purchase a Ka-band satellite (ViaSat-1) for approximately \$209.1 million, subject to purchase price adjustment based on satellite performance. The Company does not believe the purchase price paid by ViaSat to SS/L for the ViaSat-1 satellite will materially change. In addition, the Company entered into a beam sharing agreement with Loral (Beam Sharing Agreement), whereby Loral is responsible for contributing 15% of the total costs (estimated at approximately \$60.0 million) associated with the ViaSat-1 satellite project. As part of this arrangement, Loral executed a separate contract with SS/L whereby Loral is purchasing the Canadian beams on the ViaSat-1 satellite (Loral Beams) for approximately \$36.9 million (15% of the total satellite cost of \$246.0 million). In addition, Loral remains responsible under the Beam Sharing Agreement to reimburse ViaSat for costs associated with launch, launch and in-orbit insurance, and operating the satellite. The reimbursed costs to ViaSat from Loral for launch and launch and the first year of in-orbit insurance are expected to be approximately \$23.1 million. Further, under the terms of the Beam Sharing Agreement, ViaSat is appointed the sole and exclusive supplier of end-user broadband terminals and hub equipment for broadband service operated over the Loral Beams. The Company also entered into an agreement with Telesat, whereby Telesat has agreed to transfer certain orbital slot license rights to the Company for ViaSat-1 satellite operation.

Critical Accounting Policies and Estimates

Management's Discussion and Analysis of Financial Condition and Results of Operations discusses our consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. We consider the policies discussed below to be critical to an understanding of our financial statements because their application places the most significant demands on management's judgment, with financial reporting results relying on estimation about the effect of matters that are inherently uncertain. We describe the specific risks for these critical accounting policies in the following paragraphs. For all of these policies, we caution that future events rarely develop exactly as forecast, and the best estimates routinely require adjustment.

Revenue recognition

A substantial portion of our revenues are derived from long-term contracts requiring development and delivery of complex equipment built to customer specifications. Certain of these contracts are accounted for under the percentage-of-completion method of accounting under the American Institute of Certified Public Accountants' Statement of Position 81-1 (SOP 81-1), "Accounting for Performance of Construction-Type and Certain Production-Type Contracts." Sales and earnings under these contracts are recorded based on the ratio of actual costs incurred to date to total estimated costs expected to be incurred related to the contract or as products are shipped under the units-of-delivery method.

The percentage-of-completion method of accounting requires management to estimate the profit margin for each individual contract and to apply that profit margin on a uniform basis as sales are recorded under the contract. The estimation of profit margins requires management to make projections of the total sales to be generated and the total costs that will be incurred under a contract. These projections require management to make numerous assumptions and estimates relating to items such as the complexity of design and related development costs, performance of subcontractors, availability and cost of materials, labor productivity and cost, overhead and capital costs, and manufacturing efficiency. These contracts often include purchase options for additional quantities and customer change orders for additional or revised product functionality. Purchase options and change orders are accounted for either as an integral part of the original contract or separately depending upon the nature and value of the item. Anticipated losses on contracts are recognized in full in the period in which losses become probable and estimable. During the three months ended December 28, 2007 and December 29, 2006, we recorded charges of approximately \$3.0 million and \$1.1 million, respectively, related to loss contracts. During the nine months ended December 28, 2007 and December 29, 2006, we recorded charges of approximately \$6.5 million and \$2.4 million, respectively, related to loss contracts.

Assuming the initial estimates of sales and costs under a contract are accurate, the percentage-of-completion method results in the profit margin being recorded evenly as revenue is recognized under the contract. Changes in these underlying estimates due to revisions in sales and future cost estimates or the exercise of contract options may result in profit margins being recognized unevenly over a contract as such changes are accounted for on a cumulative basis in the period estimates are revised.

We believe we have established appropriate systems and processes to enable us to reasonably estimate future cost on our programs through regular quarterly evaluations of contract costs, scheduling and technical matters by business unit personnel and management. Historically, in the aggregate, we have not experienced significant deviations in actual costs from estimated program costs, and when deviations that result in significant adjustments arise, we disclose the related impact in Management's Discussion and Analysis. However, a significant change in future cost estimates on one or more programs could have a material effect on our results of operations. For example, a one percent variance in our future cost estimates on open fixed-price contracts as of December 28, 2007 would change our income before income taxes by approximately \$400,000.

We also have contracts and purchase orders where revenue is recorded on delivery of products in accordance with SAB 104, "Staff Accounting Bulletin No. 104 Revenue Recognition." In this situation, contracts and customer purchase orders are used to determine the existence of an arrangement. Shipping documents and customer acceptance, when applicable, are used to verify delivery. We assess whether the sales price is fixed or determinable based on the payment terms associated with the transaction and whether the sales price is subject to refund or adjustment, and assesses collectibility based primarily on the creditworthiness of the customer as determined by credit checks and analysis, as well as the customer's payment history.

When a sale involves multiple elements, such as sales of products that include services, the entire fee from the arrangement is allocated to each respective element based on its relative fair value in accordance with EITF, 00-21, "Accounting for Multiple Element Revenue Arrangements," and recognized when the applicable revenue recognition criteria for each element are met. The amount of product and service revenue recognized is impacted by our judgments as to whether an arrangement includes multiple elements and, if so, what sufficient objective and reliable evidence of fair value exists for those elements. Changes to the elements in an arrangement and our ability to establish evidence for those elements could affect the timing of revenue recognition.

Accounting for Stock-Based Compensation

We grant options to purchase our common stock and award restricted stock units to our employees and directors under our equity compensation plans. Eligible employees can also purchase shares of our common stock at 85% of the lower of the fair market value on the first or the last day of each six-month offering period under our employee stock purchase plan. The benefits provided under these plans are stock-based payments subject to the provisions of revised Statement of Financial Accounting Standards (SFAS) No. 123 (SFAS 123R), "Share-Based Payment." Effective April 1, 2006, we use the fair value method to apply the provisions of SFAS 123R with a modified prospective application which provides for certain changes to the method for estimating the value of stock-based compensation. The valuation provisions of SFAS 123R apply to new awards and to awards that are outstanding on the effective date, which are subsequently modified or cancelled. Under the modified prospective application method, prior periods are not revised for comparative purposes. Stock-based compensation expense recognized under SFAS 123R for the three months ended December 28, 2007 and December 29, 2006 was \$1.9 million and \$1.6 million, respectively. Stock-based compensation expense recognized under SFAS 123R for the nine months ended December 28, 2007 and December 29, 2006 was \$5.6 million and \$2.4 million, respectively. At December 28, 2007, total unrecognized estimated compensation cost including estimated forfeitures related to non-vested stock options and restricted stock units granted prior to that date, and employee stock purchase plan was \$9.4 million, \$6.8 million and \$0, respectively, which are expected to be recognized over a weighted-average period of 2.7 years, 2.8 years and 0 years, respectively.

Review of Stock Option Grant Procedures

In August 2006, we commenced and completed a voluntary internal investigation, assisted by our outside legal counsel, of our historical stock option granting practices, stock option documentation and related accounting during the period from our initial public offering in December 1996 through June 30, 2006. At the conclusion of our investigation, our outside legal counsel and the company determined that there was no evidence of a pattern of intentionally misdating stock option grants to achieve an accounting result, or that any officer, director, or senior executive at the company willfully or knowingly engaged in stock options misdating, or had knowledge of others doing so.

During the investigation, we identified certain accounting errors associated with stock options granted primarily to certain non-executive new hire employees during the ten-year period from December 1996 to June 30, 2006. Based on the results of the investigation, we identified that certain stock options to non-executive new hires had incorrectly been accounted for using an accounting measurement date prior to the date that the new hires commenced employment. We concluded, with the concurrence of the Audit Committee, that the financial impact of these errors was not material to our consolidated financial statements for any annual period in which the errors related. In accordance with Accounting Principles Board Opinion No. 28, "Interim Financial Reporting," paragraph 29, we recorded a cumulative adjustment to compensation expense in the first quarter of fiscal year 2007 of \$703,000, net

of tax, because the effect of the correcting adjustment was not material to our fiscal year 2007 net income. This non-cash compensation expense adjustment will have no impact on future periods. There is no impact on revenue or net cash provided by operating activities as a result of recording the compensation expense adjustment.

Allowance for doubtful accounts

We make estimates of the collectibility of our accounts receivable based on historical bad debts, customer credit-worthiness and current economic trends when evaluating the adequacy of the allowance for doubtful accounts. Historically, our bad debts have been minimal; a contributing factor to this is that a significant portion of our sales has been to the U.S. government. In recent years, commercial customers have comprised a larger part of our revenues. Our accounts receivable balance was \$147.4 million, net of allowance for doubtful accounts of \$1.2 million as of December 28, 2007, and our accounts receivable balance was \$139.8 million, net of allowance for doubtful accounts of \$1.2 million as of March 30, 2007.

Warranty reserves

We provide limited warranties on a majority of our products for periods of up to five years. We record a liability for our warranty obligations when we ship the products based upon an estimate of expected warranty costs. We classify the amounts we expect to incur within twelve months as a current liability. For mature products, we estimate the warranty costs based on historical experience with the particular product. For newer products that do not have a history of warranty costs, we base our estimates on our experience with the technology involved and the types of failure that may occur. It is possible that our underlying assumptions will not reflect the actual experience, and in that case, we will make future adjustments to the recorded warranty obligation.

Goodwill and other intangible assets

We account for our goodwill under SFAS No. 142 (SFAS 142) "Goodwill and Other Intangible Assets." The SFAS 142 goodwill impairment model is a two-step process. First, it requires a comparison of the book value of net assets to the fair value of the reporting units that have goodwill assigned to them. If the fair value is determined to be less than book value, a second step is performed to compute the amount of the impairment. In this process, a fair value for goodwill is estimated, based in part on the fair value of the reporting unit used in the first step, and is compared to its carrying value. The shortfall of the value below carrying value represents the amount of goodwill impairment. We test goodwill for impairment during the fourth quarter every fiscal year, and when an event occurs or circumstances change such that it is reasonably possible that an impairment may exist.

We estimate the fair values of the related operations using discounted cash flows and other indicators of fair value. We base the forecast of future cash flows on our best estimate of the future revenues and operating costs, which we derive primarily from existing firm orders, expected future orders, contracts with suppliers, labor agreements, and general market conditions. Changes in these forecasts could cause a particular reporting unit to either pass or fail the first step in the SFAS 142 goodwill impairment model, which could significantly influence whether an impairment of goodwill needs to be recorded. We adjust the cash flow forecasts by an appropriate discount rate derived from our market capitalization plus a suitable control premium at the date of evaluation.

Impairment of long-lived assets (Property and equipment and other intangible assets)

We adopted SFAS No. 144 (SFAS 144) "Accounting for the Impairment or Disposal of Long-Lived Assets" on April 1, 2002. In accordance with SFAS 144, we assess potential impairments to our long-lived assets, including property and equipment and other intangible assets, when there is evidence that events or changes in circumstances indicate that the carrying value may not be recoverable. We recognize an impairment loss when the undiscounted cash flows expected to be generated by an asset (or group of assets) are less than the asset's carrying value. Any required impairment loss would be measured as the amount by which the asset's carrying value exceeds its fair value, and would be recorded as a reduction in the carrying value of the related asset and charged to results of operations. We have not identified any such impairments.

Income taxes

Management evaluates the realizability of our deferred tax assets and assesses the need for a valuation allowance on a quarterly basis. In accordance with SFAS No. 109 (SFAS 109), "Accounting for Income Taxes," net deferred tax assets are reduced by a

valuation allowance if, based on all the available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized.

On March 31, 2007, we adopted the provisions of Financial Accounting Standards Board (FASB) Interpretation No. 48 (FIN 48), "Accounting for Uncertainty in Income Taxes-an interpretation of FASB Statement No. 109." FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements in accordance with SFAS 109. FIN 48 prescribes a recognition threshold and measurement attribute for the financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return, and provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosure, and transition. For those benefits to be recognized, a tax position must be more-likely-thannot to be sustained upon examination by taxing authorities. The amount recognized is measured as the largest amount of benefit that is greater than fifty percent likely of being realized upon ultimate settlement.

We are subject to income taxes in the United States and numerous foreign jurisdictions. In the ordinary course of our business there are calculations and transactions where the ultimate tax determination is uncertain. In addition, changes in tax laws and regulations as well as adverse judicial rulings could adversely affect the income tax provision. We believe we have adequately provided for income tax issues not yet resolved with federal, state and foreign tax authorities. However, if these provided amounts prove to be more than what is necessary, the reversal of the reserves would result in tax benefits being recognized in the period in which we determine that provision for the liabilities is no longer necessary. If an ultimate tax assessment exceeds our estimate of tax liabilities, an additional charge to expense would result.

Derivatives

We enter into foreign currency forward and option contracts to hedge certain forecasted foreign currency transactions. Gains and losses arising from foreign currency forward and option contracts not designated as hedging instruments are recorded in investment income (expense) as gains (losses) on derivative instruments. Gains and losses arising from the effective portion of foreign currency forward and option contracts that are designated as cash-flow hedging instruments are recorded in accumulated other comprehensive income (loss) as unrealized gains (losses) on derivative instruments and accrued liabilities until the underlying transaction affects our earnings and are then recorded in the same income statement line as the underlying transaction. We had \$1.4 million of notional value of foreign currency forward contracts outstanding at December 28, 2007. We had \$0 of notional value of foreign currency forward contracts outstanding at March 30, 2007.

Self-insurance liabilities

We self-insure a portion of the exposure for losses related to workers' compensation costs and employee medical benefits. Accounting for workers' compensation expense and employee medical benefits require the use of estimates and assumptions regarding numerous factors, including ultimate severity of injuries, the timeliness of reporting injuries, and health care cost increases. We insure for workers' compensation and employee medical benefit liabilities under a large deductible program where losses are incurred up to certain specific and aggregate amounts. Accruals for claims under this self-insurance program are recorded as claims are incurred. We estimate our liability for claims incurred but not paid, including claims incurred but not recorded, based on the total incurred claims and paid claims, adjusted for ultimate losses as determined by our insurance carrier. We evaluate the estimated liability on a continuing basis and adjust accordingly. To date, workers' compensation expense and employee medical benefits expense have been within the range of managements' expectations.

Results of Operations

The following table presents, as a percentage of total revenues, income statement data for the periods indicated.

	Three months ended		Nine mon	ths ended
	December 28, 2007	December 29, 2006	December 28, 2007	December 29, 2006
Revenues	100.0%	100.0%	100.0%	100.0%
Operating expenses:				
Cost of revenues	69.6	72.7	71.8	74.4
Selling, general and administrative	13.8	14.2	13.8	13.1
Independent research and development	5.5	4.5	5.7	3.9
Amortization of intangible assets	1.6	2.0	1.7	1.9
Income from operations	9.5	6.6	7.0	6.7
Income before income taxes	10.4	7.0	7.9	7.0
Net income	6.7	7.8	5.4	5.6

Three Months Ended December 28, 2007 vs. Three Months Ended December 29, 2006

Revenues

	Three mont	Three months ended		Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Revenues	\$152.1	\$124.3	\$27.7	22.3%

The increase in revenues during the third quarter of fiscal year 2008 was due to our higher beginning backlog of \$439.0 million, quarterly customer awards of \$136.0 million and the conversion of certain backlog and awards into revenues. Revenue increases were experienced in both our government segment, increasing by \$15.9 million, and commercial segment, increasing by \$11.8 million. The revenue increase in the government segment was primarily derived from increased revenues of approximately \$5.6 million in certain information assurance products, \$6.9 million in next generation military satellite communication systems and \$2.8 million from our majority owned subsidiary, TrellisWare Technologies, Inc. (TrellisWare). The revenue increase in the commercial segment was primarily derived from increased sales of consumer and mobile broadband products of approximately \$13.4 million and \$6.4 million in higher sales from our Antenna Systems product group offset by a \$7.7 million reduction in enterprise VSAT product sales.

Cost of Revenues

	Three mor	Three months ended		Percentage
a m	December 28, 2007	December 29, 2006	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Cost of revenues	\$105.8	\$90.4	\$15.5	17.1%
Percentage of revenues	69.6%	72.7%		

The increase in quarterly cost of revenues from \$90.4 million to \$105.8 million was primarily due to the increase in revenues. However, cost of revenues decreased as a percentage of revenues from 72.7% to 69.6% primarily due to product cost reductions in our consumer and mobile broadband products totaling approximately \$2.2 million and better program performance in our Antenna Systems product group totaling approximately \$1.3 million for the three months ended December 28, 2007 compared to the same period last year. We also experienced better program performance in our government segment, contributing approximately \$4.9 million in cost of revenue reductions spread across various product groups. Cost of revenues for the three months ended December 28, 2007 and December 29, 2006 included approximately \$485,000 and \$423,000, respectively, in stock-based compensation expense related to our adoption of SFAS 123R. Cost of revenues may fluctuate in future quarters depending on the mix of products sold and services provided, competition, new product introduction costs and other factors.

Selling, General and Administrative Expenses

	Three mo	Three months ended		Percentage
~	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Selling, general and administrative	\$20.9	\$17.7	\$3.2	18.2%
Percentage of revenues	13.8%	14.2%		

The increase in selling, general and administrative (SG&A) expenses in the third quarter of fiscal year 2008 compared to the third quarter of fiscal year 2007 was primarily attributable to higher support costs of approximately \$1.8 million and higher selling and proposal costs of approximately \$700,000 to support our anticipated future revenue growth. SG&A expenses consist primarily of personnel costs and expenses for business development, marketing and sales, bid and proposal, finance, contract administration and general management. Some SG&A expenses are difficult to predict and vary based on specific government and commercial sales opportunities.

Independent Research and Development

	i nree montus ended		Dollar	Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Independent research and development	\$8.4	\$5.6	\$2.8	51.3%
Percentage of revenues	5.5%	4 5%		

The increase in independent research and development (IR&D) expenses reflects the current fiscal year third quarter over prior fiscal year third quarter increases primarily in the government segment of approximately \$1.8 million for planned development of next generation information assurance and unmanned aerial vehicle (UAV) technologies. The commercial segment contributed approximately \$1.0 million in IR&D increases due to planned efforts related to next generation broadband equipment and mobile antenna technologies. The higher IR&D expenses reflect our recognition of certain opportunities in these markets and the need to invest in the development of new technologies to meet these opportunities.

Amortization of Intangible Assets. The intangible assets from acquisitions occurring in fiscal years 2001, 2002, 2006, 2007 and 2008 are being amortized over original useful lives ranging from eight months to ten years. The amortization of intangible assets will decrease each year as the intangible assets with shorter lives become fully amortized.

The current and expected amortization expense for each of the following periods is as follows (in thousands):

	Am	ortization
For the nine months ended December 28, 2007	\$	7,173
Expected for the remainder of fiscal year 2008	\$	2,389
Expected for fiscal year 2009		9,020
Expected for fiscal year 2010		5,787
Expected for fiscal year 2011		5,024
Expected for fiscal year 2012		3,798
Thereafter		2,708
	\$	28,726

Interest Income. Interest income increased to \$1.6 million for the three months ended December 28, 2007 from \$553,000 for the three months ended December 29, 2006 due to higher average invested cash balances during the respective periods.

Interest Expense. Interest expense increased to \$269,000 for the three months ended December 28, 2007 from \$92,000 for the three months ended December 29, 2006. The increase in interest expense for the third quarter of fiscal year 2008 compared to third quarter of fiscal year 2007 was mainly due to the accretion of interest on the secured borrowing agreement entered into in the fourth quarter of fiscal year 2007. Commitment fees on our line of credit availability remained the same for each period. We had no outstanding borrowings under our line of credit at December 28, 2007 or December 29, 2006.

Provision (Benefit) for Income Taxes. Our effective tax rate for the three months ended December 28, 2007 was approximately 30.4%, which is approximately equal to the 28.4% estimated annual effective tax rate for the fiscal year 2008, compared to a tax

benefit of 12.7% for the three months ended December 29, 2006. The income tax benefit of approximately 12.7% for the third quarter of fiscal 2007 is lower than the expected annual effective tax rate primarily due to the recording of research and development tax credits allowed for in the third quarter of fiscal 2007 by the Tax Relief and Health Care Act of 2006, enacted on December 20, 2006, extending the research and development tax credit from January 1, 2006 to December 31, 2007. In the first and second quarters of fiscal year 2007, our estimated annual effective income tax rate did not include the effect of the extension of the research and development tax credit, which resulted in a catch-up adjustment of approximately \$2.0 million in the third quarter of fiscal year 2007. Also as a result of the extension of the research and development tax credit, approximately \$1.3 million of research and development tax credit generated in the fourth quarter of fiscal 2006 was recognized as a discrete item in the third quarter of fiscal 2007. Our estimated effective tax rate of approximately 28.4% for fiscal year 2008 reflects the expiration of the federal research and development tax credit at December 31, 2007. If the federal research and development tax credit is reinstated, we will have a lower effective tax rate and the amount of the tax rate reduction will depend on the effective date, the terms of the reinstatement as well as the amount of eligible research and development expenses in the reinstated period.

Our Segment Results for the Three Months Ended December 28, 2007 vs, Three Months Ended December 29, 2006

Government Segment

Revenues

	Three mon	Three months ended		Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Revenues	\$84.9	\$69.0	\$15.9	23.0%

The increase in government segment revenues related primarily to a higher beginning backlog and the receipt of \$54.4 million in awards during the third quarter of fiscal year 2008. The increased revenues resulted primarily from higher sales of certain information assurance products of approximately \$5.6 million, \$6.9 million in next generation military satellite communication systems and \$2.8 million from our majority owned subsidiary, TrellisWare.

Segment Operating Profit

	Three months ended		Dollar	Percentage	
	December 28,	December 29,	Increase	Increase	
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)	
Operating profit	\$15.6	\$ 8.8	\$6.8	77.7%	
Percentage of government segment revenues	18.4%	12.7%			

The increase in government segment operating profit was primarily related to margin gains from increased revenues of \$15.9 million and better program performance contributing approximately \$4.9 million in cost of revenue reductions and approximately \$840,000 in selling and support cost reductions spread across various product groups. This increase was offset by an increase in research and development expenses of \$1.8 million.

Commercial Segment

Revenues

	Three mon	Three months ended		Percentage
(In millions, except percentages)	December 28, 2007	December 29, 2006	Increase (Decrease)	Increase (Decrease)
Satellite Networks				
Revenues	\$52.2	\$46.8	\$ 5.4	11.6%
Antenna Systems				
Revenues	\$14.9	\$ 8.5	\$ 6.4	75.6%
Total Commercial Segment				
Revenues	\$67.1	\$55.3	\$11.8	21.4%
	20			

Commercial segment revenues increased \$11.8 million primarily due to higher sales of our consumer and mobile broadband products of approximately \$13.4 million compared to the same period last year offset by a \$7.7 million reduction in enterprise VSAT product sales. Antenna Systems product sales were also higher, increasing by approximately \$6.4 million due to the higher awards levels and related program performance.

Segment Operating Profit

	Three months ended		Dollar	Percentage	
(In millions, except percentages)	December 28, 2007	December 29, 2006	Increase (Decrease)	Increase (Decrease)	
Satellite Networks					
Satellite Networks operating (loss) profit	\$(0.1)	\$ 2.0	\$(2.1)	(104.3)%	
Percentage of Satellite Network revenues	(0.2)%	4.2%			
Antenna Systems					
Antenna Systems operating profit (loss)	\$ 1.3	\$(0.1)	\$ 1.4	3,380.5%	
Percentage of Antenna Systems revenues	9.0%	(0.5)%			
Total Commercial Segment					
Segment operating profit	\$ 1.3	\$ 1.9	\$(0.7)	(35.0)%	
Percentage of commercial segment revenues	1.9%	3.5%			

The commercial segment overall operating profit decrease of approximately \$700,000 compared to the same period last year was primarily due to a \$7.7 million reduction in enterprise VSAT product sales decreasing segment operating profits by approximately \$3.9 million and additional costs related to the Company's sales and marketing of wide area networking acceleration products of \$890,000. This decrease was offset by a \$3.4 million increase in operating profits relating to our consumer and mobile broadband products, primarily through product cost reductions, and a \$1.4 million increase in operating profits relating to our Antenna System products, primarily through improved overall program performance year over year.

Nine Months Ended December 28, 2007 vs. Nine Months Ended December 29, 2006

Revenues

	Nine months ended		Dollar	Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Revenues	\$427.2	\$384.5	\$42.7	11.1%

The increase in revenues was due to our beginning backlog of \$388.7 million and customer awards of \$461.5 million for the first nine months of our fiscal year 2008 and the conversion of certain backlog and awards into revenues. Year to date revenue growth was experienced in both our government segment, increasing by \$28.1 million, and commercial segment, increasing by \$14.6 million. The revenue increase in the government segment was primarily derived from increased revenues of approximately \$3.0 million in certain information assurance products, \$20.8 million in next generation military satellite communication systems and \$2.9 million from our majority owned subsidiary, TrellisWare. The revenue increase in the commercial segment was primarily derived from increased sales of consumer broadband products of approximately \$23.3 million, various other satellite networks products of approximately \$1.5 million and \$10.1 million in higher sales from our Antenna Systems product group offset by a \$20.3 million reduction in enterprise VSAT product sales.

Cost of Revenues

	Nine months ended		Dollar	Percentage	
	December 28,	December 29,	Increase	Increase	
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)	
Cost of revenues	\$306.8	\$285.9	\$20.8	7.3%	
Percentage of revenues	71.8%	74.4%			

The increase in cost of revenues from \$285.9 million to \$306.8 million was primarily due to an increase in our revenues. However, we experienced a decrease in the cost of revenues as a percentage of revenues from 74.4% to 71.8% primarily due to product cost reductions in our consumer and mobile broadband products totaling approximately \$9.1 million and better program performance in our Antenna Systems product group totaling approximately \$4.4 million for the nine months ended December 28, 2007 compared to

the same period last year. We also experienced better program performance in our government segment, contributing approximately \$1.1 million in cost of revenue reductions spread across various product groups. Cost of revenues for the nine months ended December 28, 2007 and December 29, 2006 included approximately \$1.4 million and \$730,000, respectively, in stock-based compensation expense related to our adoption of SFAS 123R. Cost of revenues may fluctuate in future quarters depending on the mix of products sold and services provided, competition, new product introduction costs and other factors.

Selling, General and Administrative Expenses

	Nille months ended		Dollar	Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Selling, general and administrative	\$59.1	\$50.3	\$8.7	17.4%
Percentage of revenues	13.8%	13.1%		

Nine menths anded

Nine months anded

The increase in SG&A expenses in the first nine months of fiscal year 2008 compared to the first nine months of fiscal year 2007 was primarily attributable to higher support costs of approximately \$5.0 million and higher selling and proposal costs of approximately \$2.8 million to support our anticipated future revenue growth and \$3.7 million in stock-based compensation expense recorded in the first nine months of fiscal year 2008 compared to \$1.6 million in stock-based compensation expense recorded in the first nine months of fiscal year 2007 related to our adoption of SFAS 123R. SG&A expenses consist primarily of personnel costs and expenses for business development, marketing and sales, bid and proposal, finance, contract administration and general management. Some SG&A expenses are difficult to predict and vary based on specific government and commercial sales opportunities.

Independent Research and Development

	Nille months ended		Dollar	Percentage	
	December 28,	December 29,	Increase	Increase	
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)	
Independent research and development	\$24.2	\$15.2	\$9.0	59.5%	
Percentage of revenues	5.7%	3.9%			

The increase in IR&D expenses reflects year over year increases primarily in the government segment of approximately \$6.5 million for planned development of next generation information assurance and UAV technologies. The commercial segment contributed approximately \$2.5 million in IR&D increases due to planned efforts related to next generation broadband equipment and mobile antenna technologies. The higher IR&D expenses reflect our recognition of certain opportunities in these markets and the need to invest in the development of new technologies to meet these opportunities.

Amortization of Intangible Assets. The intangible assets from acquisitions occurring in fiscal years 2001, 2002, 2006, 2007 and 2008 are being amortized over original useful lives ranging from eight months to ten years. The amortization of intangible assets will decrease each year as the intangible assets with shorter lives become fully amortized.

The current and expected amortization expense for each of the following periods is as follows (in thousands):

	Am	ortization
For the nine months ended December 28, 2007	\$	7,173
Expected for the remainder of fiscal year 2008	\$	2,389
Expected for fiscal year 2009		9,020
Expected for fiscal year 2010		5,787
Expected for fiscal year 2011		5,024
Expected for fiscal year 2012		3,798
Thereafter		2,708
	\$	28,726

Interest Income. Interest income increased to \$4.5 million for the nine months ended December 28, 2007 from \$1.3 million for the nine months ended December 29, 2006 due to higher average invested cash balances year over year.

Interest Expense. Interest expense increased to \$606,000 for the nine months ended December 28, 2007 from \$383,000 for the nine months ended December 29, 2006. The increase in interest expense was mainly due to the accretion of interest on the secured

borrowing agreement entered into in the fourth quarter of fiscal year 2007. Commitment fees on our line of credit availability remained the same year over year. We had no outstanding borrowings under our line of credit at December 28, 2007 or December 29, 2006.

Provision for Income Taxes. Our effective tax rate for the nine months ended December 28, 2007 was approximately 29.1%, which is approximately equal to the 28.4% estimated annual effective tax rate for the fiscal year 2008, compared to a 19.7% tax rate for the nine months ended December 29, 2006. The income tax provision for the nine months ended December 29, 2006 included approximately \$1.3 million of research and development tax credit generated in the fourth quarter of fiscal 2006 recognized as a discrete tax benefit in the nine-month period ended December 29, 2006. Our estimated effective tax rate of approximately 28.4% for fiscal year 2008 reflects the expiration of the federal research and development tax credit at December 31, 2007. If the federal research and development tax credit is reinstated, we will have a lower effective tax rate and the amount of the tax rate reduction will depend on the effective date, the terms of the reinstatement as well as the amount of eligible research and development expenses in the reinstated period.

Our Segment Results for the Nine Months Ended December 28, 2007 vs. Nine Months Ended December 29, 2006

Government Segment

Revenues

	Nine months ended		Dollar	Percentage
	December 28,	December 29,	Increase	Increase
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)
Revenues	\$235.4	\$207.3	\$28.1	13.5%

The increase in government segment revenues related primarily to a higher beginning backlog and the receipt of \$248.4 million in awards during first nine months of fiscal year 2008. The sales increase was principally from higher sales of our next generation military satellite communication systems of approximately \$20.8 million, certain information assurance products of approximately \$3.0 million and approximately \$2.9 million from our majority owned subsidiary, TrellisWare.

Segment Operating Profit

	Nine months ended		Dollar	Percentage	
	December 28,	December 29,	Increase	Increase	
(In millions, except percentages)	2007	2006	(Decrease)	(Decrease)	
Operating profit	\$33.7	\$33.7	\$0.0	(0.1)%	
Percentage of government segment revenues	14.3%	16.2%			

The government segment operating profit was relatively flat year over year. We experienced increased revenues of \$28.1 million and better program performance in our government segment, contributing approximately \$1.1 million in cost of revenue reductions spread across various product groups. This increase was offset by increases in research and development expenses of \$6.5 million and selling and support costs of \$1.9 million when compared to the same period last year.

Commercial Segment

Revenues

	Nine mon	Nine months ended		Percentage
(In millions, except percentages) Satellite Networks	December 28, 2007	December 29, 2006	Increase (Decrease)	Increase (Decrease)
Revenues	\$152.4	\$147.9	\$ 4.5	3.0%
Antenna Systems				
Revenues	\$ 39.4	\$ 29.3	\$10.1	34.7%
Total Commercial Segment				
Revenues	\$191.8	\$177.2	\$14.6	8.2%

Commercial segment revenues increased \$14.6 million primarily due to higher sales of our consumer and mobile broadband products of approximately \$23.3 million and various other satellite networks products of approximately \$1.5 million compared to the

same period last year offset by a \$20.3 million reduction in enterprise VSAT product sales. Antenna Systems product sales increased by approximately \$10.1 million due to higher awards levels and related program performance.

Segment Operating Profit

Nine months ended		Dollar	Percentage	
December 28, 2007	December 29, 2006	Increase (Decrease)	Increase (Decrease)	
\$0.8	\$(0.1)	\$0.8	1,527.8%	
0.5%	0.0%			
\$3.0	\$(0.6)	\$3.6	596.4%	
7.6%	(2.1)%			
\$3.8	\$(0.7)	\$4.4	672.5%	
2.0%	(0.4)%			
	\$0.8 0.5% \$3.0 7.6%	December 28, 2007 December 29, 2006 \$0.8 \$(0.1) 0.5% 0.0% \$3.0 \$(0.6) 7.6% (2.1)% \$3.8 \$(0.7)	December 28, 2007 December 29, 2006 Increase (Decrease) \$0.8 \$(0.1) \$0.8 0.5% 0.0% \$3.0 \$(0.6) \$3.6 7.6% (2.1)% \$3.8 \$(0.7) \$4.4	

The commercial segment overall operating profit increase of \$4.4 million compared to the same period last year was primarily due to continued revenue growth of certain consumer and mobile broadband products combined with overall improved cost of revenues performance contributing an additional \$9.1 million to commercial segment profits, offset by enterprise VSAT products and additional costs related to the Company's sales and marketing of wide area networking acceleration products of \$3.0 million. Additionally, our antenna products generated an additional \$4.4 million in operating profits primarily through improved overall program performance year over year.

Backlog

As reflected in the table below, both funded and firm (funded plus unfunded) backlog increased during the first nine months of fiscal year 2008 with increases coming from both our government and commercial segment.

	Decem	December 28, 2007 (in million		March 30, 2007	
Firm backlog		(.0125)		
Government segment	\$	233.1	\$	220.0	
Commercial segment		189.8		168.7	
Total	\$	422.9	\$	388.7	
Funded backlog					
Government segment	\$	210.4	\$	193.2	
Commercial segment		189.8		168.7	
Total	\$	400.2	\$	361.9	
Contract options	\$	39.3	\$	39.3	

The firm backlog does not include contract options. Of the \$422.9 million in firm backlog, approximately \$116.2 million is expected to be delivered during the remaining three months of fiscal year 2008, and the balance is expected to be delivered in fiscal year 2009 and thereafter. We include in our backlog only those orders for which we have accepted purchase orders.

Backlog is not necessarily indicative of future sales. A majority of our contracts can be terminated at the convenience of the customer since orders are often made substantially in advance of delivery, and our contracts typically provide that orders may be terminated with limited or no penalties. In addition, contracts may present product specifications that require us to complete additional product development. A failure to develop products meeting such specifications could lead to a termination of the related contacts.

The backlog amounts as presented are comprised of funded and unfunded components. Funded backlog represents the sum of contract amounts for which funds have been specifically obligated by customers to contracts. Unfunded backlog (primarily associated with our government segment contracts) represents future amounts that customers may obligate over the specified contract performance periods. Our customers allocate funds for expenditures on long-term contracts on a periodic basis. Our ability to realize revenues from contracts in backlog is dependent upon adequate funding for such contracts. Although funding of our contracts is not within our control, our experience indicates that actual contract fundings have ultimately been approximately equal to the aggregate amounts of the contracts.

Liquidity and Capital Resources

We have financed our operations to date primarily with cash flows from operations, bank line of credit financing and equity financing. The general cash needs of our government and commercial segments can vary significantly and depend on the type and mix of contracts (i.e. product or service, development or production, timing of payments, etc.) in backlog, the quality of the customer (i.e. U.S. government or commercial, domestic or international) and the duration of the contract. In addition, for both of our segments, program performance significantly impacts the timing and amount of cash flows. If a program is performing and meeting its contractual requirements, then the cash flow requirements are usually lower.

The cash needs of the government segment tend to be more of a function of the type of contract rather than customer quality. Also, U.S. government procurement regulations tend to restrict the timing of cash payments on the contract. In the commercial segment, our cash needs are driven primarily by the quality of the customer and the type of contract. The quality of the customer will typically affect the specific contract cash flow and whether financing instruments are required by the customer. In addition, the commercial environment tends to provide for more flexible payment terms with customers, including advance payments.

Cash provided by operating activities for the first nine months of fiscal year 2008 was \$46.1 million as compared to \$42.3 million for the first nine months of fiscal year 2007. The \$3.8 million increase in cash provided by operating activities for the first nine months of fiscal year 2008 as compared to the first nine months of fiscal year 2007 was primarily attributable to higher year over year net income of \$1.4 million, increase in non-cash add-backs for depreciation of \$1.2 million and \$1.9 million in non-cash stock-based compensation offset by various other changes in operating assets net of liabilities. Accounts receivable increased by \$7.6 million from year end primarily due to delayed payments from a government customer who changed payment systems and experienced difficulty in processing payments. We expect the delayed payments to be made in the fourth quarter of fiscal year 2008.

Cash used in investing activities for the first nine months of fiscal year 2008 was \$35.2 million as compared to \$12.3 million for the first nine months of fiscal year 2007. The increase in cash used in investing activities primarily relates to the purchase of approximately \$11.8 million of short-term investments classified as held-to-maturity, \$8.7 million in cash paid to certain former Efficient Channel Coding, Inc. (ECC) stockholders under the terms of the acquisition agreement for ECC, \$260,000 in cash paid to former stockholders of Enerdyne Technologies, Inc. (Enerdyne) under the terms of the Enerdyne acquisition agreement and approximately \$851,000 in cash paid for the acquisition of JAST on the closing date under the terms of the JAST acquisition agreement. In addition approximately \$13.6 million in cash outflow relates to capital expenditures for the first nine months of fiscal year 2008.

Cash provided by financing activities for the first nine months of fiscal year 2008 was \$6.7 million as compared to cash provided by financing activities for the first nine months of fiscal year 2007 of \$9.7 million. The majority of the activity for both years is due to cash received from the exercise of employee stock options, stock purchases through our employee stock purchase plan and cash inflows related to the incremental tax benefit from stock option exercises slightly offset, in fiscal year 2008, by the repurchase of common stock related to net share settlement for certain employee tax liabilities in connection with the vesting of restricted stock unit awards during the third quarter of fiscal year 2008.

In January 2008, we entered into several agreements with Space Systems/Loral (SS/L), Loral Space & Communications (Loral) and Telesat Canada (Telesat) related to the Company's high capacity satellite system. Under the satellite construction contract with SS/L, the Company will purchase a Ka-band satellite (ViaSat-1) for approximately \$209.1 million, subject to purchase price adjustment based on satellite performance. The Company does not believe the purchase price paid by ViaSat to SS/L for the ViaSat-1 satellite will materially change. In addition, the Company entered into a beam sharing agreement with Loral (Beam Sharing Agreement), whereby Loral is responsible for contributing 15% of the total costs (estimated at approximately \$60 million) associated with the ViaSat-1 satellite project. As part of this arrangement, Loral executed a separate contract with SS/L whereby Loral is purchasing the Canadian beams on the ViaSat-1 satellite (Loral Beams) for approximately \$36.9 million (15% of the total satellite cost of \$246.0 million). In addition, Loral remains responsible under the Beam Sharing Agreement to reimburse ViaSat for costs associated with launch, launch and in-orbit insurance, and operating the satellite. The reimbursed costs to ViaSat from Loral for launch and launch and the first year of in-orbit insurance are expected to be approximately \$23.1 million.

On May 23, 2006, in connection with our ECC acquisition, we agreed under the terms of the ECC acquisition agreement to pay the maximum additional consideration amount to the former ECC stockholders in the amount of \$9.0 million, which was accrued as of March 30, 2007. The \$9.0 million was payable in cash or stock, at our option, in May 2007. Accordingly, on May 30, 2007, we paid approximately \$9.0 million of additional cash consideration to the former stockholders of ECC. The additional purchase price consideration of \$9.0 million was recorded as additional goodwill in the Satellite Networks product group in the commercial segment in the first quarter of fiscal year 2007.

As of March 30, 2007, in connection with our Enerdyne acquisition and under the terms of the Enerdyne acquisition agreement, we owed an additional consideration amount to the former Enerdyne stockholders in the amount of \$5.9 million, which was accrued and recorded as additional goodwill in the government segment as of March 30, 2007. The \$5.9 million was payable in cash and stock in accordance with certain terms of the arrangement, in May 2007. Accordingly, on May 3, 2007, we paid \$5.9 million of additional consideration to the former stockholders of Enerdyne, which was comprised of 170,763 shares of common stock and \$260,000 in cash.

At December 28, 2007, we had \$133.0 million in cash, cash equivalents and short-term investments, \$240.0 million in working capital and no outstanding borrowings under our line of credit. At March 30, 2007, we had \$103.4 million in cash and cash equivalents and short-term investments, \$187.4 million in working capital and no outstanding borrowings under our line of credit.

On January 31, 2005, we entered into a three-year, \$60 million revolving credit facility (the Facility) which was amended on January 25, 2008, extending the term of the Facility to April 30, 2008. At December 28, 2007, we had \$8.2 million outstanding under standby letters of credit leaving borrowing availability under the Facility of \$51.8 million.

Borrowings under the Facility are permitted up to a maximum amount of \$60 million, including up to \$15 million of letters of credit. Borrowings under the Facility bear interest, at our option, at either the lender's prime rate or at LIBOR (London Interbank Offered Rate) plus, in each case, an applicable margin based on the ratio of our total funded debt to EBITDA (income from operations plus depreciation and amortization). The Facility is collateralized by substantially all of our personal property assets.

The Facility contains financial covenants that set a minimum EBITDA limit for the twelve-month period ending on the last day of any fiscal quarter at \$30 million, a minimum tangible net worth as of the last day of any fiscal quarter at \$135 million and a minimum quick ratio (sum of cash and cash equivalents, accounts receivable and marketable securities, divided by current liabilities) as of the last day of any fiscal quarter at 1.50 to 1.00. We were in compliance with our loan covenants at December 28, 2007.

In April 2007, we filed an updated universal shelf registration statement with the Securities and Exchange Commission for the future sale of up to \$400 million of debt securities, common stock, preferred stock, depositary shares and warrants. The securities may be offered from time to time, separately or together, directly by us or through underwriters at amounts, prices, interest rates and other terms to be determined at the time of the offering. We currently intend to use the net proceeds from the sale of the securities under the shelf registration statement, if any, for general corporate purposes, including acquisitions, capital expenditures and working capital.

Our future capital requirements will depend upon many factors, including the timing of cash required for the ViaSat-1 satellite project, expansion of our research and development and marketing efforts and the nature and timing of orders. Additionally, we will continue to evaluate possible acquisitions of, or investments in complementary businesses, products and technologies which may require the use of cash. We believe that our current cash balances and net cash expected to be provided by operating activities will be sufficient to meet our operating requirements for at least the next twelve months. However, we may sell additional equity or debt securities or obtain credit facilities to further enhance our liquidity position. The sale of additional securities could result in additional dilution of our stockholders. We invest our cash in excess of current operating requirements in short-term, interest-bearing, investment-grade securities.

Contractual Obligations

The following table sets forth a summary of our obligations under operating leases, irrevocable letters of credit, purchase commitments and other long-term liabilities for the periods indicated (in thousands):

		For the	remainder of		For the fiscal years			
	Total	fiscal year 2008		2009-2010	2011-2012	After 2013		
Operating leases	\$124,869	\$	2,700	\$ 26,103	\$ 26,914	\$ 69,152		
Standby letters of credit	8,218		148	6,043	2,027	_		
Secured borrowings and accrued interest	5,111		5,111	_	_	_		
Purchase commitments	124,799		50,110	74,689	_	_		
Total	\$ 262,997	\$	58,069	\$ 106,835	\$ 28,941	\$ 69,152		

We purchase components from a variety of suppliers and use several subcontractors and contract manufacturers to provide design and manufacturing services for our products. During the normal course of business, we enter into agreements with subcontractors, contract manufacturers and suppliers that either allow them to procure inventory based upon criteria as defined by us or that establish the parameters defining our requirements. In certain instances, these agreements allow us the option to cancel, reschedule and adjust our requirements based on our business needs prior to firm orders being placed. Consequently, only a portion of our reported purchase commitments arising from these agreements are firm, non-cancelable and unconditional commitments.

Our Condensed Consolidated Balance Sheets as of December 28, 2007 and March 30, 2007 include \$17.4 million and \$13.3 million, respectively, classified as "Other liabilities." This caption primarily consists of our long-term warranty obligations, deferred lease credits, long-term portion of our secured borrowing, and long-term unrecognized tax position liabilities. The secured borrowing obligations have been included in the table above based on the terms of the arrangement. These remaining liabilities have been excluded from the above table as the timing and/or the amount of any cash payment is uncertain. See Note 10 of the Notes to Consolidated Financial Statements for additional information regarding our income taxes and related tax positions and Note 7 for a discussion of our product warranties.

Subsequent Event

In January 2008, the Company entered into several agreements with Space Systems/Loral (SS/L), Loral Space & Communications (Loral) and Telesat Canada (Telesat) related to the Company's high capacity satellite system. Under the satellite construction contract with SS/L, the Company will purchase a Ka-band satellite (ViaSat-1) for approximately \$209.1 million, subject to purchase price adjustment based on satellite performance. The Company does not believe the purchase price paid by ViaSat to SS/L for the ViaSat-1 satellite will materially change. In addition, the Company entered into a beam sharing agreement with Loral (Beam Sharing Agreement), whereby Loral is responsible for contributing 15% of the total costs (estimated at approximately \$60.0 million) associated with the ViaSat-1 satellite project. As part of this arrangement, Loral executed a separate contract with SS/L whereby Loral is purchasing the Canadian beams on the ViaSat-1 satellite (Loral Beams) for approximately \$36.9 million (15% of the total satellite cost of \$246.0 million). In addition, Loral remains responsible under the Beam Sharing Agreement to reimburse ViaSat for costs associated with launch, launch and in-orbit insurance, and operating the satellite. The reimbursed costs to ViaSat from Loral for launch and launch and the first year of in-orbit insurance are expected to be approximately \$23.1 million. Further, under the terms of the Beam Sharing Agreement, ViaSat is appointed the sole and exclusive supplier of end-user broadband terminals and hub equipment for broadband service operated over the Loral Beams. The Company also entered into an agreement with Telesat, whereby Telesat has agreed to transfer certain orbital slot license rights to the Company for ViaSat-1 satellite operation.

Recent Accounting Requirements

In September 2006, the FASB issued Statement No. 157 (SFAS 157), "Fair Value Measurements." SFAS 157 defines fair value, establishes a framework and gives guidance regarding the methods used for measuring fair value, and expands disclosures about fair value measurements. SFAS 157 is effective for financial statements issued for fiscal years beginning after November 15, 2007 (our fiscal year 2009), and interim periods within those fiscal years. We are currently assessing the impact SFAS 157 will have on our results of operations and financial position.

In February 2007, the FASB issued Statement No. 159 (SFAS 159), "The Fair Value Option for Financial Assets and Financial Liabilities," which permits entities to choose to measure many financial instruments and certain other items at fair value that are not

currently required to be measured at fair value. SFAS 159 will be effective for us in fiscal year 2009. We are currently evaluating the impact of adopting SFAS 159 on our financial position, cash flows, and results of operations.

In June 2007, the FASB ratified Emerging Issues Task Force (EITF) Issue No. 07-3 (EITF 07-3), "Accounting for Nonrefundable Advance Payments for Goods or Services to Be Used in Future Research and Development Activities." This issue provides that nonrefundable advance payments for goods or services that will be used or rendered for future research and development activities should be deferred and capitalized. Such amounts should be recognized as an expense as the related goods are delivered or the related services are performed. EITF 07-3 is effective for us in fiscal year 2009. The adoption of EITF Issue No. 07-3 is not expected to have a material impact on our consolidated financial position, results of operations and cash flows.

In December 2007, the FASB issued SFAS No. 141 (revised 2007), "Business Combinations" (SFAS 141R). The purpose of issuing the statement is to replace current guidance in SFAS 141 to better represent the economic value of a business combination transaction. The changes to be effected with SFAS 141R from the current guidance include, but are not limited to: (1) acquisition costs will be recognized as expenses separately from the acquisition; (2) known contractual contingencies at the time of the acquisition will be considered part of the liabilities acquired measured at their fair value; all other contingencies will be part of the liabilities acquired measured at their fair value only if it is more likely than not that they meet the definition of a liability; (3) contingent consideration based on the outcome of future events will be recognized and measured at the time of the acquisition; (4) business combinations achieved in stages (step acquisitions) will need to recognize the identifiable assets and liabilities, as well as noncontrolling interests, in the acquiree, at the full amounts of their fair values; and (5) a bargain purchase (defined as a business combination in which the total acquisition-date fair value of the identifiable net assets acquired exceeds the fair value of the consideration transferred plus any noncontrolling interest in the acquiree) will require that excess to be recognized as a gain attributable to the acquirer. SFAS 141R will be effective for us in fiscal year 2010. We are currently evaluating the impact of SFAS 141R.

In December 2007, the FASB issued SFAS No. 160 (SFAS 160), "Noncontrolling Interests in Consolidated Financial Statements — an amendment of ARB No. 51." SFAS 160, which changes the accounting and reporting for business acquisitions and non-controlling interests in subsidiaries, was issued to improve the relevance, comparability, and transparency of financial information provided to investors. Moreover, SFAS 160 eliminates the diversity that currently exists in accounting for transactions between an entity and noncontrolling interests by requiring they be treated as equity transactions. SFAS 160 will be effective for us in fiscal year 2010. We are currently evaluating the impact that SFAS 160 will have on our financial statements and disclosures.

Off-Balance Sheet Arrangements

We had no material off-balance sheet arrangements at December 28, 2007 as defined in Regulation S-K Item 303(a)(4) other than as discussed under Contractual Obligations above or disclosed in the notes to our financial statements included in this filing or in our 2007 Annual Report on Form 10-K.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Our financial instruments consist of cash and cash equivalents, short-term investments, trade accounts receivable, accounts payable, and short-term obligations including the revolving line of credit. We consider investments in highly liquid instruments purchased with a remaining maturity of 90 days or less at the date of purchase to be cash equivalents. Our exposure to market risk for changes in interest rates relates primarily to short-term investments and short-term obligations. As a result, we do not expect fluctuations in interest rates to have a material impact on the fair value of these securities.

As of December 28, 2007, there were two foreign currency exchange contracts outstanding which are intended to reduce the foreign currency risk for amounts payable to vendors in Euros. The foreign exchange contracts with a notional amount of \$1.4 million had a fair value of a net liability of approximately \$11,000 as of December 28, 2007. The fair value of this foreign currency forward contract as of December 28, 2007 would have changed by \$136,000 if the foreign currency exchange rate for the Euro to the U.S. dollar on this forward contract had changed by 10%.

Item 4. Controls and Procedures

We maintain disclosure controls and procedures designed to provide reasonable assurance of achieving the objective that information in our Exchange Act reports is recorded, processed, summarized and reported within the time periods specified and pursuant to the requirements of the Securities and Exchange Commission's rules and forms. We carried out an evaluation, with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures as of December 28, 2007, the end of the period covered by this Quarterly Report. Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective at the reasonable assurance level as of December 28, 2007.

During the period covered by this Quarterly Report, there have been no changes in our internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II — OTHER INFORMATION

Item 1. Legal Proceedings

A review of our current litigation is disclosed in the Notes to Condensed Consolidated Financial Statements. See "Notes to Condensed Consolidated Financial Statements - Note 8 - Commitments and Contingencies."

Item 1A. Risk Factors

In addition to the risk factors set forth in our Securities and Exchange Commission filings, including without limitation, our most recent Annual Report on Form 10-K and Quarterly Reports on Form 10-Q, we face new risks associated with our satellite project.

In January 2008, we executed an agreement to purchase ViaSat-1, our first broadband satellite. We also plan to develop next generation SurfBeam ground infrastructure and terminals for use with the satellite. We currently plan to launch our ViaSat-1 satellite in early 2011 and introduce service later in 2011. If we are unable to successfully launch this satellite and implement our satellite service business in a timely manner, or at all, including as a result of any of the following risks, we will be unable to realize the anticipated benefits from our satellite and associated service business, and our business, financial condition and results of operations could be materially adversely affected:

- Cost and Schedule Risks. The cost of completing the satellite and developing the associated next generation SurfBeam ground infrastructure may be more than anticipated and there may be delays in completing the satellite and SurfBeam infrastructure within the expected timeframe. We may be required to spend in excess of our current forecast for the completion, launch and launch insurance of the ViaSat-1 satellite or for the development associated with the next generation SurfBeam equipment. The construction and launch of satellites are often subject to delays, including satellite and launch vehicle construction delays, cost overruns, periodic unavailability of reliable launch opportunities, and delays in obtaining regulatory approvals. If the ViaSat-1 construction schedule is not met, there may be even further delays because there can be no assurance that a launch opportunity will be available at the time the satellite is ready to be launched and we may not be able to obtain or maintain regulatory authority or International Telecommunication Union (ITU) priority necessary to implement ViaSat-1 as proposed.
- Business Plan. We may be unsuccessful in implementing our business plan for the satellite service, or we may not be able to achieve the revenue that we expect from the satellite service. A failure to attract either distributors or customers in a sufficient number would result in lower revenues than anticipated. In addition, we will incur startup losses associated with the launch and operation of the satellite service until we acquire a sufficient number of customers, which may not occur as expected or at all.
- Regulatory Risk. If we do not obtain all requisite regulatory approvals for the construction, launch and operation of the ViaSat-1 satellite, or the licenses obtained impose operational restrictions on us, our ability to generate revenue and profits could be materially adversely affected. In addition, under certain circumstances, government licenses are subject to revocation or modification, and upon expiration, renewal may not be granted. In certain cases, satellite system operators are obligated by governmental regulation and procedures of the ITU to coordinate the operation of their systems with other users of the radio spectrum in order to avoid causing interference to those other users. Coordination may require a satellite system operator to reduce power, avoid operating on certain frequencies, relocate its satellite to another orbital location and/or otherwise modify planned or existing operations. Federal Communications Commission (FCC) satellite authorizations are customarily subject to conditions imposed by the FCC in addition to the FCC's general authority to modify, cancel or revoke those authorizations. Failure to comply with such requirements, or comply in a timely manner, could lead to the loss of authorizations and could have a material adverse effect on our ability to generate revenue.
- Launch Risks. There are risks associated with the launch of satellites, including launch failure, damage or destruction during launch and improper orbital placement. Launch failures result in significant delays in the deployment of satellites because of the need both to construct replacement satellites, which can take up to 36 months, and obtain other launch opportunities. The overall historical loss rate in the satellite industry for all launches of commercial satellites in fixed orbits in the last five years is estimated by some industry participants to be 10% but could at any time be higher.
- *In-Orbit Risks*. The ViaSat-1 satellite will be subject to similar potential satellite failures or performance degradations as other satellites. Satellites are subject to in-orbit risks including malfunctions, commonly referred to as anomalies, and collisions with meteoroids, decommissioned spacecraft or other space debris. Anomalies occur as a result of various factors, such as satellite manufacturing errors, problems with the power systems or control systems of the satellites and general failures resulting from operating satellites in the harsh space environment. To the extent there is an anomaly or other in-orbit failure with respect to the ViaSat-1 satellite, we will not have a replacement satellite. Additionally, we could be required to reposition the antennas of our customers, which could require new or modified licenses from regulatory authorities.
- *Minimum Design Life.* Our ability to earn revenue depends on the usefulness of the ViaSat-1 satellite. Each satellite has a limited useful life. A number of factors affect the useful lives of the satellites, including, among other things, the quality of their construction, the durability of their component parts, the ability to continue to maintain proper orbit and control over the satellite's functions, the efficiency of the launch vehicle used, and the remaining on-board fuel following orbit insertion. The minimum design life of ViaSat-1 is estimated to be 15 years. In addition, continued improvements in satellite technology

- may obsolete the ViaSat-1 satellite prior to the end of its life. Therefore, we can provide no assurance as to the actual useful life of ViaSat-1.
- Insurance Risks. We intend to seek launch and in-orbit insurance for the ViaSat-1 satellite, but we may not be able to obtain insurance on reasonable economic terms or at all. If we are able to obtain insurance, it will contain customary exclusions and will not likely cover the full cost of constructing and launching the ViaSat-1 satellite, nor will it cover business interruptions or similar losses. In addition, the occurrence of any anomalies on other satellites, including Ka-band satellites, may materially adversely affect our ability to insure ViaSat-1 at commercially reasonable premiums, if at all.

Item 4. Submission of Matters to a Vote of Security Holders

At our 2007 annual meeting of stockholders held on October 3, 2007 the stockholders voted on the following proposal and cast their votes as follows:

	For	Withheld
1. Election of Directors:		
B. Allen Lay	26,300,947	1,390,376
Dr. Jeffrey M. Nash	25,151,164	2,540,159

Item 5. Other Information

In connection with a contract dated January 7, 2008 that we executed with Space Systems/Loral, Inc. (SS/L) for the construction by SS/L of a high capacity broadband satellite, we entered into a Beam Sharing Agreement dated January 11, 2008 with Loral Space & Communications Inc. (Loral), the parent of SS/L, pursuant to which Loral is investing in the Canadian coverage portion of the satellite. The Beam Sharing Agreement provides for, among other things, (1) the purchase by Loral of a portion of the satellite payload providing coverage into Canada, (2) payment by Loral of 15% of the total costs of the satellite, launch and associated services, launch insurance, and telemetry, tracking and command services for the satellite, and (3) the appointment of ViaSat as the sole and exclusive supplier of end-user broadband terminals and hub equipment for broadband service operated over the satellite payload providing coverage into Canada. The aggregate price to be paid by Loral for the foregoing is estimated to be approximately \$60 million.

On January 25, 2008, the Company amended the January 31, 2005 \$60 million revolving Second Amended and Restated Revolving Loan Agreement extending the agreement's current terms and conditions to April 30, 2008.

The foregoing discussion does not purport to be complete and is qualified in its entirety by reference to the Beam Sharing Agreement and Second Amendment to Second Amended and Restated Revolving Loan Agreement, a copy of which is attached to this Quarterly Report as Exhibit 10.2 and Exhibit 10.3 and is incorporated herein by reference.

Item 6. Exhibits

The Exhibit Index on page 41 is incorporated herein by reference as the list of exhibits required as part of this report.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

VIASAT, INC.

February 6, 2008

/s/ Mark D. Dankberg

Mark D. Dankberg Chairman of the Board and Chief Executive Officer (Principal Executive Officer)

/s/ Ronald G. Wangerin

Ronald G. Wangerin Vice President, Chief Financial Officer (Principal Financial and Accounting Officer)

40

EXHIBIT INDEX

Exhibit Number	Description
10.1*	Contract for the ViaSat Satellite Program dated as of January 7, 2008 between ViaSat, Inc. and Space Systems/Loral, Inc.
10.2	Beam Sharing Agreement dated January 11, 2008 between ViaSat, Inc. and Loral Space & Communications, Inc.
10.3	Second Amendment to Second Amended and Restated Revolving Loan Agreement dated January 25, 2008 between ViaSat,Inc. and Union Bank of California, N.A. and Comerica Bank.
31.1	Certification of Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certifications Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

^{*} Portions of this exhibit (indicated by asterisks) have been omitted pursuant to a request for confidential treatment.

CERTAIN MATERIAL (INDICATED BY AN ASTERISK) HAS BEEN OMITTED FROM THIS DOCUMENT PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT. THE OMITTED MATERIAL HAS BEEN FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION.

CONTRACT

Between

ViaSat, Inc.

And

Space Systems/Loral, Inc.

for the

ViaSat Satellite Program

The attached Contract and information contained therein are confidential and proprietary to ViaSat, Inc. and Space Systems/Loral, Inc. and shall not be published or disclosed to any third party except as permitted by the terms and conditions of this Contract.



SS/L-TP20701 ViaSat Contract

TABLE OF CONTENTS

ARTICLE TABLE OF CO PREAMBLE	ONTENTS	PAGE i 1
RECITALS		2
Article 1 -	DEFINITIONS	3
Article 2 -	SCOPE OF WORK	16
Article 3 -	DELIVERABLE ITEMS AND DELIVERY SCHEDULE	26
Article 4 -	PRICE	28
Article 5 -	PAYMENTS	33
Article 6 -	PURCHASER-FURNISHED ITEMS	37
Article 7 -	COMPLIANCE WITH LAWS AND DIRECTIVES	40
Article 8 -	ACCESS TO WORK-IN-PROGRESS	43
Article 9 -	SATELLITE PRE-SHIPMENT REVIEW (SPSR) AND DELIVERY	47
Article 10 -	ACCEPTANCE OF SATELLITE, LAUNCH SUPPORT AND MISSION OPERATIONS SUPPORT SER	
, =0	IN-ORBIT TEST	51
Article 11 -	ACCEPTANCE OF DELIVERABLE ITEMS OTHER THAN THE SATELLITE	53
Article 12 -	TRANSFER OF TITLE AND RISK OF LOSS	58
Article 13 -	ORBITAL PERFORMANCE INCENTIVES	60
Article 14 -	CORRECTIVE MEASURES IN THE SATELLITE AND OTHER DELIVERABLE ITEMS	65
Article 15 -	REPRESENTATIONS AND WARRANTIES	66
Article 16 -	CHANGES	73
Article 17 -	FORCE MAJEURE	75
Article 18 -	PURCHASER DELAY OF WORK	78
Article 19 -	INTELLECTUAL PROPERTY INDEMNITY	79
Article 20 -	INDEMNIFICATION	81
Article 21 -	TERMINATION FOR CONVENIENCE	84
Article 22 -	LIQUIDATED DAMAGES	86
Article 23 -	TERMINATION FOR DEFAULT	94
Article 24 -	RESERVED	98
Article 25 -	DISPUTE RESOLUTION	99
Article 26 -	INTER-PARTY WAIVER OF LIABILITY FOR A LAUNCH	102
Article 27 -	LIMITATION OF LIABILITY	104
Article 28 -	DISCLOSURE AND HANDLING OF PROPRIETARY INFORMATION	106

LORAL

SS/L-TP20701 ViaSat Contract

ARTICLE		PAGE
Article 29 -	CONTRACT TECHNOLOGY ESCROW	109
Article 30 -	PUBLIC RELEASE OF INFORMATION	111
Article 31 -	NOTICES	112
Article 32 -	RISK MANAGEMENT SERVICES	114
Article 33 -	ORDER OF PRECEDENCE	118
Article 34 -	GENERAL	119
Article 35 -	GROUND STORAGE	124
Article 36 -	ORBIT MANEUVER [***] AND LAUNCH VEHICLE [***]	126
Article 37 -	CONTRACTOR PERSONNEL	128
Article 38 -	SUBCONTRACTS	130
Article 39 -	INTELLECTUAL PROPERTY	132
Article 40 -	LENDER REQUIREMENTS	134
Article 41 -	SECURITY INTEREST	135

Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



PREAMBLE

This Contract is entered into effective as of the 7th day of January, 2008 (the "Effective Date of Contract" or "EDC"), by and between ViaSat, Inc. ("ViaSat"), a corporation organized and existing under the laws of the state of Delaware having an office and place of business at 6155 El Camino Real, Carlsbad, CA 92009-1699 (hereinafter referred to as "Purchaser"), and Space Systems/Loral, Inc., a corporation organized and existing under the laws of the state of Delaware, having an office and place of business at 3825 Fabian Way, Palo Alto, CA 94303-4604 (hereafter referred to as "Contractor", and Purchaser and Contractor are hereafter referred to collectively as the "Parties" or individually as a "Party"), regarding the ViaSat Satellite program.



SS/L-TP20701 ViaSat Contract

RECITALS

WHEREAS, Purchaser desires to procure a communications Satellite, a Dynamic Satellite Simulator, Launch Support Services, Mission Operations Support Services, Training services and other items and services to the extent and subject to the terms and conditions set forth herein, and

WHEREAS, Purchaser has entered into a CondoSat Agreement with Loral Space & Communications, Inc., a Delaware company, whereby the parties thereto intend that Purchaser will take title to the Satellite, except for the Loral Space Payload portion (to which CondoSat Associate will take title) at Intentional Ignition; however, except for certain limited rights to be identified in a separate agreement between CondoSat Associate and Contractor, Purchaser shall have the exclusive rights (including the exclusive rights of exercise) associated with the Satellite and other Deliverable Items as further set forth herein; and

WHEREAS, Contractor is willing to furnish such Satellite (including the transfer of title to the Loral Space Payload directly to CondoSat Associate per separate agreement), a Dynamic Satellite Simulator, Launch Support Services, Mission Operations Support Services, Training services and other items and services to the extent of and subject to the terms and conditions set forth herein, in consideration of the price and other valid consideration.

NOW, THEREFORE, in consideration of the price and other valid consideration and the mutual covenants and agreements contained herein and intending to be legally bound, the Parties agree as follows:



SS/L-TP20701 ViaSat Contract

ARTICLE 1 — DEFINITIONS

Capitalized terms used and not otherwise defined herein shall have the following meanings:

- 1.1 "Acceptance" (i) with respect to the Satellite shall be as provided in Article 10, and (ii) with respect to any Deliverable Item other than the Satellite shall be as provided in Article 11.
- 1.2 "[***]" has the meaning set forth in Article 22.2.1.
- 1.3 "Actual Costs" shall mean Contractor's direct and actual costs as determined in accordance with Contractor's standard accounting practices uniformly applied, including any indirect costs, but excluding any profit, margin, mark-up, or other fees.
- 1.4 "Additional Beams" has the meaning set forth in Article 2.1(B).
- 1.5 "Affiliate" means, with respect to an entity, any other entity, directly or indirectly, Controlling or Controlled by or under common Control with such first named entity.
- 1.6 "Anomaly" means, with respect to a Satellite in-orbit, a known condition or occurrence that has or reasonably can be predicted to have an adverse impact on the Stated Life or performance of the Satellite, and includes those items listed in paragraph 2.5.8.2(f) of the SOW.
- 1.7 "Article" means an article of this Contract.
- 1.8 "Attachment(s)" means any and all attachment(s) that are attached hereto or to any Exhibit and incorporated herein or therein, as may be amended from time to time in accordance with the terms hereof.
- 1.9 "[***]" has the meaning set forth in Article 4.6.
- 1.10 "Business Day" means any day other than a Saturday, Sunday or any other day on which national banks are authorized to be closed in New York City, New York.
- 1.11 "Canadian Beams" has the meaning set forth in Article 2.1(B).
- 1.12 "Candidate Launch Vehicles" has the meaning set forth in Article 3.3.
- 1.13 "[***]" has the meaning set forth in Article 2.1(B).
- 1.14 "[***]" has the meaning set forth in Article 4.6.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

- 1.15 "Certification" has the meaning set forth in Article 9.1.6.
- 1.16 "Collateral" has the meaning set forth in Article 41.1.1.
- 1.17 "Competitor" means the following entities or any successors or assigns thereof that are in direct competition with Contractor for the manufacture and sale of commercial satellites: Orbital Sciences Corporation; Lockheed Martin Commercial Space Systems; Boeing Satellite Systems; Thales Alenia Space; EADS Astrium; General Dynamics Spectrum Astro Space Systems; Mitsubishi Electric Corp. (MELCO); and Northrop Grumman Space Technology.
- 1.18 "Component" means any unit, system, subsystem, or piece of equipment or hardware or software to be employed on a Satellite.
- 1.19 "CondoSat Agreement" means the agreement between ViaSat, Inc. and Loral Space & Communications, Inc., dated of even date hereof.
- 1.20 "CondoSat Associate" means Loral Space & Communications, Inc., and any permitted assigns or successors in interest thereto under the CondoSat Agreement.
- 1.21 "CondoSat Associate Portion" has the meaning set forth in Article 4.1.
- 1.22 "Configuration Period" has the meaning set forth in Article 2.1(B).
- 1.23 "Configuration Process" has the meaning set forth in Article 2.1(B).
- 1.24 "Contract" means the terms and conditions (Preamble, Recitals and Articles) of this executed contract, its Exhibits and its Attachment(s) as set forth in Articles 2.1 and 33, as may be amended from time to time in accordance with the terms hereof.
- 1.25 "Contractor" has the meaning set forth in the Preamble and any successor or assignee permitted hereunder.
- 1.26 "Contractor Intellectual Property" shall mean such Intellectual Property owned by Contractor and provided to Purchaser related to this Contract (before or after EDC) and all Intellectual Property Rights related thereto. Contractor Intellectual Property shall also include any derivatives, improvements, or modifications made by Purchaser or Contractor thereto, except for derivatives, improvements, or modifications that can be used by Purchaser without infringing or violating the pre-existing Intellectual Property Rights of Contractor.



- 1.27 "Contract Technology" means all: (a) Software (in object code and source code format unless otherwise specified); and (b) data and documentation required for the performance, maintenance, or operation of any Deliverable Item.
- 1.28 "Control" and its derivatives mean, with respect to an entity, (i) the legal, beneficial, or equitable ownership, directly or indirectly, of fifty percent (50%) or more of the capital stock (or other ownership interest if not a corporation) of such entity ordinarily having voting rights, or (ii) the power to direct, directly or indirectly, the management policies of such entity, whether through the ownership of voting stock, by contract, or otherwise.
- 1.29 "Cure Letter" has the meaning set forth in Article 9.1.6.
- 1.30 "Daily Rate" has the meaning set forth in Article 13.
- 1.31 "Defect" means (a) with respect to the Satellite, the DSS, the Satellite Control Facility Equipment (if the options therefor are exercised), and the Spares, any failure to meet the applicable specifications of this Contract, including failures due to non-compliant materials or workmanship, or (b) with respect to any Deliverable Data, Software, or Deliverable Services, any failure to meet the applicable requirements of this Contract.
- 1.32 "Deliverable Data" means the data and documentation required to be delivered to Purchaser as specified in Article 3.1.
- 1.33 "Deliverable Item" means any of the items or services listed in Article 3.1, as may be amended from time to time in accordance with the terms hereof and, collectively, the "Deliverable Items."
- 1.34 "Deliverable Services" means those services listed in Article 3.1, as may be amended from time to time in accordance with the terms hereof.
- 1.35 "Delivery" has the meaning set forth in Article 3.2.
- 1.36 "Dispute" has the meaning set forth in Article 25.
- 1.37 "Dynamic Satellite Simulator" or "DSS" means the dynamic satellite simulator that is to be delivered to Purchaser pursuant to this Contract.
- 1.38 "Effective Date of Contract" or "EDC" has the meaning set forth in the Preamble.
- 1.39 "Escrow Agreement" has the meaning set forth in Article 29.1.



- 1.40 "Eutelsat" has the meaning set forth in Article 4.2.
- 1.41 "Exhibit(s)" means the exhibit(s) identified in Article 2.1 and attached hereto and incorporated herein, as may be amended from time to time in accordance with the terms hereof.
- 1.42 "Exploit" means with regard to Contractor Intellectual Property to use such Intellectual Property (1) to launch, use, operate and maintain the Satellite, and to sell, offer for sale, lease, market and rent capacity on, market and provide services with respect to, and otherwise obtain the benefits of, the Satellite, and, (2) to use, operate, reproduce, and maintain, as applicable, Deliverable Items in carrying out Purchaser's business related to the Satellite.
- 1.43 "FCC" means the Federal Communications Commission or any successor agency or governmental authority.
- 1.44 "Final Configuration" has the meaning set forth in Article 2.1(B).
- 1.45 "Final Orbit Location" has the meaning set forth in Article 2.1(B).
- 1.46 "Final Statement" has the meaning set forth in Article 21.1.3.
- 1.47 "Financing Entity" means any entity (other than Contractor or parties related to Contractor or a Competitor of Contractor), which has been specifically identified in a written notification to Contractor providing money to Purchaser to fund any portion of the Satellite or the Work.
- 1.48 "Firm Fixed Price" has the meaning set forth in Article 4.1.
- 1.49 "Force Majeure" has the meaning set forth in Article 17.
- 1.50 "[***]" has the meaning set forth in Article 35.2.
- 1.51 "Ground Insurance" has the meaning set forth in Article 32.5.
- 1.52 "Ground Storage" means that period where the Satellite is held on the ground for an extended period prior to Launch.
- 1.53 "In-Orbit Testing" or "IOT" means the testing of the Satellite in-orbit in accordance with Exhibit D, Satellite Program Test Plan.
- 1.54 "IOT Complete Date" has the meaning set forth in Article 10.3.

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



- 1.55 "Integration Activities" has the meaning set forth in Article 3.3.
- 1.56 "Intellectual Property" means all designs, techniques, analyses, methods, concepts, formulae, layouts, software, inventions (whether or not patented or patentable), discoveries, improvements, processes, ideas, technical data and documentation, technical information, engineering, manufacturing and other drawings, specifications and similar matter in which an Intellectual Property Right subsists, regardless of whether any of the foregoing has been reduced to writing or practice.
- 1.57 "Intellectual Property Claim" has the meaning set forth in Article 19.
- 1.58 "Intellectual Property Right(s)" means all common law and statutory proprietary rights with respect to Intellectual Property, including patent, patent application, copyright, trademark, service mark, trade secret, mask work rights, data rights, moral rights, and similar rights existing from time to time under the intellectual property laws of the United States, any state or foreign jurisdiction, or international treaty regime, regardless of whether such rights exist as of the date hereof or arise or are required at any time in the future.
- 1.59 "Intentional Ignition" means, with respect to the Satellite, the start of the ignition process of the Launch Vehicle for the purpose of Launch, which is the time at which the command signal is sent to the Launch Vehicle. This definition shall be modified to reflect the definition of "intentional ignition" in the Launch Services Agreement applicable to Launch of the Satellite.
- 1.60 "Intermittent Failure" means a Transponder Failure for a period in excess of [***] continuous minutes or [***] cumulative minutes in any day during the Orbital Performance Incentive Period.
- 1.61 "Junior Lien" has the meaning specified in Article 41.1.4.
- 1.62 "ITAR" means 22 C.F.R. Sections 120 through 130, International Traffic in Arms Regulations, as amended.
- 1.63 "Key Milestone Date(s)" has the meaning set forth in Article 22.1.
- ** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



- 1.64 "Key Personnel" has the meaning set forth in Article 37.2.
- 1.65 "Launch" means, with respect to the Satellite, Intentional Ignition followed by lift-off. This definition shall be modified to incorporate the definition of "launch" from the Launch Services Agreement applicable to the Launch of the Satellite.
- 1.66 "Launch and In-Orbit Insurance Policy" has the meaning set forth in Article 32.2.
- 1.67 "Launch Agency" means the provider of Launch Services responsible for the Launch of the Satellite.
- 1.68 "Launch Services" means those services provided by the Launch Agency for the Launch of the Satellite pursuant to the Launch Services Agreement.
- 1.69 "Launch Services Agreement" means the contract between the Purchaser and the Launch Agency that provides Launch Services for the Satellite.
- 1.70 "Launch Site" means the location that will be used by the Launch Agency for purposes of launching the Satellite, except in the case of Sea Launch it shall mean the home port located in Long Beach, CA.
- 1.71 "Launch Support" or "Launch Support Services" means those services specified in Exhibit A, Statement of Work to be provided by Contractor in support of Launch Services.
- 1.72 "Launch Vehicle" means the launch vehicle used to provide Launch Services for the Satellite.
- 1.73 "Loral Space Payload" means that portion of the Satellite payload consisting of nine (9) Canadian Beams.
- 1.74 "Loral Space Payload Contract" means the contract between Contractor and CondoSat Associate for provision of the Loral Space Payload, dated of even date hereof.
- 1.75 "Losses" has the meaning set forth in Article 20.1.
- 1.76 "Major Subcontract" has the meaning set forth in Article 38.1.
- 1.77 "Major Subcontractor" means a Subcontractor who is a party to a Major Subcontract.
- 1.78 "Milestone" means a portion of the Work upon completion of which a payment is to be made in accordance with Exhibit E, Payment Plan and Termination Liability



- Schedule. Milestone completion criteria and processes are as set forth in Exhibit A, Statement of Work and/or Exhibit E, Payment Plan and Termination Liability Schedule, as applicable. Orbital Performance Incentives shall not constitute a Milestone.
- 1.79 "Mission Assurance Plan" means the mission assurance plan attached as Exhibit C, as may be amended from time to time in accordance with the terms hereof.
- 1.80 "Mission Operations Support Services" means the orbit-raising, IOT and related services specified in Exhibit A, Statement of Work, to be performed by Contractor for the Satellite.
- 1.81 "Notice of Non-Conformance" has the meaning set forth in Article 9.1.6.
- 1.82 "NSP" means "not separately priced" and included in the Firm Fixed Price.
- 1.83 "OML Payment" has the meaning set forth in Article 36.
- 1.84 "Option Satellite" has the meaning set forth in Article 2.2.
- 1.85 "Option Satellite Exercise Period" has the meaning set forth in Article 2.2.
- 1.86 "Orbital Performance Incentive Period" means, with respect to the Satellite, the period commencing at (and including) 12:01 a.m. Greenwich Mean Time on the first day following the IOT Complete Date for such Satellite and ending on the last day of the Stated Life.
- 1.87 "Orbital Performance Incentives" means the amount set forth in Article 13.2, which may be earned by Contractor based on in-orbit performance of such Satellite as set forth in Article 13.
- 1.88 "Orbital Storage" means, with respect to the Satellite, any period of time of intentional non-use by Purchaser of such Satellite after the IOT Complete Date, provided that such Satellite has been placed into orbit and is capable of performing in accordance with Exhibit B, Satellite Performance Specification.
- 1.89 "Paragraph" means a paragraph under any Article hereof or section in an Exhibit or Attachment.
- 1.90 "Partial Loss" means, with respect to the Satellite on or after Intentional Ignition, that Transponder Failures have occurred, but the Satellite is not a Total Loss.
- 1.91 "Party" or "Parties" means Purchaser, Contractor or both, as the context requires.



- 1.92 "Payment Plan and Termination Liability Schedule" means the payment plan for the applicable Deliverable Item, and termination liability schedule for this Contract, as set forth in Exhibit E, Payment Plan and Termination Liability Schedule, as may be amended from time to time in accordance with the terms hereof.
- 1.93 **"Performance Specification"** means the applicable performance specification for the Satellite or other Deliverable Item, as appropriate, in the context of the applicable clause, as such specification may be amended from time to time in accordance with the terms hereof.
- 1.94 "Permitted Debt" has the meaning specified in Article 41.1.4.
- 1.95 "Preamble" means the preamble section of this Contract.
- 1.96 "Proceeds" has, for purposes of Article 41, the meaning ascribed to such term by the UCC.
- 1.97 "Proposal" has the meaning set forth in Article 16.
- 1.98 "Proprietary Information" has the meaning set forth in Article 28.
- 1.99 "Purchaser" has the meaning set forth in the Preamble and any successor or assignee permitted hereunder.
- 1.100 "Purchaser Delay" has the meaning set forth in Article 18.
- 1.101 "Purchaser Intellectual Property" shall mean such Intellectual Property owned by Purchaser and provided to Contractor related to this Contract (before or after EDC) and all Intellectual Property Rights related thereto. Purchaser Intellectual Property shall also include any derivatives, improvements, or modifications made by Purchaser or Contractor thereto, except for derivatives, improvements, or modifications that can be used by Contractor without infringing or violating the pre-existing Intellectual Property Rights of Purchaser.
- 1.102 "Reasonable Efforts" means standards, practices, methods, and procedures conforming to applicable law and that degree of effort, skill, diligence, prudence, and foresight that would reasonably and ordinarily be expected from a skilled and experienced commercial satellite contractor or commercial satellite owner, as the case may be.
- 1.103 "Recitals" means the recitals section of this Contract.



- 1.104 "Remaining OML" has the meaning set forth in Article 36.
- 1.105 "Replacement Satellite" has the meaning set forth in Article 2.3.
- 1.106 "Replacement Satellite Exercise Period" has the meaning set forth in Article 2.3.
- 1.107 "Satellite" means the communications satellite that is to be manufactured by Contractor and to be delivered to Purchaser pursuant to this Contract, and includes the Loral Space Payload. The Satellite is also referred to as the "Primary Satellite."
- 1.108 "Satellite Beams" has the meaning set forth in Article 2.1(B).
- 1.109 "SCF" means the Satellite Control Facility, which includes the Spares, as defined in Exhibit F.
- 1.110 "Satellite Control Facility Equipment" or "SCF Equipment" means those items listed in section 3.1 of Exhibit F, Satellite Control Facility Requirement Specification.
- 1.111 "Satellite Payload Specifications" means the payload portion of the Satellite Performance Specifications set forth in Section 3 of Exhibit B hereto.
- 1.112 "Satellite Performance Specification" means the Satellite performance specification attached as Exhibit B, as may be amended from time to time in accordance with the terms hereof.
- 1.113 "Satellite Pre-Shipment Review" or "SPSR" has the meaning set forth in Article 9.
- 1.114 "Satellite Program Test Plan" means the Satellite program test plan attached as Exhibit D, as may be amended from time to time in accordance with the terms hereof.
- 1.115 "Secured Obligations" means, collectively, all present and future obligations of the Contractor to or for the benefit of Purchaser under this Contract, as well as all damages to which Purchaser may be entitled as a result of any breach of this Contract. All such obligations are "Secured Obligations".
- 1.116 "Security Interest" has the meaning set forth in Article 41.1.1.
- 1.117 "Selection Criteria" has the meaning set forth in Article 2.1(B).
- 1.118 "Software" means the machine readable computer programs (including firmware, files, databases, interfaces, documentation and other materials related thereto, and any third party Software sublicensed by Contractor hereunder), as such Software is



revised, upgraded, updated, corrected, modified, and enhanced from time-to-time and provided to Purchaser pursuant to this Contract.

- 1.119 "Spare" means any spare part to be delivered with the Satellite Control Facility Equipment.
- 1.120 "Stated Life" means, with respect to the Satellite, the contracted for life of fifteen (15) years for such Satellite, commencing upon the IOT Complete Date for such Satellite.
- 1.121 "Statement of Work" or "SOW" means the statement of work attached as Exhibit A, as may be amended from time to time in accordance with the terms hereof.
- 1.122 "Storage Costs" has the meaning set forth in Article 35.2.
- 1.123 "Subcontract" means a contract or purchase order awarded by Contractor to a Subcontractor or a contract or purchase order awarded by a Subcontractor at any tier for performance of any of the Work.
- 1.124 "Subcontractor" means any person or business entity that has been awarded a Subcontract.
- 1.125 **"Substantially Similar Satellite"** means a satellite that is identical to the Satellite (but does not include the nine (9) Canadian Beams of the Loral Space Payload and, further, CondoSat Associate will have no right, title or interest in such Substantially Similar Satellite) in the following parameters:
 - number of user beams;
 - number of gateway beams;
 - antenna assembly quantity, aperture size and optics;
 - antenna feed design types;
 - antenna allowable interface to the rest of the Satellite bus (except for RF pathway connection to/from repeater);
 - repeater (same TWTA quantity and types, same LNA quantity, etc.);
 - frequency plan (same frequency converter and filter quantity and types); and
 - bus (solar array, battery, thermal, propulsion, autotrack, etc).

The following parameters may be different:

- CONUS orbital slot;
- CONUS user beam locations; and



SS/L-TP20701 ViaSat Contract

CONUS gateway beam locations.

For the Substantially Similar Satellite, Contractor will optimize EIRP, G/T and C/I performance with the same types of repeater and antenna Component design corresponding to the new orbital slot, new coverage and new beam plan, which may result in performance difference from the Satellite.

- 1.126 "[***] Coverage" has the meaning set forth in Article 2.1(B).
- 1.127 "TCC" has the meaning set forth in Article 38.3.
- 1.128 "Technical Assistance Agreement" has the meaning set forth in section §120.22 of the U.S. International Traffic in Arms Regulations, 22 CFR §120 130.
- 1.129 "**Terminated Ignition**" means that, following the time when the electronic signal is sent to command the opening of any first stage propellant valves, the first stage engines of the Launch Vehicle shut down for any reason before the hold down mechanism is released and the Launch pad is declared safe by the Launch Agency. This definition shall be modified to incorporate the definition of "terminated ignition" (or other similar term) from the Launch Service Agreement applicable to the Launch of the Satellite.
- 1.130 "Total Combined Price" has the meaning set forth in Article 4.1.
- 1.131 "**Total Loss**" means with respect to the Satellite on or after Intentional Ignition (i) the complete loss, destruction or failure of such Satellite, or (ii) as defined in Purchaser's Launch and In-Orbit Insurance Policy if defined differently therein.
- 1.132 "Training" means the training to be provided by Contractor in accordance with Exhibit A, Statement of Work.
- 1.133 "**Transponder**" means any one of the primary designated Ka-Band transponders specified in Exhibit B, Satellite Performance Specification, including both forward and return transponders. A forward transponder receives signals from a single gateway location and radiates those signals to a single user beam. A return transponder receives signals from a single user beam and radiates those signals to a single gateway location.
- Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

- 1.134 "Transponder Failure" means, with respect to a forward Transponder, the failure (including permanent or Intermittent Failures) to meet: (a) the G/T value for the applicable gateway location, as set forth in Table 2 of Exhibit B, Satellite Performance Specification; or (b) the EIRP value for the applicable beam, as set forth in Table 3 of Exhibit B, Satellite Performance Specification. With respect to a return Transponder, "Transponder Failure" means the failure (including permanent or Intermittent Failures) to meet: (a) the EIRP value for the applicable gateway location, as set forth in Table 2 of Exhibit B, Satellite Performance Specification; or (b) the G/T value for the applicable beam, as set forth in Table 3 of Exhibit B, Satellite Performance Specification. When a Transponder is operating with redundant and/or spare components and meets the EIRP and G/T values set forth above, it shall not be considered to be experiencing a Transponder Failure.
- 1.135 "TWTs" has the meaning set forth in Article 38.3.
- 1.136 "UCC" means the Uniform Commercial Code as in effect in the State of New York, as it may be amended from time to time.
- 1.137 "U.S. Beams" has the meaning set forth in Article 2.1(B).
- 1.138 "Work" means all design, development, construction, manufacturing, labor, and services, including tests to be performed, and any and all Deliverable Items, including the Satellite, the DSS, Deliverable Data, Mission Operations Support Services, Launch Support Services, Training, and equipment, materials, articles, matters, services, and things to be furnished to Purchaser under this Contract.
- 1.139 "Work-in-Process" means the following goods, services, and rights to be provided to Purchaser by Contractor under this Contract but in the case of goods only such goods as have been designated for use under this Contract under Contractor's internal material resource planning system and have been installed: (a) the Satellite, (b) the DSS, (c) the Satellite Components, (d) all other Deliverable Items, including items purchased pursuant to exercised options set forth herein, (e) all parts, materials, inventories, and associated warranties, and (f) the rights in Intellectual Property as set forth in Article 39, and Proprietary Information as set forth in Article 28. The foregoing shall constitute Work-in-Process as the same shall be in the process of performance, manufacture, testing, integration, delivery or completion at any given point in time.





ARTICLE 2 — SCOPE OF WORK

2.1 Provision of Services and Materials

A. General

Contractor shall provide Purchaser with the Deliverable Items completed in all respects in accordance with the provisions of this Contract. Without limiting the generality of the foregoing, the Contractor shall provide the necessary personnel, material, services, and facilities to design, manufacture, test and ship the Satellite, together with all other Deliverable Items referred to in Article 3.1. Contractor shall furnish and perform the Work in accordance with the provisions of this Contract, including the following Exhibits, which are attached hereto and made a part hereof:

- 2.1.1 Exhibit A, Statement of Work, dated January 7, 2008
- 2.1.2 Exhibit B, Satellite Performance Specification, dated January 7, 2008, including all Annexes thereto
- 2.1.3 Exhibit C, Mission Assurance Plan, dated September 22, 2006
- 2.1.4 Exhibit D, Satellite Program Test Plan, dated January 7, 2008
- 2.1.5 Exhibit E, Payment Plan and Termination Liability Schedule, dated January 7, 2008
- 2.1.6 Exhibit F, Satellite Control Facility Requirements Document, dated January 7, 2008
- 2.1.7 Exhibit G, Escrow Agreement
- 2.1.8 Exhibit H, Dynamic Satellite Simulator Statement of Work and Functional Requirements Document dated January 7, 2008
- 2.1.9 Exhibit I, Guaranty Agreement
- 2.1.10 Exhibit J, [***] Escrow Agreement
- 2.1.11 Exhibit K, Default Configuration

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

B. Finalization of the Satellite Configuration

As of EDC, the Satellite configuration has not been finally defined, but the Parties agree it will consist of nine (9) Canadian service beams with three 2 GHz fixed gateway locations (provided that in the event Purchaser and CondoSat Assocaite share a gateway under section 2.8 of the CondoSat Agreement, then there shall only be two 2 Ghz gateways) ("Canadian Beams"), sixty (60) U.S. service beams with thirteen 2 Ghz fixed gateway locations and three 1.5 Ghz fixed gateway locations ("U.S. Beams") (all of the foregoing as set forth in Exhibit K, Default Configuration) plus, potentially, three (3) additional service beams and one (1) associated gateway ("Additional Beams"), the location of such Additional Beams to be established as set forth below. Through February 3, 2008 (the "Configuration Period"), the Parties will work together to complete an optimal Satellite configuration for Purchaser (the "Configuration Process"), consisting of the following:

(i) The Parties shall use reasonable best efforts during the Configuration Period to develop an optimal configuration of the U.S. Beams, the Canadian Beams and Additional Beams (together, the "Satellite Beams") for Purchaser. The objectives in developing the optimal Satellite configuration for Purchaser shall, among other things, include: (1) calculated [***] (as calculated using the [***] defined in Article 4.6) equal to or higher than [***] (not including the Canadian Beams); (2) user beams located in coverage areas which are, in Purchaser's reasonable opinion, desirable for its intended services; (3) gateway locations with reasonable economical access to sufficient existing fiber backhaul; (4) limited overlap between the user beams; (5) complete inter-beam coverage for user beams (no "[***] Coverage" as exemplified in Attachment A hererto, unless otherwise approved by Purchaser); and (6) optimal number of gateways for Purchaser's business plan, but not less than the number of gateways reflected in Exhibit K, Default Configuration (items (1) through (6) immediately above the "Selection Criteria"). The Parties shall at a minimum complete the following actions in an effort to develop a final Satellite configuration for Purchaser:

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

- a. On or before 4, January 2008, Contractor will submit to Purchaser the antenna performance files, beam laydown (including coverage map, antenna-beam mapping, and frequencies assigned to each beam) and gateway location information ("[***]") defined in Annex B to Exhibit B, Satellite Performance Specification, separately for; (i) the Canadian Beams; (ii) the U.S. Beams; and (iii) multiple configurations of the Additional Beams (consistent with the criteria below). In the event Contractor proposes a Satellite configuration to Purchaser that leads to any [****] ("[****] Coverage"), Purchaser will notify Contractor (and provide Contractor with reasonable guidance on rectifying such [****] Coverage), and Contractor will adjust the proposed Satellite configuration in order to [****] as part of the process described below (even if removing such [****] reduces Satellite [****]).
- b. On or before 8 January 2008, Purchaser will provide Contractor with initial comments and feedback on Contractor's proposed configuration for the Satellite Beams (including alternative candidate locations in the US and/or Mexico for the Additional Beams).
- c. On or before 11 January 2008, Purchaser shall identify up to three (3) alternative Satellite configurations for the Additional Beams to be considered by Purchaser and Contractor for the final Satellite configuration.
 - The locations to be considered by the Purchaser for location of the user beams for the Additional Beams are, in order of priority:
 - Major metropolitan areas of the central U.S. not covered by the U.S. Beams, including locations in Denver, Colorado, Salt Lake City, UT, and Phoenix/Tucson AZ;
 - Mexico; and
 - overlay on three (3) of the sixty (60) U.S. service beams.
- d. On or before 18 January 2008, Contractor shall provide [***] for two (2) of the three (3) alternative configurations identified by Purchaser

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



- for the Additional Beams and within seven (7) days thereafter shall provide [***] for the final alternative configuration identified by Purchaser for the Additional Beams as provided above, for assessment of [***] (using the [***]).
- e. On or before 25 January 2008, Purchaser and Contractor will determine if any further improvements to the proposed Satellite configurations should be considered, and, provided that Contractor is reasonably able to provide updated [***] on such improvements by 31 January 2007, Contractor will update the [***] accordingly.
- f. On 3 February 2008, Purchaser shall select the final Satellite configuration ("Final Configuration") from among the above proposed Satellite configurations (including Exhibit K, Default Configuration) in accordance with the following principles:
 - In the event Purchaser selects a Satellite configuration consisting of only the U.S. Beams and the Canadian Beams, Purchaser shall select the Satellite configuration based on the Selection Criteria and, which in Purchaser's reasonable judgment optimizes its business prospects, while also considering in good faith the configuration that maximizes the calculated [***] (as calculated using the [***] defined in Article 4.6). In the event that Purchaser chooses a configuration that does not provide the maximum calculated [***] as provided above, Purchaser shall provide Contractor with the rationale therefor, which rationale shall be consistent with the optimization of Purchaser's business prospects and the Selection Criteria. For the purpose of clarification, in the event that Purchaser reasonably determines that the benefits associated with the Additional Beams does not exceed the costs associated with the Additional Beams, Purchaser may elect to reject the Additional Beams and select the Satellite configuration in accordance with the terms of this paragraph, and shall provide Contractor with the rationale therefor.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



- In the event Purchaser selects a Satellite configuration consisting of the U.S. Beams, the Canadian Beams and Additional Beams, Purchaser shall select the Satellite configuration which, in its reasonable judgment, optimizes its business prospects, while also considering in good faith the configuration that maximizes the calculated [***] (as calculated using the [****] defined in Article 4.6).
- In the case of the Canadian Beams, Purchaser shall elect the coverage locations which, in its reasonable judgment, optimizes the Canadian capacity/coverage area.
- (ii) In the event, for any reason, Contractor has not provided the [***] to Purchaser on or prior to the dates set forth above, except for delays excused due to Force Majeure events or Purchaser-caused delays, then Purchaser shall be entitled to select the Exhibit K Default Configuration consisting of only the U.S. Beams and the Canadian Beams (i.e., without the Additional Beams).
- (iii) Contractor acknowledges that, as part of Purchaser's finalization of the Satellite configuration, Purchaser will need to offset the station-keeping box from the 115.0 WL position (between 114.7 and 115.3 WL) (the "Final Orbit Location") so that it will not overlap the station-keeping box of any other satellite in accordance with FCC regulations. No later than twenty-one (21) days after EDC, Purchaser will provide the Final Orbit Location. For purposes of clarification, the [***] of the Final Configuration will be calculated pursuant to Article 4.6 assuming use of the 115.0 WL position, and the [***] calculated pursuant to Article 22.2.1 assuming use of the Final Orbit Location.
- (iv) Upon Purchaser selection of the Final Configuration: (1) the [***] of the Final Configuration will be calculated (using the [***]) and the Firm Fixed Price shall be subject to adjustment, up or down, pursuant to Article 4.6; and (2) Exhibit B, Satellite Performance Specification, will be updated accordingly, including the following:

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



- (A) Section 3.3, Coverage Area, will be updated with the resulting polygon area, gateway locations, and beam locations associated with the selected configuration;
- (B) Table 2 and Table 3, Gateway Locations and User Beam Locations, will be updated based upon the data files provided during the Configuration Process for the Final Configuration selected by Purchaser. The values incorporated into the tables shall be the values set forth in the data files, taking into account pointing error consistent with Section 3.6.1 of Exhibit B, Satellite Performance Specification, minus 0.5 dB; and
- (C) Tx and Rx performance (specifically, the EIRP, G/T, and C/I values specified in Tables 4, 5, 8, 9, 10, 11, 12, 13, 16, and 17) will be updated based upon the data files provided during the Configuration Process for the Final Configuration selected by Purchaser. The [***], which provides distributions on all relevant parameters, will be used to generate the cumulative distributions on the metrics set forth in sections 3.6.3, Forward Uplink (from Gateway), 3.6.4, Forward Downlink (to User), 3.6.5, Return Uplink (from User), and 3.6.6, Return Downlink (to Gateway).

2.2 Option Satellite

Contractor hereby grants to Purchaser an option to purchase one (1) additional Substantially Similar Satellite (including a DSS, updates to Deliverable Data, as required, Launch Support Services and Mission Operations Support Services) (the "Option Satellite") at the optional pricing set forth in the table below (to be established following the process defined in Article 2.1 above, and to reflect the adjustments necessary to convert from the Primary Satellite configuration to a sixty (60) U.S. beam only configuration, with such pricing to be provided no later than March 2, 2008). Delivery for the Option Satellite shall be no later than the later to occur of: (i) thirty six (36) months after execution of the second contract or Contract amendment identified below; and (ii) four (4) months after Delivery of the Primary Satellite. Commencing upon

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

the date of execution of such second contract or Contract amendment, Contractor shall immediately proceed with all Work necessary to build such Option Satellite.

The option shall remain valid from EDC until two (2) years from EDC (the "Option Satellite Exercise Period"), provided that Purchaser may require Contractor to extend the validity of the option and/or change the design and/or specifications of the Option Satellite under a change order pursuant to Article 16. If Purchaser wishes to exercise such option, it may do so by providing notice to Contractor at any time prior to the expiration of the Option Satellite Exercise Period. In such event, the Parties shall document Purchaser's agreement to purchase the Option Satellite and Contractor's agreement to build and deliver such satellite by entering into either, at Purchaser's option, an amendment to this Contract or a second contract that has terms and conditions identical in all relevant material respects to this Contract except for such differences as are reasonably necessary or appropriate to indicate that the second contract applies to the Option Satellite. Only upon execution of such second contract or Contract amendment shall Purchaser be financially obligated to Contractor with respect to the Option Satellite and shall Contractor be obligated to perform with respect thereto.

Satellite Option Exercised Within EDC + 6 months	Firm Fixed Price of Option Satellite(i) TBD
Between EDC + 6 months and EDC + 12 months	TBD
Between EDC + 12 months and EDC + 18 months	TBD
Between EDC + 18 months and EDC + 24 months	TBD



SS/L-TP20701 ViaSat Contract

(i) [***]% of the Firm Fixed Price of Option Satellite constitutes the maximum Orbital Performance Incentives earnable for the Option Satellite. Firm Fixed Price of Option Satellite includes Deliverable Data, Training, Launch Support Services (as they may be adjusted pursuant to Article 4.3), Mission Operations Support Services, and a DSS.

2.3 Replacement Satellite

Contractor hereby grants to Purchaser an option to purchase one (1) replacement satellite identical to the Primary Satellite provided hereunder (including a DSS, updates to Deliverable Data, as required, Launch Support Services and Mission Operations Support Services) (the "Replacement Satellite"). The price for such Replacement Satellite shall be [***] U.S. dollars (\$[***]) as adjusted pursuant to Article 4.6 ([***] percent ([***]%) of which constitutes the maximum Orbital Performance Incentives Contractor may earn with respect to the Replacement Satellite) and apportioned as follows: (i) [***] U.S. dollars (U.S.\$ [***] is the portion of the Replacement Satellite price to be paid by Purchaser hereunder; and [***] U.S. dollars (U.S.\$[***]) is the portion of the Replacement Satellite to be paid by [***]. Delivery for the Replacement Satellite shall be no later than the later to occur of: (i) thirty-four (34) months after execution of the second contract or Contract amendment identified below and (ii) four (4) months after Delivery of the Satellite). Commencing upon the date of execution of such second contract or Contract amendment, Contractor shall immediately proceed with all Work necessary to build such Replacement Satellite. The option shall remain valid until one (1) year from Delivery of the Satellite (the "Replacement Satellite Exercise Period"), provided that Purchaser may require Contractor to extend the validity of the option and/or change the design and/or specifications of the Replacement Satellite under a change order pursuant to Article 16. If Purchaser wishes to exercise such option, it may do so by providing notice to Contractor at any time prior to the expiration of the Replacement Satellite Exercise Period. In such event, the Parties shall document Purchaser's agreement to purchase the Replacement Satellite and Contractor's agreement to build and deliver such satellite by entering into either, at Purchaser's option, an amendment to this Contract or a second contract that has terms and conditions identical in all relevant material respects to this Contract except for such differences as are reasonably necessary or appropriate to indicate that the second contract applies to the Replacement Satellite. Only upon execution of such second contract or Contract amendment shall Purchaser be financially obligated to Contractor

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

with respect to the Replacement Satellite and shall Contractor be obligated to perform with respect thereto.

2.4 SCF Equipment

Contractor hereby grants to Purchaser an option to purchase the SCF Equipment (including the Spares) listed in Exhibit F, Satellite Control Facility Requirement Specification. The Parties acknowledge that, as of EDC, the Parties are updating Exhibit F, Satellite Control Facility Requirements Document, to be consistent with SCF requirements agreed-to between the Parties, and which updates are within the agreed scope of Work hereunder so as not to constitute a change hereunder; the Parties will continue to work in good faith to complete the revisions to Exhibit F, Satellite Control Facility Requirements Document no later than EDC plus thirty (30) days. The price for such SCF Equipment shall be [***] U.S. dollars (\$[****]). This option shall remain valid until 6 months from EDC (the "SCF Equipment Exercise Period"), provided that Purchaser may require Contractor to extend the validity of the option and/or change the design and/or specifications of the SCF Equipment under a change order pursuant to Article 16. If Purchaser wishes to exercise such option, it may do so by providing notice to Contractor at any time prior to the expiration of the SCF Equipment Exercise Period. Delivery for the SCF Equipment shall be no later than August 1, 2010. Payment for the SCF Equipment shall be due upon Acceptance of the SCF Equipment.

2.5. Cooperation with Launch Agency

Contractor shall provide all reasonably necessary assistance to, and shall fully communicate and cooperate with, the Launch Agency so as to ensure successful, on-time completion of the Work and integration of the Satellite with the Launch Vehicle in accordance with the terms of this Contract and the Launch Services Agreement. Purchaser shall provide all reasonably necessary assistance to Contractor so as to ensure Contractor and any affected Subcontractors have the necessary access and information from the Launch Agency to perform as specified. All communications of Contractor and its Subcontractors with the Launch Agency are subject to any required export authorizations.

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

2.6 Relationship to Loral Space Payload Contract

Except as to certain rights related to refunds, termination, access and title transfer associated with the Loral Space Payload Contract, Purchaser shall have the exclusive rights (including all exclusive rights of exercise) associated with the Satellite and other Deliverable Items to be provided hereunder. CondoSat Associate shall have no right or authority to give Contractor direction as to the Satellite, including as to the Loral Space Payload, and the Loral Space Payload Contract shall reflect that CondoSat Associate agrees to, and shall be bound by, all acts and decisions of the Purchaser under this Contract as regards the Satellite, including the Loral Space Payload. Purchaser agrees that any action related to the Satellite, including the Loral Space Payload portion thereof, other than as regards payment and refunds, shall be the responsibility of Purchaser and Purchaser shall act on behalf of Purchaser and CondoSat Associate, with no direct right of action or enforcement against Contractor by CondoSat Associate. Purchaser retains all rights associated with Deliverable Items 2 through 6 set forth in Article 3.1, except that CondoSat Associate shall be entitled to copies of the Deliverable Data for information purposes only, and subject to restrictions on use and disclosure substantially similar to those contained herein. Contractor agrees to place the Loral Space Payload Contract on terms reflecting the foregoing.



SS/L-TP20701 ViaSat Contract

ARTICLE 3 — DELIVERABLE ITEMS AND DELIVERY SCHEDULE

3.1 <u>Deliverable Items</u>

Subject to the other terms and conditions of this Contract, the items to be delivered under this Contract are specified below (each a "Deliverable Item"). Contractor shall deliver such Items on or before the corresponding Delivery dates and at locations specified as follows:

Item 1.	Description Satellite	Delivery Date Delivery to the Launch Site on EDC + 36 months	Delivery Location Per Article 3.2
2.	Deliverable Data	Per SOW, Exhibit A	Per SOW, Exhibit A
3.	Training	Per SOW, Exhibit A	Per SOW, Exhibit A
4.	Launch Support Services	Per SOW, Exhibit A	Per SOW, Exhibit A
5.	Mission Operations Support Services	Per SOW, Exhibit A	Per SOW, Exhibit A
6.	DSS	Per Exhibit H	Purchaser's SCF

3.2 Delivery

Delivery of each Deliverable Item shall occur upon arrival of such Deliverable Item at the location required by this Article 3, after having successfully completed any required reviews and testing. In the case of the Satellite, Delivery shall occur upon delivery of the Satellite to the Launch Site or upon placement in Ground Storage.

3.3 Selection of Launch Vehicle

Purchaser shall be responsible for providing the Launch Services for Launch of the Satellite(s). Contractor shall provide, at no additional cost to Purchaser, engineering, Launch Vehicle integration activities, and other customary services to maintain compatibility of the Satellite for Launch (including but not limited to environmental analysis, mission planning, and Launch Vehicle electrical and mechanical interfaces) (collectively, "Integration Activities") with the list of candidate Launch Vehicles set forth in Paragraph 2.4 of Exhibit B, Satellite Performance Specification (the "Candidate Launch



SS/L-TP20701 ViaSat Contract

Vehicles") up to eighteen (18) months prior to the scheduled Delivery date of the Satellite set forth in Article 3.1. On or before eighteen (18) months prior to the then-currently scheduled Delivery date of the Satellite set forth in Article 3.1, Purchaser shall provide a written notification to Contractor of two (2) Launch Vehicles that may be used for such Satellite. Provided that Purchaser has entered into the necessary agreements with the potential Launch Agencies to reasonably support Contractor's Integration Activities, Contractor shall perform dual Integration Activities up through twelve (12) months prior to the start of SPSR, at which time Purchaser shall designate its final Launch Vehicle selection. Failure to notify Contractor of its Launch Vehicle selection(s) by the date(s) set forth in this Paragraph shall be a Purchaser delay of Work subject to Article 18, and any change to the Launch Vehicle selection thereafter shall be subject to Article 16.

3.4 Avoidance and Mitigation of Delays

Contractor shall notify Purchaser promptly by telephone and confirm in writing any event, circumstance or development that will likely result in a material non-conformance of the Work or any part thereof with the requirements of this Contract or the Delivery dates established therefor. Contractor shall use Reasonable Efforts to avoid and/or mitigate the effect of such event, circumstance, or development. Within ten (10) Business Days of any notification hereunder, Contractor shall provide written notice to Purchaser of any affected Work, as well as a proposed work-around. The work-around plan shall: (a) set forth Contractor's Reasonable Efforts to mitigate the effect of any such event, circumstance, or development and include a schedule for such mitigation; and (b) contain sufficient detail for Purchaser to be able to evaluate such plan. If appropriate, such work-around plan shall use work-around schedules, payment of expedited fees, twenty four (24) hour operations, and the use of alternate Subcontractors (to be approved by Purchaser if required under this Contract).

3.5. Packing and Shipping

Packing and shipping shall be in accordance with all applicable laws, rules, and regulations, and standard commercial practices in the aerospace industry.



SS/L-TP20701 ViaSat Contract

ARTICLE 4 — PRICE

4.1 Firm Fixed Price

The total price to be paid by Purchaser and CondoSat Associate to Contractor for the Work, including but not limited to Deliverable Items 1 through 6 set forth in Article 3.1 hereof, as detailed in Exhibit A, Statement of Work, shall be a total firm fixed price of two hundred and forty-six million U.S. dollars (U.S. \$246,000,000) (the "Total Combined Price") which is apportioned as follows: (i) two hundred and nine million, one hundred thousand U.S. dollars (U.S.\$209,100,000) is the portion of the Total Combined Price to be paid by Purchaser hereunder (the "Firm Fixed Price"); and thirty-six million, nine hundred thousand U.S. dollars (U.S.\$36,900,000) is the portion of the Total Combined Price for which CondoSat Associate is responsible pursuant to the Loral Space Payload Contract (the "CondoSat Associate Portion"). Both the Total Combined Price and the Firm Fixed Price are subject to the [***] adjustment set forth in Article 4.6, below (however, the price of the Loral Space Payload Contract is not subject to adjustment pursuant to Article 4.6 below). Purchaser does not guarantee payment of the CondoSat Associate Portion, and Contractor agrees to look solely and exclusively to CondoSat Associate for payment of the CondoSat Associate Portion. Failure of CondoSat Associate to make any payment of the CondoSat Associate Portion when due shall not be considered a breach or default of this Contract and Contractor shall nonetheless remain obligated to perform its obligations and duties under this Contract. Contractor acknowledges and agrees that Purchaser shall have no liability to Contractor shall have no recourse whatsoever against Purchaser in such case.

Except as otherwise expressly provided in Articles 2.2 and 2.3 (only if Purchaser elects to effectuate its purchase of the Option Satellite or the Replacement Satellite by means of an amendment to this Contract rather than by means of another contract), and Articles 2.4, 4.2, 4.3, 4.6, 6.2, 6.3, 6.5, 12.1(B), 15.4, 16, 18, 35, and any options set forth in this Contract, the Firm Fixed Price is not subject to any escalation or to any adjustment or revision. The price for those items subject to an option under this Contract, if any, are described in the particular Articles that set forth such options.

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

The itemization of the Firm Fixed Price is as follows:

<u>Item</u> 1.	Description Satellite (excludes the Loral Space Payload price and Orbital Performance Incentives)	Amount [***] percent ([***]%) of the Firm Fixed Price (as of EDC, this amount is U.S. \$ [***])
1(a).	Maximum Orbital Performance Incentives	[***] percent ([***]%) of the Firm Fixed Price (as of EDC, this amount is U.S. $[***]$)
2.	Deliverable Data	NSP
3.	Training	NSP
4.	Launch Support Services	NSP(i)
5.	Mission Operations Support Services	NSP
7.	DSS	NSP
	Total Firm Fixed Price	U.S. \$209,100,000 (ii)

(i) The price for the Launch Support Services shall be modified by the amount set forth for such services in Article 4.3, based upon Purchaser's selection of Launch Vehicle for the Launch of the Satellite.

(ii) Not including the price for any incremental Launch Support Services per Article 4.3, or long-term Ground Storage per Article 35. Firm Fixed Price includes SPT thrusters, but does not include the Caribou command link (TT&C) encryption or Ferrite switches. Purchaser may elect to include the Caribou command link (TT&C) encryption at the price of \$[****], so long as Purchaser provides written notice of its election no later than February 3, 2008, and Purchaser may also elect to include Ferrite switches at the price of \$[****], so long as Purchaser provides written notice of its election no later than February 3, 2008. If Purchaser elects to include the Caribou command link (TT&C), then all unpaid Milestone payments associated with the Satellite shall automatically be increased by a pro rata amount such that the applicable increase is applied across all remaining, unpaid Milestone payments on the Satellite.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

4.2 Discount for Purchase of a Eutelsat Satellite

In the event Eutelsat or any of its Affiliates (Eutelsat and its Affiliates shall collectively be referred to as "Eutelsat") (or a joint venture consisting of Purchaser and Eutelsat) enters into a contract with Contractor for the purchase of a Ka-band spot beam satellite on or before 1 March 2008, ("Eutelsat Satellite"), the Firm Fixed Price for the Primary Satellite set forth above shall be reduced by [***] U.S. dollars (USD\$[***]). If Purchaser is entitled to such reduction, then all unpaid Milestone payments associated with the Satellite shall automatically be reduced by a pro rata amount such that the applicable discount is applied across all remaining, unpaid Milestone payments on the Satellite.

4.3 Launch Support Services

Upon Purchaser's selection of the Launch Vehicle for the Launch of the Satellite as provided in Article 3.3 above, the Firm Fixed Price of the Contract shall be increased or decreased in the following applicable amount to reflect the Launch Support Services associated with the selected Launch Vehicle. Such increase or decrease shall be applied on a pro-rata basis to those Milestone payments specified in Exhibit E, Payment Plan and Termination Liability Schedule remaining to be paid as of the date of Purchaser's selection.

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Selected Launch Vehicle	Adjustment to Firm Fixed Price
Atlas	(U.S. \$[***]([***])
Proton	U.S. \$[***]([***])
Sea Launch	(U.S. \$[***]) ([***])
Ariane	U.S. \$[***]([***])

4.4 Fees and Other Expenses

The Firm Fixed Price stated above includes all fees, charges, expenses, costs, and other amounts payable by Purchaser for any portion of the Work, including but not limited to the design, manufacturing, tests, the Satellite, the DSS, Deliverable Data, Training, Orbital Performance Incentives, Mission Operations Support Services, Launch Support Services, risk management services as required by Article 32, packing and transport of the Satellite to the Launch Site, transit insurance and such other insurance as is required by Article 32, but does not include such amounts payable for Launch Services, or the Launch and In-Orbit Insurance Policy (the responsibility for which shall reside exclusively with Purchaser). Under no circumstances will Purchaser be obligated to pay any fees, charges, expenses, costs or other amounts in connection with any portion of the Work other than the amounts set forth in this Paragraph, as adjusted in accordance with those Articles set forth in Article 4.1.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

4.5 Taxes

The Firm Fixed Price includes, and Contractor shall remit when due, all applicable federal, state, local and foreign taxes duties, assessments, and similar liabilities (including interest, fines, penalties, or additions attributable to, or imposed on, or with respect to, any such taxes, duties and similar liabilities) imposed by any federal, state, local, or foreign government in connection with the Work, including those related to the export or import of any Satellite from or into any jurisdiction for Launch, all imposts, and all sales, use, excise, value added, import and export taxes levied in connection with Contractor's performance of the Work, wherever the Work is being carried out under this Contract.

4.6 [***] Adjustment

Promptly upon finalization of the Satellite configuration pursuant to Article 2.1(B), Purchaser shall calculate, and Contractor shall have the opportunity to verify (pursuant to the licenses and limitations set forth in Article 22.2), the [***] of the U.S. Beams and the Additional Beams, if any, (but not including the Canadian Beams) (and the resulting calculated [***] shall be the "[***]") using the version of Purchaser's proprietary software program in existence as of EDC (operating in the mode that [***]) (the "[****]"); provided, however, that in the event that Purchaser elects to include Mexico coverage for any of the Additional Beams in the Final Configuration, then [***] contributed by such beam(s) will be considered in determining the [***], unless all necessary landing rights and any other rights necessary for Purchaser to offer its intended service in Mexico have been obtained. Contractor acknowledges that the Satellite Final Configuration [***] for the U.S. Beams and the Additional Beams will be measured assuming the use of the [***] frequencies and such [***] calculation shall not assume the use of the [***] frequencies. Based on that calculation, the [***] as follows (with such amounts to be pro-rated): [***] For example, if the [***]. By way of further example, if the [****]. All Milestones shall be updated on a pro-rata basis to account for [****].

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 5 — PAYMENTS

5.1 Payment Plan

Payments by Purchaser to Contractor of the Firm Fixed Price set forth in Article 4 and of the amounts for options, if any, exercised by Purchaser pursuant to this Contract shall be in accordance with Exhibit E, Payment Plan and Termination Liability Schedule, as applicable thereto.

5.2 Payment Conditions

- 5.2.1 <u>Time Payments</u>. The "Effective Date of Contract" time payment due from Purchaser shall be paid no later than the due date specified in Exhibit E, Payment Plan and Termination Liability Schedule. Contractor shall submit to Purchaser an invoice in accordance with the requirements of this Contract with respect to such payment prior to such payment being due.
- 5.2.2 <u>Milestone Payments</u>. Each Milestone payment specified in Exhibit E, Payment Plan and Termination Liability Schedule shall in each case become payable upon Contractor's completion of each Milestone in accordance with the Contract and satisfaction of the Conditions for Milestone Completion and Payment set forth in Exhibit E, after which Contractor shall submit an invoice for payment, provided however, that Contractor shall not invoice any amount which when cumulated with other Milestone payments previously made hereunder, exceeds the cumulative Milestone payment amounts due as of such date as reflected in such Exhibit E. A Milestone shall not be deemed completed until all Work relevant to the Milestone has been completed and documented in accordance with Exhibit A, Statement of Work and/or Exhibit E, Payment Plan and Termination Liability Schedule, as applicable.

Purchaser shall only be responsible for paying those amounts set forth in the "Firm Fixed Price (paid by ViaSat)" column of Exhibit E, Payment Plan and Termination Liability Schedule. All payments due from Purchaser upon the completion of a Milestone described in Exhibit E, Payment Plan and Termination Liability Schedule, shall be paid no later than thirty (30) days after the receipt by Purchaser of an invoice and certification in the form attached hereto as Attachment A that the Milestone has been completed in accordance with the



SS/L-TP20701 ViaSat Contract

requirements of this Contract, together with the necessary or appropriate supporting data and documentation as required hereunder, if any, or as Purchaser may reasonably request within ten (10) Business Days of receipt of invoice. Notwithstanding the foregoing, and without prejudice to Purchaser's rights under Article 5.6, Purchaser, in its sole discretion, may agree to make a partial payment to Contractor for partial completion of a Milestone or for completion of a Milestone prior to the applicable Milestone Date.

Purchaser shall pay in full all undisputed amounts payable hereunder within thirty (30) days after receipt of an invoice therefor completed in accordance with this Article 5.2.

- 5.2.3 <u>Orbital Performance Incentives</u>. The Orbital Performance Incentives payments due from Purchaser pursuant to Article 13 shall be due and payable no later than thirty (30) days after receipt of an invoice completed in accordance with Article 5.4.
- 5.2.4 Non-Warranty and Other Payments. Subject to Article 5.6, and without prejudice to Purchaser's audit rights set forth in Article 5.8, below, all amounts payable to Contractor with respect to non-warranty work performed pursuant to Article 15.4 or for any other payments to be made pursuant to this Contract and not otherwise subject to Article 5.2.1 through 5.2.3 above shall be paid no later than thirty (30) days after receipt of Contractor's invoice completed in accordance with Article 5 therefor.

5.3 Late Payment

5.3.1 Interest. Except in the case of a payment disputed pursuant to Article 5.6, below, in the event that any payment due under this Contract is not made when due hereunder, without prejudice to the other rights and remedies of the Party entitled to such payment, such Party shall also be entitled to interest at the rate of [***] percent ([***]%) per annum, compounded monthly, on the unpaid balance thereof from ten (10) days after the date that the non-paying Party receives written notice to cure a delinquent payment until such payment is made.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

5.4 Invoices

Invoices required to be delivered by Contractor hereunder shall be submitted by facsimile and air mail to Purchaser (original plus one (1) copy) at the following address:

	ViaSat, Inc.
Attn.:	David Abrahamian
	6155 El Camino Real
	Carlsbad, California 92009
Fax:	760-795-1045

or to such other address as Purchaser may specify in writing to Contractor.

5.5 Payment Bank

All payments made to Contractor hereunder shall be in U.S. currency and shall be made by electronic funds transfer to the following account:

SPACE SYSTEMS/LORAL, INC. ACCOUNT NO. [***]

or such other account or accounts as Contractor may specify in writing to Purchaser.

5.6 <u>Disputed Amounts</u>

If the event covered by a Milestone payment has not been completed in accordance with the requirements of this Contract, Purchaser shall so notify Contractor in writing within fifteen (15) days of receipt of the applicable invoice and may withhold some or all of the applicable payment in good faith. Such notification shall state in reasonable detail the Contract requirements associated with the applicable Milestone event that have not been met. Upon correction of the noted discrepancy(ies) and completion of the Milestone in accordance with Contract requirements, the Milestone invoice shall be reinstated for payment. Failure to pay any amount subject to a reasonable good faith dispute shall not constitute a material breach under this Contract until resolved by the Parties.

5.7 Right of Offset

In the event that one Party has not paid the second Party any amount that is due and payable to the second Party under this Contract, without prejudice to other rights and

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

remedies, such second Party shall have the right to set off such amount against any payment(s) due and payable or to become due and payable to the first Party under this Contract, provided, however, any amount being disputed under Article 5.6 hereof shall not be considered due and payable until the dispute is finally resolved.

In addition, Purchaser shall be entitled to offset against Milestone payments due to Contractor under this Contract, all undisputed amounts due from CondoSat Associate to Purchaser pursuant to the CondoSat Agreement for (1) Launch Services and (2) the Initial Insurance (as defined in the CondoSat Agreement), but only to the extent such amounts remain unpaid and in default after expiration of all time periods allowed for payment thereunder, including time periods to cure such default. As a condition to exercising any rights of offset under this provision, Purchaser shall assign to Contractor, including by way of subrogation, its rights against CondoSat Associate to such payments and all associated remedies.

5.8 Audit Rights and Procedures

Contractor shall keep complete, true and accurate books of account and records pursuant to its standard accounting system for the purpose of showing the derivation of all Actual Costs where any payments to be made by Purchaser are based on Actual Costs. Contractor will keep such books and records at Contractor's principal place of business for at least five (5) years following the end of the calendar quarter to which they pertain and make them available at all reasonable times for audit by an independent auditor reasonably acceptable to Purchaser and Contractor. Purchaser may direct an audit of any Actual Costs claimed by Contractor pursuant to this Contract to be performed by an industry recogized independent certified public accounting firm reasonably acceptable to Contractor. Any such audits will be at the expense of Purchaser unless the audit shows that Contractor has overcharged amounts due hereunder during the audited period by more than five percent (5%). In such case, the expense of the audit will be paid by Contractor. Contractor will promptly pay Purchaser the full amount of any overpayment, together with interest at the annual rate of [***] percent ([***]%), compounded monthly, from the date such payment was to have been made. The independent auditor will be directed to report reasons for its findings, and the independent auditor's findings will be binding upon Purchaser and Contractor, provided

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

that Contractor shall have the right to redact any detailed rate information from any report or finding provided by the independent auditor to Purchaser.

ARTICLE 6 — PURCHASER-FURNISHED ITEMS

6.1 Purchaser-Furnished Support

To enable Contractor to perform Launch Support and Mission Operations Support Services, Purchaser shall be responsible for timely making available to Contractor the Purchaser-furnished equipment, facilities and services described in Paragraph 2.6 of Exhibit A, Statement of Work. Such equipment, facilities and services shall be in good working condition and adequate for the required purposes and, for the Launch of the Satellite hereunder, shall be made available free of charge for Contractor's use during the period commencing on the date established therefore at the technical interchange meetings described in Paragraph 2.6.1 of Exhibit A, Statement of Work and continuing through the IOT Complete Date for the Satellite. Purchaser and Contractor will conduct an interface meeting on the date established therefor at the technical interchange meeting described in Paragraph 2.6.1 of Exhibit A, Statement of Work to confirm the availability and adequacy of such Purchaser-furnished equipment, facilities and services.

6.2 Communications Authorizations

Purchaser shall be responsible, at its cost and expense, for preparing, coordinating and filing all applications, registrations, reports, licenses, permits and authorizations with the FCC if required to do so and with any other national governmental agencies having jurisdiction over Purchaser, for the construction, Launch and operation of the Satellite. Contractor shall, at no additional cost to Purchaser, provide all cooperation reasonably necessary in support of Purchaser's preparation, coordination and filing of such applications, registrations, reports, licenses, permits and authorizations, and upon Purchaser's request, Contractor shall actively provide reasonable support to Purchaser's efforts to obtain any licenses, permits, and aurthorizations required by any cognizant regulatory agency. Nothing in this Article 6.2 shall be used to interpret or prejudice any of the rights under the CondoSat Agreement.



SS/L-TP20701 ViaSat Contract

6.3 Radio Frequency Coordination

Purchaser shall be responsible for the timely preparation and submission of all filings required by the International Telecommunication Union (or any successor agency thereto) and all relevant domestic communications regulatory authorities regarding radio frequency and orbital position coordination. Such filings shall be made in accordance with the Radio Regulations of the International Telecommunication Union (or any successor agency) and the laws and regulations of all domestic communications regulatory authorities having jurisdiction over Purchaser. Contractor shall, at no additional cost to Purchaser provide all cooperation reasonably necessary in support of Purchaser's efforts in the preparation and submission of such filings, and upon Purchaser's request, Contractor shall actively provide reasonable support to Purchaser's efforts to obtain any licenses, permits, and aurthorizations required by any cognizant regulatory agency. Nothing in this Article 6.3 shall be used to interpret or prejudice any of the rights under the CondoSat Agreement.

6.4 Satellite Performance Data

Commencing with the first full calendar month following the IOT Complete Date, Purchaser shall provide a report to Contractor describing the general health and operating status of the Satellite. Such report shall be provided to Contractor on a monthly basis thereafter, delivered to Contractor promptly after the end of each month during the Satellite Stated Life. In the event of a Satellite Anomaly that occurs during the Satellite Stated Life, Purchaser shall timely provide Contractor with or give Contractor access to any of Purchaser's data Contractor may reasonably require to investigate and correct (per Article 15.3 hereof) such Satellite Anomaly and/or support Purchaser in making and perfecting claims for insurance recovery relating to such Satellite Anomaly as set forth in Article 32.

6.5 Late Delivery of Purchaser-Furnished Items or Services

Exhibit A, Statement of Work, paragraph 2.6 sets forth a complete listing of all Purchaser-furnished items or services and the delivery date for such items and services. The late delivery of Purchaser-furnished items or services, individually or combined, shall be considered an event beyond the reasonable control of Contractor, and Contractor shall be entitled to adjustments in the Firm Fixed Price and schedule and other materially affected terms of the Contract, in accordance with Article 18, provided



SS/L-TP20701 ViaSat Contract

that: (a) such late delivery of Purchaser-furnished items or services was substantially due to no fault of Contractor; (b) Contractor promptly notifies Purchaser in writing of any applicable late delivery of Purchaser-furnished items or services and the expected impacts therefor; and (c) Contractor uses Reasonable Efforts to avoid and/or mitigate the effect of the late delivery of Purchaser-furnished items or services.



SS/L-TP20701 ViaSat Contract

ARTICLE 7 — COMPLIANCE WITH LAWS AND DIRECTIVES

7.1 General

Each Party shall, at its expense, perform its obligations hereunder in accordance with all applicable laws, regulations, and policies of the any federal, state, local, or foreign government and the conditions of all applicable federal, state, local, or foreign government approvals, permits, or licenses.

7.2 Compliance with U.S. Export Control Laws

Any obligation of a Party hereunder to provide hardware, software, Deliverable Data, other technical information, technical services, Training, or any access to facilities of the other Party and its personnel and/or its representatives shall be subject to applicable U.S. Government export control and security laws, regulations, policies and license conditions. The Parties shall work cooperatively and in good faith to implement this Contract in compliance with such laws, regulations, policies and license conditions. If and to the extent required by U.S. law, the Parties and their personnel and/or representatives shall enter into U.S. Government-approved agreement(s), including a Technical Assistance Agreement(s), separate from this Contract, governing the provision of hardware, software, Deliverable Data, other technical information, technical services, Training, or access to facilities in connection with this Contract.

7.3 Licenses and Other Approvals

Contractor shall use Reasonable Efforts to obtain and shall maintain all applicable approvals, permits, and licenses as may be required by any government, foreign or domestic, for the performance of the Work, including but not limited to all authorizations required for the import or export of any Deliverable Item, or any part thereof, as well as any agreements and other approvals of the U.S. Government that are required for Purchaser and "foreign person" personnel and/or representatives of Purchaser (including, but not limited to, Eutelsat, foreign subsidiaries and related entities of Purchaser involved with the procurement) as well as Purchaser's insurance providers, to have access to Contractor facilities, hardware, software, Deliverable Data, Training, other technical information or technical services in connection with the performance of this Contract. In addition, Contractor shall provide reasonable support to Purchaser in obtaining any necessary approvals, permits, and licenses for the performance of Purchaser's obligations hereunder. A "foreign person" shall be as defined in the U.S.



SS/L-TP20701 ViaSat Contract

International Traffic in Arms Regulations, 22 C.F.R. §120.16. As early as practicable, and in no event later than ninety (90) days after EDC, Purchaser shall provide Contractor with a list of countries (if other than the U.S.) of which "foreign person" personnel and/or representatives of Purchaser (including, but not limited to foreign subsidiaries and related entities of Purchaser involved with the procurement) as well as Purchaser's insurance providers, if such personnel, representatives, insurance providers will or may have access to U.S. export-controlled items or services under this Contract. Purchaser shall provide such reasonable cooperation and support as necessary for Contractor to apply for and maintain such required U.S. export licenses, agreements and other approvals, and shall promptly notify Contractor of any occurrence or change in circumstances of which it becomes aware that is relevant to or affects such export licenses, agreements and approvals. Contractor shall review with Purchaser any application Contractor makes to any government department, agency or entity for any permit, license, agreement or approval that will be signed by Purchaser as may be required for performance of the Work, prior to submission of such application. Contractor shall provide Purchaser a minimum of five (5) Business Days to review such application prior to submission to such governmental entity, and Contractor shall in good faith consider any comments and proposed revisions made by Purchaser for incorporation into such application. At Purchaser's request, Contractor shall include Purchaser (and related entities involved with the procurement) as a named party in any application for approval of such U.S. export licenses, agreements and other approvals so as to permit Purchaser to be present during any discussion with or meetings where Purchaser's foreign subsidiaries/related entities, or insurance providers, may receive from, or discuss, with Contractor any U.S. exportcontrolled items and/or services. Contractor shall provide the parties to such U.S. export licenses and agreements copies of the export licenses and agreements, including any U.S. Government provisos related to same

NOTWITHSTANDING ANY PROVISION IN THIS CONTRACT, IN NO EVENT SHALL EITHER PARTY BE OBLIGATED UNDER THIS CONTRACT TO PROVIDE ACCESS TO THE OTHER PARTY'S FACILITIES OR SUBCONTRACTOR FACILITIES, PROVIDE ACCESS TO OR FURNISH HARDWARE, SOFTWARE, DELIVERABLE DATA OR OTHER TECHNICAL INFORMATION, OR PROVIDE TECHNICAL/DEFENSE SERVICES OR TRAINING, TO ANY PERSON EXCEPT IN COMPLIANCE WITH



SS/L-TP20701 ViaSat Contract

APPLICABLE U.S. EXPORT CONTROL LAWS, REGULATIONS, POLICIES AND LICENSE CONDITIONS.

7.4 No Unauthorized Exports or Retransfers

NEITHER PARTY SHALL RE-EXPORT OR RE-TRANSFER TO ANY THIRD PARTY ANY HARDWARE, SOFTWARE, DELIVERABLE DATA, OTHER TECHNICAL INFORMATION, TECHNICAL SERVICES, OR OTHER ITEMS FURNISHED HEREUNDER, EXCEPT AS EXPRESSLY AUTHORIZED BY THE U.S. GOVERNMENT IN ACCORDANCE WITH THE EXPORT LICENSES, AGREEMENTS AND OTHER APPROVALS REFERENCED IN THIS ARTICLE 7 OR AS OTHERWISE EXPRESSLY AUTHORIZED UNDER U.S. EXPORT CONTROL LAWS.



SS/L-TP20701 ViaSat Contract

ARTICLE 8 — ACCESS TO WORK-IN-PROGRESS

8.1 Work in Progress at Contractor's or Major Subcontractor's Facility

Subject to Article 7, compliance with Contractor's normal and customary safety and security regulations and practices (or, if applicable, those of a Major Subcontractor), which shall be provided in writing to Purchaser upon Purchaser's request prior to any facility visit, and the protection of third party proprietary information, Purchaser personnel (such term to include Eutelsat and Purchaser's duly appointed consultants and agents) shall be allowed access to all Work being performed at Contractor's or any Major Subcontractor's facility for the Satellite and other Deliverable Items, for the purpose of observing the progress of such Work; provided, however, access by Purchaser personnel to any Major Subcontractor facility shall require such Major Subcontractor's prior written consent, which Contractor shall employ Reasonable Efforts to obtain, and Contractor shall be entitled to accompany any Purchaser personnel to the facility. Such access shall be upon reasonable prior written notice to Contractor and shall occur during normal working hours or at such other hours as Contractor may agree. Subject to Article 7, compliance with Contractor's normal and customary safety and security regulations and practices (or, if applicable, those of a Major Subcontractor), which shall be provided in writing to Purchaser upon Purchaser's request prior to any facility visit, and the protection of third party proprietary information, Purchaser personnel will have reasonable access for evaluation, inspection, and use in connection with the planned operation of the Satellite to (i) Deliverable Data; (ii) work-in-progress and technical and schedule data and documentation relevant to the Work; (iii) drawings, circuit diagrams/schematics, specifications, standards or process descriptions applicable to the Work; and (iv) data and documentation provided to Contractor by its Subcontractors applicable to the Work. Contractor shall provide Purchaser access to a program web site (from which Purchaser will be permitted to download, print, and save documents) to serve as a repository for such information and Contractor shall provide a typical site map or index to the contents of that web site. All Purchaser personnel (and Purchaser's duly appointed consultants and agents) who are "U.S. persons" under the ITAR and otherwise meet the requirements for access set forth in this Article shall be provided non-escort badges to agreed work areas while the Work is being performed, though such access may be restricted to relevant buildings and areas where the subject Work in Progress resides. Contractor will use Reasonable Efforts to obtain similar access to



SS/L-TP20701 ViaSat Contract

Work that is being performed at the facilities of Contractor's Major Subcontractors subject to the conditions set forth in the first sentence of this Paragraph. For the avoidance of doubt, any communications between Purchaser personnel and any foreign Subcontractor shall be conducted through Contractor.

8.2 Electronically-Generated Information

With regard to electronically generated information, Contractor will provide Purchaser with an electronic copy thereof and/or electronic access (via the internet or Purchaser e-mail) to information regarding program performance and documentation that will advise Purchaser, on a current basis, of program specific issues, decisions and problems. Contractor shall establish data links between its and Purchaser's facilities such that Purchaser has remote electronic access to those project-related documents identified in Exhibit A, Satellite Statement of Work. Contractor will also provide Purchaser personnel with "real time" access to all measured data, when feasible, for the Work on a non-interference, no-cost basis.

8.3 On-Site Facilities for Purchaser's Personnel

For the purpose of monitoring the progress of the Work being performed by Contractor hereunder, Contractor shall provide office facilities at Contractor's plant for resident Purchaser personnel (and/or Purchaser's duly appointed consultants and agents) through the IOT Complete Date. The office facilities to be provided shall include a reasonable amount of office space, office furniture, local telephone service, reasonable long-distance telephone usage, access to copy machines, facsimile machines, internet account and access, reasonable clerical support, and meeting rooms to the extent necessary to enable Purchaser personnel to monitor the progress of Work under this Contract. Contractor will use Reasonable Efforts to obtain similar office facilities at the plants of Contractor's Major Subcontractors for visits and meetings of Purchaser and Contractor to such plants.

8.4 Purchaser Personnel as Competitors/Foreign Persons

Purchaser shall use Reasonable Efforts to verify that Purchaser's consultants and agents used in connection with this Contract are not currently employed by companies or entities that are Competitors of Contractor. Purchaser shall notify Contractor in writing of the name, title or function, business relationship, employer, citizenship status



SS/L-TP20701 ViaSat Contract

under U.S. export laws and such other information as may be reasonably requested by Contractor, with respect to each of its intended consultants and agents, and cause each such consultant and agent (except for attorneys and other professionals who are already bound by obligations of confidentiality) to: (i) execute a confidentiality agreement directly with Contractor in form and substance reasonably satisfactory to Contractor and containing terms substantially the same as those set forth in Article 28; and, (ii) pursuant to Article 7, execute a Technical Assistance Agreement or other agreement to ensure compliance with applicable U.S. export control laws and regulations to the extent required by applicable U.S export laws or regulations. Without prejudice to Contractor's rights and obligations under Article 7, Contractor may in its reasonable discretion deny any consultant or agent of Purchaser access to Contractor or a Subcontractor's facilities, products or information if such consultant or agent is currently employed by a Competitor, or if such consultant or agent fails to execute any of the agreements identified in (i) and (ii) above, where necessary. Notwithstanding the foregoing, Purchaser consultants or agents who are also employed or engaged by Contractor Competitors shall not be denied access to Contractor or a Subcontractor's facilities, products or information if Purchaser notifies Contractor of the identity of such consultants or agents, identifies the Competitor(s) that such consultant(s) or agent(s) are employed by, and Purchaser implements appropriate protective measures (firewalls) to safeguard the confidentiality of Contractor's data, documentation and information as required by this Contract.

8.5 <u>Meetings and Presentations</u>

Purchaser shall be entitled to reasonable notice of and to attend: (a) all formal program- specific meetings; and (b) reviews required in Exhibit A, Statement of Work. Purchaser shall have the right to participate in and to make recommendations in all such meetings and reviews at the system, subsystem and unit level. In addition, Contractor and Purchaser shall mutually agree on which informal program-specific meetings to which Purchaser will be invited to attend.

Copies of presentations or other documents utilized during these meetings shall be furnished or made available to Purchaser and Purchaser's consultants. Purchaser's management personnel, as may be deemed appropriate by Purchaser, shall be invited to the Quarterly Summary Executive Reviews. Contractor shall be represented by its



SS/L-TP20701 ViaSat Contract

Program Manager and such other personnel as are required to support the particular presentation.

8.6 <u>Interference with Operations</u>

Purchaser shall exercise its rights under this Article 8 in a manner that does not unreasonably interfere with Contractor's or its Subcontractors' normal business operations or Contractor's performance of its obligations under this Contract or any agreement between Contractor and its Subcontractors.

8.7 Purchaser Inspection Not Acceptance

The inspection, examination, or observation by Purchaser with regard to any portion of Work produced under this Contract shall not constitute any Acceptance thereof, nor shall it relieve Contractor from fulfilling its contractual obligations hereunder.

8.8 CondoSat Associate

Purchaser agrees that Contractor may provide the same access to the Work as provided to Purchaser hereunder to CondoSat Associate personnel and representatives on the same terms and with the same restrictions as set forth herein. In the event that Purchaser desires that any CondoSat Associate personnel and/or representatives be denied such access, Purchaser shall so notify Contractor in writing, including the name of CondoSat Associate personnel and/or representative, and thereafter Contractor shall notify CondoSat Associate of the restriction and deny access to such persons.



SS/L-TP20701 ViaSat Contract

ARTICLE 9 — SATELLITE PRE-SHIPMENT REVIEW (SPSR) AND DELIVERY

- 9.1 Satellite Pre-Shipment Review (SPSR)
 - 9.1.1 Contractor to Conduct a Review of the Satellite Prior to Shipment. Contractor shall conduct a detailed and comprehensive review of the Satellite prior to Contractor's shipment of the Satellite to the Launch Site or its entering into storage. This review shall be conducted in accordance with the terms of this Article 9 and Paragraph 2.4.5 of Exhibit A, Statement of Work (a "Satellite Pre-Shipment Review" or "SPSR").
 - 9.1.2 <u>Time, Place and Notice of SPSR</u>. The SPSR shall take place at Contractor's facility. Contractor shall notify Purchaser in writing on or before thirty (30) days prior to the date that the Satellite will be available for the SPSR, which shall be the scheduled date for commencement of such SPSR. If Purchaser cannot attend the SPSR on such initially scheduled date, Contractor shall make Reasonable Efforts to accommodate Purchaser's scheduling requirements. Purchaser agrees that CondoSat Associate personnel and/or representatives shall be allowed to attend the SPSR, but that Purchaser shall have final approval of the list of non-Contractor attendees.
 - 9.1.3 Conduct and Purpose of SPSR. The SPSR shall be conducted in accordance with the terms of this Article 9 and Paragraph 2.4.5 of Exhibit A, Statement of Work. The purpose of the SPSR shall be to: (i) review test data and analyses for the Satellite; (ii) demonstrate testing has been completed in accordance with the applicable portions of Exhibit D, Satellite Program Test Plan: and, (iii) determine whether the Satellite meets applicable Exhibit B, Satellite Performance Specification requirements (except those that have been waived pursuant to Article 9.1.4 below) and is therefore ready for shipment to the Launch Site.
 - 9.1.4 <u>Waivers and Deviations</u>. Contractor shall submit to Purchaser any request for a waiver of, or deviation from, provisions(s) of the Performance Specification applicable to the Satellite or other Deliverable Item. Purchaser shall consider each such request in good faith in accordance with industry



SS/L-TP20701 ViaSat Contract

standard practices. A request for waiver or deviation shall be deemed granted only if it has been approved in writing by a duly authorized representative of Purchaser (as defined in Article 31.3). Each such waiver or deviation approved by Purchaser shall be deemed an amendment to the Performance Specification for such Satellite or Deliverable Item, permitting such waiver thereof, or deviation therefrom, effective on or after the date of such approval for such Satellite or Deliverable Item. In the event that Purchaser approves any waiver or deviation under this paragraph, Purchaser shall be entitled to an equitable reduction in the Firm Fixed Price and Contractor shall be entitled to an equivalent reduction in the [***], if applicable.

- 9.1.5 <u>Purchaser's Inspection Agents</u>. Purchaser may, subject to prior written notice to Contractor, cause any Purchaser personnel, consultant or agent designated by Purchaser to observe the SPSR pursuant to this Article 9; provided, however, that the provisions of Article 7 and Article 8.4 shall apply to any such agent.
- 9.1.6 SPSR Results. In the event that the SPSR demonstrates that: (i) testing has been performed in accordance with Section 5.2 of Exhibit D, Satellite Program Test Plan, and (ii) such Satellite conforms to the applicable requirements of Exhibit B, Satellite Performance Specification (including any waivers or deviations approved by Purchaser pursuant to Article 9.1.4), Purchaser shall provide written certification (a "Certification") to Contractor (within 72 hours after Purchaser receives written notice of completion of the SPSR from Contractor) of its concurrence with the results of the SPSR (including any waiver of its right to compel correction of those non-conformances to the requirements of Exhibit B, Satellite Performance Specification, specified by Purchaser in such notice), and the Satellite shall be deemed ready for shipment to the Launch Site.

In the event that such SPSR discloses (i) any failure to conduct testing in accordance with Section 5.2 of Exhibit D, Satellite Program Test Plan, or (ii) any non-conformance of such Satellite to the requirements of Exhibit B, Satellite Performance Specification, either of which is not the subject of any waivers or deviations approved by Purchaser pursuant to Article 9.1.4, Purchaser shall

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

provide written notification thereof (a "Notice of Non-Conformance") to Contractor within 72 hours after Purchaser receives written notice of completion of the SPSR from Contractor, which written notification shall state each such non-conformance Purchaser requires to be corrected or repaired (with reference to the specific provision of Exhibit D, Satellite Program Test Plan, or Exhibit B, Satellite Performance Specification, deemed not met). Contractor shall correct or repair each non-conformance with all deliberate speed and thereafter conduct additional testing and/or a "delta" SPSR, in accordance with the provisions of this Article 9, to the extent necessary to demonstrate that the Satellite conforms to the requirements of Exhibit B, Satellite Performance Specification, after which Purchaser shall provide a Certification or Notice of Non-Conformance within the applicable time frame specified above. Contractor shall be required to repeat the process described in this Paragraph until Purchaser provides Contractor with a Certification pursuant to the requirements of this Article 9.

If Purchaser fails to provide a written Certification or Notice of Non-Conformance as required by either of the preceding Paragraphs of this Article 9.1.6 within the time specified, and Contractor has provided written notice advising Purchaser of such (a "Cure Letter"), the SPSR shall be deemed to have been successfully completed if Purchaser has still not submitted a written Certification or Notice of Non-Conformance within one (1) Business Day of Purchaser's receipt of the Cure Letter and the Satellite shall be deemed ready for shipment to the Launch Site.

Upon receipt of Purchaser's Certification in accordance with this Article 9, Contractor shall thereafter transport such Satellite in accordance with Contractor's standard commercial practices to the Launch Site (and shall be deemed Delivered when received at the Launch Site), and proceed to prepare the Satellite for Launch. Contractor shall not ship the Satellite to the Launch Site until all non-conformances are corrected, repaired or have a Purchaser-approved waiver or deviation.

9.1.7 <u>Inspection Costs Borne by Purchaser</u>. All costs and expenses incurred by Purchaser and its agents in the exercise of its inspection rights under this Article 9, including travel and living expenses, shall be borne solely by Purchaser.



SS/L-TP20701 ViaSat Contract

9.1.8 Correction of Deficiencies after SPSR. If at any time following completion of SPSR of the Satellite and prior to Intentional Ignition (or in the event of a Terminated Ignition, prior to any subsequent Intentional Ignition), it is discovered that such Satellite has a Defect or fails to meet the requirements of Exhibit B, Satellite Performance Specification, as they may be modified as of such time pursuant to Article 9.1.4, Contractor shall promptly correct such deficiencies prior to Intentional Ignition (or in the case of a Terminated Ignition, prior to any subsequent Intentional Ignition) in accordance with the applicable terms of this Contract, including price and/or schedule adjustments for which Contractor may be entitled pursuant to Article 12.1(B). Contractor shall use Reasonable Efforts to avoid and minimize delays associated with any such Defects as further described in Article 3.4. In the event of a Terminated Ignition, Contractor and Purchaser shall proceed in accordance with Article 12.1(B) for those actions necessary to prepare the Satellite for relaunch.



SS/L-TP20701 ViaSat Contract

ARTICLE 10 — ACCEPTANCE OF SATELLITE, LAUNCH SUPPORT AND MISSION OPERATIONS SUPPORT SERVICES AND IN-ORBIT TEST

10.1 Satellite Acceptance

Acceptance of the Satellite by Purchaser (including the Loral Space Payload on behalf of CondoSat Associate) shall occur finally and irrevocably for all purposes hereunder upon the earlier to occur of: (i) the IOT Complete Date (as defined in Article 10.3. below); or (ii) the instant immediately prior to an event on or after Intentional Ignition resulting in the Total Loss of the Satellite (or the Satellite being reasonably determined to be a Total Loss).

10.2 Launch Support and Mission Operations Support Services

Upon arrival of the Satellite at the Launch Site, Contractor shall proceed with the provision of Launch Support Services in accordance with Exhibit A, Statement of Work. After Launch of the Satellite by the Launch Agency, Contractor shall proceed with the provision of Mission Operations Support Services in accordance with Exhibit A, Statement of Work.

10.3 In-Orbit Testing (IOT)

Thirty (30) days prior to the scheduled Launch of the Satellite, Contractor shall notify Purchaser in writing of the IOT schedule with respect to such Satellite. Purchaser personnel, including but not limited to its contractors and agents may, at Purchaser's election and subject to Article 7 and Article 8.4, observe such IOT at either Purchaser's or Contractor's facilities. CondoSat Associate personnel and/or representatives shall also be allowed to observe such IOT at Contractor's facilities, if so authorized by Purchaser.

After the Satellite has been placed in its orbital slot, Contractor shall perform IOT in accordance with Exhibit D, Satellite Program Test Plan, and conduct an IOT review with Purchaser and CondoSat Associate within three (3) days of completing IOT (with a summary IOT report being submitted at least 24 hours prior to conducting the IOT review), all in accordance with the applicable provisions of Exhibit A, Statement of Work, and Exhibit D, Satellite Program Test Plan. Upon the completion of the IOT review, Purchaser shall be deemed to have accepted the IOT results (the "IOT Complete Date").



SS/L-TP20701 ViaSat Contract

10.4 Orbital Performance Incentives

From and after the IOT Complete Date, Contractor shall have the right to earn Orbital Performance Incentives in accordance with Article 13.



SS/L-TP20701 ViaSat Contract

ARTICLE 11 — ACCEPTANCE OF DELIVERABLE ITEMS OTHER THAN THE SATELLITE

11.1 DSS

11.1.1 With respect to the DSS (including Software integrated into the DSS), Contractor shall conduct an inspection and test the DSS in accordance with Exhibit H, Dynamic Satellite Simulator Statement of Work and Functional Requirements Document, in coordination with Purchaser or Purchaser's representative. If the tests establish that the DSS meets the requirements of Exhibit H, Contractor shall deliver the test results and provide a certification to Purchaser that the DSS meets the requirements of this Contract. Based upon this Contractor's certification and the inspection and test results, and the results of any additional inspection or testing that Purchaser may reasonably conduct in coordination with Contractor, Purchaser shall either Accept the same in writing ("Acceptance" with respect to the DSS) or notify Contractor in writing of those particulars in which the DSS is unacceptable. Should Purchaser fail to notify Contractor in writing of those particulars in which the DSS is unacceptable within ten (10) Business Days of certification, the DSS shall be deemed Accepted by Purchaser. Upon receipt of a notice that the DSS is unacceptable to Purchaser, which notice shall state the particulars relating to such unacceptability. Contractor shall remedy the non-conformances, conduct additional testing as appropriate, and schedule another test of the DSS as appropriate in the presence of Purchaser or Purchaser's representative. When such particulars have been remedied to conform to all applicable requirements of Exhibit H, the DSS shall be promptly Accepted by Purchaser in writing. Contractor shall be required to repeat the process described in this Paragraph until the DSS shall have been Accepted by the Purchaser in writing pursuant to the criteria of this Paragraph. Contractor shall take all appropriate measures to ensure that Acceptance of the DSS is achieved prior to Acceptance of the Satellite. In the event that Acceptance of the Satellite occurs prior to Acceptance of the DSS, Contractor shall, at Contractor's sole cost and expense, and at Purchaser's option, (i) continue to operate such Satellite for Purchaser until the time of Acceptance of such DSS and (ii) expeditiously complete the tasks necessary to facilitate Acceptance of the DSS.



SS/L-TP20701 ViaSat Contract

11.2 Deliverable Data

For any Deliverable Data (including Deliverable Items of Software but excluding Software integrated into the DSS and Satellite, which are subject to Article 11.1 and Article 10.1, respectively) that requires Purchaser approval pursuant to Exhibit A, Statement of Work, Purchaser shall, within ten (10) Business Days of Delivery, notify Contractor in writing that such Deliverable Data has either been: (i) Accepted, or (ii) that such Deliverable Data does not comply with the requirements of Exhibit A, Statement of Work, identifying each such non-conformance. If Purchaser notifies Contractor of any non-conformance pursuant to the foregoing, Contractor shall promptly correct any non-compliant aspect of such Deliverable Data identified in such notice from Purchaser, and re-submit the Deliverable Data to Purchaser for a subsequent inspection to verify that Contractor has corrected the previously-identified non-compliance and that the Deliverable Data complies with the requirements of Exhibit A, Statement of Work. If Purchaser fails to provide notice within the time specified above, Acceptance shall be deemed to have occurred with respect to such Deliverable Data. Contractor shall be required to repeat the process described in this Paragraph until such time as the relevant Deliverable Data has either been Accepted by Purchaser in writing, or has been deemed Accepted, in accordance with this Paragraph. For Deliverable Data that does not require Purchaser approval pursuant to Exhibit A, Statement of Work, Acceptance of such Deliverable Data shall be deemed to have occurred upon Delivery, provided that such Deliverable Data complies with the requirements of Exhibit A, Statement of Work and further provided that in the event of any non-compliance, Purchaser shall have the remedies set forth in Article 15.2.4.

11.3 Training

Acceptance of Training, or any part thereof, required by Exhibit A, Statement of Work shall occur in accordance with this Article 11.3. Training, or any part thereof, furnished to Purchaser shall be accompanied by written notice from Contractor specifying that portion of the Training being furnished. Acceptance of Training, or any part thereof, shall be deemed to occur ten (10) Business Days after completion of such Training or part thereof unless within such time Purchaser provides a written notice to Contractor that the Training, or part thereof, does not meet Purchaser's reasonable satisfaction with reference to the requirements of Exhibit A, Statement of Work, identifying each non-conformance. In the event that Purchaser does not Accept Training, or any part thereof,



SS/L-TP20701 ViaSat Contract

as set forth in this Paragraph, Contractor shall repeat the Training, or relevant part thereof.

11.4 Deliverable Services

Acceptance of Deliverable Services, or any part thereof, shall occur in accordance with this Article 11.4. To the extent feasible, Deliverable Services furnished to Purchaser shall be accompanied by written notice from Contractor specifying that portion of the Deliverable Services being furnished. Acceptance of Deliverable Services, or any part thereof, shall be deemed to occur: (a) in a case where re-performance of the Deliverable Services is not practical (for example, Launch Support Services are performed real-time), upon completion of such Deliverable Services or part thereof; or (b) in a case where re-performance of the Deliverable Services is practicable, ten (10) Business Days after completion of such Deliverable Services or part thereof, unless within such time Purchaser provides a written notice to Contractor that the Deliverable Services, or part thereof, do not meet the requirements of Exhibit A, Statement of Work, identifying each non-conformance.

11.5 Reserved

11.6 Satellite Control Facility Equipment

If Purchaser exercises its option set forth in Article 2.4, upon completion of the installation of the Satellite Control Facility equipment (not including the DSS, which is subject to Article 11.1), Contractor shall conduct an inspection and test the Satellite Control Facility equipment in accordance with Exhibit F, Satellite Control Facility Requirements Document, in coordination with Purchaser or Purchaser's representative. If such inspection and tests, and data resulting from such inspection and tests, establish that the Satellite Control Facility equipment meets the requirements of Exhibit F, Satellite Control Facility Requirements Document, Contractor shall deliver the inspection and test results and associated data and provide a certification to Purchaser that the Satellite Control Facility equipment meets the requirements of this Contract, together with a notification of the period in which Purchaser may provide Acceptance or notice of non-conformance for the equipment. Based upon such Contractor certification and the inspection and test results, and any inspection or testing that Purchaser may reasonably conduct in coordination with Contractor, Purchaser shall either Accept the same in writing or notify Contractor in writing of those particulars in which the Satellite Control



SS/L-TP20701 ViaSat Contract

Facility equipment is unacceptable. Should Purchaser fail to notify Contractor in writing of those particulars in which the Satellite Control Facility Equipment is unacceptable within ten (10) Business Days of certification, and Contractor has provided written notice advising Purchaser of such, the Satellite Control Facility equipment shall be deemed Accepted by Purchaser. Upon receipt of a notice that the Satellite Control Facility equipment is unacceptable to Purchaser, which notice shall state the particulars relating to such unacceptability, Contractor shall remedy the non-conformances, conduct additional testing as appropriate, and schedule another test of the Satellite Control Facility equipment as appropriate in the presence of Purchaser or Purchaser's representative. Upon remedy of such particulars to conform to all applicable requirements of Exhibit F, Satellite Control Facility Requirements Document, the Satellite Control Facility Equipment shall be Accepted by Purchaser in writing. Contractor shall be required to repeat the process described in this Paragraph until the Satellite Control Facility equipment shall have been Accepted by the Purchaser in writing as set forth in this Paragraph. Contractor shall take all appropriate measures to ensure that Acceptance of the Satellite Control Facility Equipment is achieved prior to Acceptance of the Satellite. In the event that Acceptance of the Satellite occurs prior to Acceptance of the Satellite Control Facility Equipment, Contractor shall, at Contractor's sole cost and expense, and at Purchaser's option, (i) continue to operate such Satellite for Purchaser until the time of Acceptance of such Satellite Control Facility Equipment and (ii) expeditiously complete the tasks necessary to facilitate Acceptance of the Satellite Control Facility Equipment.

11.7 Purchaser's Inspection Agents

Purchaser may, upon giving prior written notice to Contractor, cause any Purchaser consultant or agent designated by Purchaser to observe or conduct the acceptance inspection pursuant to this Article 11 in whole or in part; provided, however, that the provisions of Article 7 and Article 8.4 shall apply to any such consultant or agent and such consultant or agent shall comply with Contractor's normal and customary safety and security regulations provided to Purchaser in writing in advance of such inspection.

11.8 Waivers and Deviations

Waivers of or deviations from the Performance Specification applicable to any Deliverable Item subject to acceptance inspection pursuant to this Article 11 shall be addressed as set forth in Article 9.1.4.



SS/L-TP20701 ViaSat Contract

11.9 <u>Inspection Costs Borne by Purchaser</u>

All costs and expenses incurred by Purchaser or its consultants or agents in the performance of its inspection rights under this Article 11, including travel and living expenses, shall be borne solely by Purchaser.

11.10 Warranty Obligations

In no event shall Contractor be released from any of its warranty obligations applicable to any Deliverable Item as a result of such Deliverable Item having been Accepted as set forth in this Article 11.



SS/L-TP20701 ViaSat Contract

ARTICLE 12 — TRANSFER OF TITLE AND RISK OF LOSS

12.1 Satellite

(A) Title to and risk of loss or damage for the Satellite, except for the Loral Space Payload, shall pass from Contractor to Purchaser (and title to the Loral Space Payload portion will pass to CondoSat Associate simultaneously, pursuant to the Loral Space Payload Contract) at the time of Intentional Ignition. In the event of a Terminated Ignition not resulting in Total Loss, title to and risk of loss or damage for the Satellite (including the Loral Space Payload portions of the Satellite) shall revert to Contractor upon such Terminated Ignition. If Contractor re-acquires title and risk of loss or damage as set forth in the immediately preceding sentence, title to and risk of loss or damage for the Satellite, except for the Loral Space Payload, shall again pass to Purchaser (title to the Loral Space Payload portion of the Satellite will again pass to CondoSat Associate simultaneously, pursuant to the Loral Space Payload Contract) upon the subsequent Intentional Ignition. Prior to Intentional Ignition, the Parties shall agree to appropriate revisions to certain definitions (i.e., Intentional Ignition, Launch, Terminated Ignition, and other related provisions) as required to ensure that risk of loss transfers from Contractor to Purchaser in a manner that does not result in a lack (or gap in time) of insurance coverage for the Satellite at any time between the time period covered by the Ground Insurance to be provided by Contractor pursuant to Article 32.5 and the applicable coverage start date for any Launch and In-Orbit Insurance Policy.

(B) In the event of a Terminated Ignition, once the launch pad has been declared safe and the Launch Agency authorizes the start of the demating operations, Contractor shall, at Purchaser's request, immediately take all necessary actions to prepare the Satellite for a relaunch, including: (a) supporting the Launch Agency in demating the Satellite from the Launch Vehicle and conducting defueling operations; (b) directly performing inspection and testing, refurbishment, storage, repair and replacement of damaged Component(s), and transportation of the Satellite to and from the Launch Site; and (c) providing additional Launch support services for the subsequent Launch of the Satellite. Such actions shall be handled as a change pursuant to Article 16.1, and for the avoidance of doubt is not considered to be performance of Work pursuant to the terms of the warranty set forth in Article 15.2.1. Promptly after a Terminated Ignition, the Parties shall establish a new due date for the repaired/refurbished Satellite. In the event of a



SS/L-TP20701 ViaSat Contract

Terminated Ignition that results in the Total Loss of the Satellite, title to and risk of loss or damage for the Satellite shall remain with Purchaser.

(C) EXCEPT WITH RESPECT TO WILLFUL MISCONDUCT OR GROSS NEGLIGENCE, UPON AND AFTER INTENTIONAL IGNITION, UNLESS AND TO THE EXTENT OF A TERMINATED IGNITION WHERE CONTRACTOR RE-ACQUIRES TITLE TO AND RISK OF LOSS OR DAMAGE FOR THE SATELLITE AS PROVIDED IN ARTICLE 12.1, CONTRACTOR'S SOLE OBLIGATION, LIABILITY OR FINANCIAL RISK, AND THE SOLE AND EXCLUSIVE REMEDIES OF PURCHASER OR ANY PARTY ASSOCIATED WITH PURCHASER, WITH RESPECT TO THE SATELLITE'S DESIGN, WORKMANSHIP, CONFORMITY TO SPECIFICATION, USE OR PERFORMANCE OF SUCH SATELLITE, INCLUDING ANY ASSISTANCE OR ADVICE (ACTUAL OR ATTEMPTED) PROVIDED OR OMITTED AS CONTEMPLATED BY ARTICLES 15.3 AND 32 HEREOF (INCLUDING WITH RESPECT TO ANY ACTUAL OR CLAIMED DEFECT CAUSED OR ALLEGED TO BE CAUSED AT ANY TIME, WHETHER BEFORE OR AFTER INTENTIONAL IGNITION, BY CONTRACTOR'S OR ANY OF ITS SUBCONTRACTORS' NEGLIGENCE) SHALL BE AS SET FORTH IN ARTICLES 5.2.2 (WITH RESPECT POST-LAUNCH MILESTONE PAYMENTS), 13, 15.3, 19, AND 32, IN ALL CASES SUBJECT TO THE LIMITATION OF LIABILITY SET FORTH IN ARTICLE 27. THE FOREGOING SHALL NOT PREJUDICE OR LIMIT EITHER PARTY'S RIGHTS AND REMEDIES WITH RESPECT TO ARTICLE 28, DISCLOSURE AND HANDLING OF PROPRIETARY INFORMATION AND ARTICLE 39, INTELLECTUAL PROPERTY.

12.2 Deliverable Items Other Than the Satellite

Title to and risk of loss or damage for the DSS and the Satellite Control Facility Equipment (including the Spares), shall pass from Contractor to Purchaser upon Acceptance thereof pursuant to Article 11.1 and Article 11.6, respectively. Purchaser's rights in and to Deliverable Data (including, without exception, Deliverable Items of Software and Software integrated into hardware) are as set forth in Article 39, Intellectual Property.



SS/L-TP20701 ViaSat Contract

ARTICLE 13 — ORBITAL PERFORMANCE INCENTIVES

13.1 General

Contractor may earn, and Purchaser shall pay, if earned, Orbital Performance Incentives with respect to the Satellite in the amounts set forth in Article 13.2. Contractor may earn and accrue the Orbital Performance Incentives over the Orbital Performance Incentive Period with respect thereto. All measurements, computations and analyses made pursuant to this Article 13 shall be made in accordance with good engineering practice applying standards generally applicable in the aerospace industry.

13.2 Daily Rate of Orbital Performance Incentives

Contractor shall be entitled to earn Orbital Performance Incentives in the cumulative amount of [***] percent ([***]%) of the Firm Fixed Price (after all Firm Fixed Price adjustments made to this Contract) over the Orbital Performance Incentive Period at a daily rate calculated by dividing the total amount of Orbital Performance Incentives by 5,479 (as of EDC, the daily rate is \$[***]/ 5,479 = \$[***]) for each day that there is no Transponder Failure (the "Daily Rate").

For each day (or any portion thereof) during the Orbital Performance Incentive Period that there are one or more Transponder Failures, the Daily Rate shall be adjusted in accordance with the following formula:

 $ADR = D \times (T/N)$

Where:

ADR = Adjusted Daily Rate;

D = Daily Rate;

N = Number of Transponders (which shall be two times the number of the U.S. Beams and Additional Beams set forth in Exhibit B, Satellite Performance Specification); and

T = number of Transponders that have not experienced a Transponder Failure

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

during the day. For the purposes of determining "T", the term "Transponder" and "Transponder Failure" shall not include transponders allocated to the Loral Space Payload.

For example, assuming N=126, for each Transponder Failure that occurs in any day during the Orbital Performance Incentive Period, the Daily Rate for that day is reduced by U.S. \$[***](U.S. \$[***]).

13.3 On-Board Redundancy; Spares

On-board redundancy and/or spare Components shall be taken into consideration to maintain service on a Satellite, and such use shall be deemed normal operating procedure for purposes of this Article 13 so long as the applicable criteria of Exhibit B are met by such Satellite, as the case may be.

13.4 Orbital Storage

If Purchaser places the Satellite in Orbital Storage (other than due to the inability to operate the Satellite for unexcused causes attributable to Contractor), Contractor shall continue to earn Orbital Performance Incentives at the same Daily Rate as Contractor would be earning such incentives if the Satellite were in service.

13.5 Total Loss of the Satellite During Orbital Performance Incentive Period

In the event that the Satellite becomes a Total Loss during the Orbital Performance Incentive Period, and such Total Loss is not attributable to Purchaser or Purchaser's representatives, consultants or subcontractors as set forth in Article 13.10, Contractor shall not be entitled to earn any further Orbital Performance Incentives with respect to such Satellite and shall have no further claim against Purchaser regarding such Orbital Performance Incentives. Contractor shall not lose any Orbital Performance Incentives earned prior to time that the Satellite becomes a Total Loss as provided above.

13.6 Total Loss or Partial Loss Not Attributable to Contractor

If, after Intentional Ignition, the Satellite is determined to be a Total Loss, or if a Partial Loss has occurred (unless and to the extent of a Terminated Ignition as provided in Article 12.1) and, in either case, such loss is due to causes not attributable to Contractor

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

(for avoidance of doubt, causes attributable to the Launch Agency shall not be considered causes attributable to Contractor), Contractor shall be deemed to have earned immediately the corresponding amount of Orbital Performance Incentives that would otherwise be lost pursuant to this Article 13, provided that Contractor shall only be entitled to the interest payments set forth in Article 13.9 through the date of payment for the corresponding amounts of Total Loss or Partial Loss.

13.7 Purchaser Operation of the Satellite

If, the Satellite operates in a manner that is not in accordance with any requirements of Exhibit B, Satellite Performance Specification, and to the extent such operation is due to any act or omission on the part of Purchaser or Purchaser's representatives, consultants or subcontractors in the operation of, testing of, or communication with, the Satellite (unless such act or omission was performed in accordance with directions or instructions provided by Contractor), Contractor shall continue to earn Orbital Performance Incentives with respect to such Satellite at the rate that applied prior to the act or omission resulting in degraded performance (subject to later adjustments pursuant to this Article with respect to failures to meet the requirements of Exhibit B, Satellite Performance Specification, not resulting from any such act or omission on the part of Purchaser or Purchaser's representatives, consultants or subcontractors).

13.8 Contractor Access

During the Orbital Performance Incentive Period, Contractor shall have reasonable access to applicable performance data of the Satellite for purposes of evaluating any degradation in the performance of the Satellite.

13.9 Payment

Payment of amounts earned under this Article 13 plus interest thereon calculated from the first day of the Orbital Performance Incentive Period until the date of payment, at an annual rate of [****] percent ([****]%), compounded monthly, shall be paid as follows: (a) the first payment shall be due no later than the end of the first full month following the month in which occurred successful completion of In-Orbit Testing, and shall cover the first partial month of earned Orbital Performance Incentives, if any; and (b) thereafter, all

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

subsequent payments shall be due no later than thirty (30) days following the end of each calendar month for the prior month's earned Orbital Performance Incentives.

13.10 Complete Loss of Unearned Incentives

Contractor shall no longer be entitled to earn any remaining unearned Orbital Performance Incentives if: (a) Purchaser permanently withdraws the Satellite from operational service or is unable to operate the Satellite for the purpose of operating a broadband internet subscriber service, prior to the end of the Stated Life, due to performance problems attributable to the Contractor; (b) pointing errors exceed the values set forth in Exhibit B, Satellite Performance Specification by more than [***] percent ([***]%) and Contractor is unable to fix such errors using Reasonable Efforts; or (c) Contractor has lost or is reasonably predicted to lose more than [***] percent ([***]%) of the Orbital Performance Incentives.

13.11 Ground Storage

If after Satellite Pre-Shipment Review of a Satellite, Purchaser places such Satellite in Ground Storage for a period of [***] for reasons not primarily due to the fault of Contractor, Purchaser shall pay Contractor interest on the full amount of the Orbital Performance Incentives at an annual rate of [***] percent ([***]%), compounded monthly, commencing on the first day of the seventh month of Ground Storage and ending upon the date of shipment of the Satellite to the Launch Site.

13.12 Insurance on Orbital Performance Incentives

Contractor agrees that it will not obtain insurance covering the Orbital Performance Incentives.

13.13 Contention with Canadian Beam Transponders

Purchaser acknowledges that Contractor is entitled to earn additional orbital performance incentives under the Loral Space Payload Contract based on the performance of the Transponders allocated to the Loral Space Payload. Nevertheless, in the event that the overall Satellite experiences any Anomaly which will affect the use of the Satellite or one (1) or more Transponders, Purchaser has the sole right to

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

determine which portion of the Satellite or Transponder(s) shall be affected, which decision shall be binding hereunder, and Purchaser shall advise Contractor accordingly. Contractor agrees to reflect Purchaser's rights contained in this Article 13.13 in the Loral Space Payload Contract.



SS/L-TP20701 ViaSat Contract

ARTICLE 14 — CORRECTIVE MEASURES IN THE SATELLITE AND OTHER DELIVERABLE ITEMS

If the data available from any satellite manufactured by Contractor (whether in-orbit or on the ground) or other information known to Contractor shows that the unlaunched Satellite to be delivered under this Contract contains or is predicted to contain a Defect, Contractor shall: (i) provide Purchaser prompt written notice thereof, provided Contractor shall not be required to disclose to Purchaser information that is confidential to any third party, and (ii) at Contractor's sole cost, correct such Defect or potential Defect notwithstanding that a payment may have been made in respect thereof, and regardless of prior reviews, inspections, or approvals. Contractor shall use its Reasonable Efforts to avoid and minimize delays as provided in Article 3.4. If the data available from any satellite manufactured by Contractor (whether in-orbit or on the ground) or other information known to Contractor shows that the launched Satellite delivered under this Contract contains or is predicted to contain a Defect or Anomaly, Contractor shall proceed in accordance with Article 15.3.



SS/L-TP20701 ViaSat Contract

ARTICLE 15 — REPRESENTATIONS AND WARRANTIES

15.1 Representations

Each Party represents that, as of the EDC:

- 1. It is a corporation duly organized in the State of Delaware.
- 2. It has all requisite corporate power and authority to own and operate its material properties and assets and to carry on its business as now conducted in all material respects.
- 3. It is duly qualified to transact business and is in good standing in each jurisdiction in which the failure to so qualify would have a material adverse effect on the other Party.
- 4. It has all requisite corporate power and authority to enter into this Contract and to carry out the transactions contemplated by this Contract.
- 5. The execution, delivery and performance of this Contract and the consummation of the transactions contemplated by this Contract have been duly authorized by the requisite corporate action and do not conflict with any material agreement or obligation to which it is a party or which binds its assets.
- 6. This Contract is a valid and binding obligation, enforceable against each Party making this representation in accordance with its terms.
- 7. No suit, claim, action, arbitration, legal, administrative, or other proceeding is pending, or to the best of each Party's knowledge, threatened against it that would likely affect the validity or enforceability of this Contract, or the ability of each Party making this representation to fulfill its commitments hereunder in any material respect.

15.2 Terms and Period of Warranty

15.2.1 Satellite. Contractor warrants that, from completion of SPSR up to Intentional Ignition of the Launch Vehicle (unless and to the extent of a Terminated Ignition where Contractor reacquires title and risk of loss for the Satellite as provided in Article 12.1 (A), in which case the warranty obligation will reattach but, as to any damaged Component, only after repair and replacement of such damaged Component pursuant to Article 12.1 (B) and the warranty period shall continue through the subsequent Intentional Ignition), the Satellite shall be free of any Defects and shall be manufactured and will perform in conformity with applicable requirements of Exhibit B, Satellite Performance



SS/L-TP20701 ViaSat Contract

Specification (as modified by any waiver and/or deviation pursuant to Article 9.1.4). If the Satellite or any part thereof (including Software integrated into the Satellite) does not conform to the above warranty, Contractor shall at any time during the applicable warranty period and irrespective of prior inspections proceed in accordance with Article 9.1.8 hereof and Contractor's only liability under the preceding sentence shall be as and to the extent set forth in such Article 9.1.8.

15.2.2 DSS And SCF Equipment

(A) <u>DSS</u>. Contractor warrants that the DSS (including the Software integrated into the DSS) delivered under this Contract shall be manufactured and/or developed in conformity with the applicable requirements of Exhibit H (as modified by any waiver and/or deviation pursuant to Article 11.8) and will be free from Defects during the period commencing on the date of Acceptance of such DSS pursuant to Article 11 and ending on the later of the first anniversary of: (a) the Launch of the Satellite; or (b) Acceptance of the DSS.

During the period specified above, as Purchaser's sole and exclusive remedy for any non-conformance or defect in the DSS of which Purchaser notifies Contractor in writing, such non-conformance or Defect shall be remedied by Contractor at Contractor's expense by repair or replacement of the defective Component (at Contractor's election). For any such non-conformance or Defect, Contractor shall determine if repair or replacement is required to be performed at Contractor's plant. If required, Purchaser shall ship the DSS to Contractor's designated facility. Contractor shall be responsible for the cost of shipment (including transportation, transit insurance, taxes and/or duties), and the cost of return shipment (including transportation, transit insurance, taxes and duties) to Purchaser at the location designated in Article 3.1. Risk of loss for the DSS shall transfer to Contractor upon delivery of the DSS to the shipping carrier by Purchaser, and risk of loss to the DSS shall transfer again to Purchaser once such DSS is repaired or replaced pursuant to this Article 15.2.2, upon receipt thereof by Purchaser at the location designated therefor in Article 3.1. After the expiration of the period specified above and through the Stated Life, a failure of the software integrated into the DSS to meet the applicable requirements of Exhibit H, Dynamic Satellite Simulator Statement of Work and Functional Requirements Document will, if such failure results in



SS/L-TP20701 ViaSat Contract

a Satellite Anomaly, be resolved in the response to the Satellite Anomaly pursuant to Article 15.3, below.

(B) <u>SCF Equipment</u>. If Purchaser exercises its option set forth in Article 2.4, Contractor warrants that the SCF Equipment (for purposes of this Article 15.2.2, "SCF Equipment" shall include the Software integrated into the SCF and the Spares) delivered under this Contract shall be manufactured in conformity with the applicable requirements of Exhibit F (as modified by any waiver and/or deviation pursuant to Article 11.8) and will be free from Defects during the period commencing on the date of Acceptance of such SCF Equipment pursuant to Article 11 and ending on the later of the first anniversary of: (a) the Launch of the Satellite; or (b) Acceptance of the SCF Equipment.

During the period specified above, as Purchaser's sole and exclusive remedy for any non-conformance or defect in such SCF Equipment of which Purchaser notifies Contractor in writing, such non-conformance or Defect shall be remedied by Contractor at Contractor's expense by repair or replacement of the defective Component (at Contractor's election). For any such non-conformance or Defect, Contractor shall determine if repair or replacement is required to be performed at Contractor's plant. If required, Purchaser shall ship the SCF Equipment to Contractor's designated facility. Contractor shall be responsible for the cost of shipment (including transportation, transit insurance, taxes and/or duties), and the cost of return shipment (including transportation, transit insurance, taxes and duties) to Purchaser at the location designated in Article 3.1. Risk of loss for the SCF Equipment shall transfer to Contractor upon delivery of the SCF Equipment to the shipping carrier by Purchaser, and risk of loss to the SCF Equipment shall transfer again to Purchaser once such SCF Equipment is repaired or replaced pursuant to this Article 15.2.2, upon receipt thereof by Purchaser at the location designated therefor in Article 3.1. After the expiration of the period specified above and through the Stated Life, a failure of the software integrated into the SCF and the Spares to meet the applicable requirements of Exhibit F, Satellite Control Facility Requirement Specification will, if such failure results in a Satellite Anomaly, be resolved in the response to the Satellite Anomaly pursuant to Article 15.3, below.

15.2.3 <u>Training, Mission Operations Support Services and Launch Support Services</u>. Contractor warrants that the Training, Mission Operations Support Services and Launch Support Services Purchaser accepts pursuant to this Contract will, upon Acceptance, conform to the highest professional standards for



SS/L-TP20701 ViaSat Contract

the commercial satellite communications industry for work similar in type, scope, and complexity to the Work at the time such Work is provided. In the event Contractor breaches this warranty and is so notified by Purchaser within twenty (20) Business Days of Acceptance or as soon as possible where time is of the essence, as Purchaser's sole and exclusive remedy, Contractor shall correct the deficiencies in the provision of such Training, Launch Support Services and Mission Operations Support Services where it is possible to do so (e.g., Contractor cannot correct deficiencies in, or re-perform, Launch Support Services from and after Launch).

- 15.2.4 <u>Deliverable Data</u>. Upon Acceptance, each item of Deliverable Data shall be complete, up-to-date, and shall otherwise conform with the requirements of this Contract. In the event the Contractor breaches this warranty and is so notified by Purchaser, as Purchaser's sole and exclusive remedy, Contractor shall correct the deficiencies in the Deliverable Data and resubmit such Deliverable Data to Purchaser.
- 15.2.5 Software. In addition to the warranties set forth in Articles 15.2.2 (A) and (B), Contractor further represents and warrants that, through the Stated Life of the Satellite, Contractor shall: (i) demonstrate that all PC based-Software delivered hereunder is free of viruses or similar items by testing such Software using commercially reasonable anti-viral software; (ii) not introduce into any delivered Software, without Purchaser's prior written approval, any code that would have the effect of disabling or otherwise shutting down all or any portion of the delivered Software (an exception to the foregoing is licensed software that will be included in some delivered Software; such license software will prevent any unlicensed copies of delivered Software from running); and (iii) not seek to gain access to the Work through any special programming devices or methods, including trapdoors or backdoors, to bypass, without Purchaser's prior written approval, any Purchaser security measures protecting the Work. Although the Parties contemplate that all necessary software support will be provided pursuant to Article 15.3, below, Purchaser shall have the option to purchase additional maintenance and support of the Software at commercially reasonable terms and prices.



SS/L-TP20701 ViaSat Contract

15.2.6. <u>Title.</u> Contractor represents and warrants that it shall provide good and marketable title to all Deliverable Items for which title shall pass in accordance with Article 12 free and clear of any liens and encumbrances of any kind (except those, if any, of Purchaser), at the time title passes to Purchaser.

15.2.7. Reserved

15.2.8 <u>Disclaimer</u>. EXCEPT AND TO THE EXTENT EXPRESSLY PROVIDED IN THIS ARTICLE 15, CONTRACTOR HAS NOT MADE NOR DOES IT HEREBY MAKE ANY REPRESENTATION OR WARRANTY, WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION, ANY WARRANTY OF DESIGN, OPERATION, CONDITION, QUALITY, SUITABILITY OR MERCHANTABILITY OR FITNESS FOR USE OR FOR A PARTICULAR PURPOSE, ABSENCE OF LATENT OR OTHER DEFECTS, WHETHER OR NOT DISCOVERABLE, WITH REGARD TO THE SATELLITE, ANY OTHER DELIVERABLE ITEM OR ANY OTHER WORK. CONTRACTOR MAKES NO WARRANTY WITH RESPECT TO LAUNCH SERVICES OR THE CONDITION, USE OR PERFORMANCE OF ANY LAUNCH VEHICLE.

15.3 Satellite Anomalies

In addition to Purchaser's warranty remedy set forth in Article 15.2.1 hereof, Contractor shall investigate any Satellite Anomaly or Defect occurring on the Satellite after Launch and during the Stated Life of such Satellite and known to it or as notified in writing by Purchaser, and shall use Reasonable Efforts to correct such Anomaly or Defect as promptly as possible in accordance with the Satellite Anomaly resolution support services set forth in Paragraph 2.5.8.2 of Exhibit A, Statement of Work. Such Reasonable Efforts shall be conducted at Contractor's expense to resolve Defects or Anomalies by on-ground means, including software patches or updates, or transmission by Contractor of commands to the Satellite to eliminate or mitigate any adverse impact resulting from any such Anomalies, to establish work-around solutions, or to otherwise resolve such Defects or Anomalies. Contractor shall coordinate and consult with Purchaser concerning such on-ground resolution of Defects or Anomalies in the launched Satellite. If for any reason any such Satellite Defect or Anomaly cannot be or is not corrected as set forth above, and as a result thereof, such Satellite suffers any loss, including loss of Transponders or becomes a Total Loss, Purchaser shall look



SS/L-TP20701 ViaSat Contract

solely to its Launch and In-Orbit Insurance Policy and any subsequent insurance policy for the Satellite procured by Purchaser to compensate it for its loss and any consequences therefrom, and Contractor shall have no liability or obligation to Purchaser or any other person in respect of such loss, except as set forth in Article 13 and to provide the services set forth in 32, subject to the conditions and limitations of such Articles and in all cases subject to the limitation of liability stated in Article 27. In the event and to the extent that the occurrence of a Satellite Defect or Anomaly is due to causes attributable to Purchaser, or Purchaser's agents, contractors and subcontractors (not including Contractor), Purchaser shall pay Contractor the Actual Costs reasonably incurred of all services provided by Contractor associated with such Satellite Anomaly services, plus a markup of [***] percent ([****]%) of such costs, which costs and markup shall be invoiced and paid pursuant to the provisions of Article 5.

15.4 Use Conditions Not Covered by Warranty

With respect to the DSS and SCF Equipment (including the Software integrated into the SCF Equipment and the DSS, and Spares), the warranty under this Article 15 shall not apply to the extent that any adjustment, repair or parts replacement is required as a result, directly or indirectly, of accident not attributable to Contractor, unusual physical or electrical stress beyond the item's design tolerances, Purchaser or third party negligence or misuse, failure of environmental control prescribed in operations and maintenance manuals, or repair or alterations by any party other than Contractor without Contractor's advance written consent, or by causes other than normal and ordinary use. The warranty provided pursuant to this Article 15 is conditioned upon Contractor being given reasonable access, if required, to Deliverable Items delivered at Purchaser's facility or as otherwise directed by Purchaser in order to effect any repair or replacement thereof. If the Defect repaired or remedied by Contractor is not covered by the warranty provided pursuant to this Article 15, Purchaser shall pay Contractor the Actual Costs reasonably incurred of such repair or replacement plus a markup of [***] percent ([***]%) of such costs, which costs and markup shall be invoiced and paid pursuant to the provisions of Article 5.

*** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

15.5 <u>Subcontractor Warranties</u>

Contractor shall pass on or assign to Purchaser all applicable warranties on goods or services given by Subcontractors to the extent to which Contractor is permitted by the terms of its purchase contracts with such suppliers or manufacturers, and further to the extent that is such assignment does not interfere with Contractor's performance if its obligations hereunder.



SS/L-TP20701 ViaSat Contract

ARTICLE 16 — CHANGES

16.1 Change Orders

Purchaser may, at any time between the EDC and the completion of this Contract, direct a change within the general scope of this Contract (which includes the Loral Space Payload) in drawings, designs, specifications, method of shipment or packing, quantities of items to be furnished, place of Delivery, postpone Delivery under a stop-work order (but not more than, in the cumulative, eighteen (18) months), require additional Work, or direct the omission of Work. In the event Purchaser directs a change, Contractor shall: (a) immediately proceed with the Work as changed; and (b) within thirty (30) days of Purchaser's direction, submit to Purchaser a written proposal of the effect of such a change on the Firm Fixed Price, specification, payment plan, time required for performance and/or other affected terms and conditions, including any effect to the [***] (the "Proposal").

Prior to directing such change, Purchaser may issue a request to Contractor for a Proposal. Within thirty (30) days of receipt of such request for the Proposal, Contractor shall provide Purchaser with the Proposal.

If any change directed by Purchaser causes an increase or decrease in costs of, or the time required for, the performance of this Contract, and/or other terms and conditions, an equitable adjustment shall be made in the Firm Fixed Price, Delivery schedule, and/or other terms and conditions, this Contract shall be modified in writing accordingly. Firm Fixed Price adjustments shall account only for the net impact on the Actual Costs incurred by Contractor as a result of the change including a reasonable profit thereon not to exceed [***] percent ([***]%). Such Actual Costs shall be determined in accordance with Contractor's standard accounting practices, and, if requested by Purchaser, Contractor will provide reasonably detailed back-up data to support its claim for equitable adjustment. If the cost of supplies or materials made obsolete or excess as a result of a change is included in Contractor's claim for adjustment, Purchaser shall have the right to prescribe the manner of disposition of such supplies or materials. If Purchaser directs the omission of Work and, as a result, it causes an increase in the cost of the remaining Work to be performed hereunder, such increase in cost shall constitute an allowable element of Contractor's claim for equitable adjustment.

*** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Upon receipt of Contractor's Proposal, the Parties shall commence good faith negotiations concerning the appropriate equitable adjustment to the Contract. In the event that the Parties have not reached agreement on the amount and manner of Contractor's compensation within sixty (60) days of Purchaser's direction for change, Purchaser shall pay to Contractor the agreed-upon portions of the proposed changes to the Firm Fixed Price and the Payment Plan and Termination Liability Schedule, and, pending agreement or other resolution, deposit the balance of the amount set forth in the Proposal with a recognized escrow agent selected by Purchaser on standard terms. The Parties shall exert all good faith efforts to reach resolution of the appropriate equitable adjustment within (30) days after Purchaser makes such deposit into escrow.

This Article shall not apply to [***] set forth in Article [***].

** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 17 — FORCE MAJEURE

17.1 Force Majeure Defined

Contractor shall not be responsible for late Delivery or delay of the final completion date or nonperformance of its contractual obligations due to Force Majeure. Force Majeure shall be any event beyond the reasonable control of Contractor or its suppliers and subcontractors and shall include but not be limited to: (1) acts of God; (2) acts of a public enemy; (3) acts of a government in its sovereign capacity or subject to the DPAS regulations (including any action or inaction affecting the import or export of items); (4) war and warlike events; (5) catastrophic weather conditions such as hurricanes, tornadoes and typhoons; (6) fire, earthquakes, floods, epidemics, quarantine restrictions; (7) actions taken in response to GIDEP direction concerning technical problems generally affecting the telecommunications industry; and (8) sabotage, riot and embargoes. ("Force Majeure"). For an event to qualify for Force Majeure relief hereunder: (1) the event must be beyond the control and without fault or negligence of a Party or its subcontractors hereunder; (2) the resulting delay cannot be circumvented by Reasonable Efforts to establish work-around plans, payment of expedited fees, alternate sources, or other means; and (3) Contractor must provide Purchaser with written notice thereof as soon as possible but in no event later than within ten (10) Business Days after Contractor's performance hereunder is impacted by such an event. Notwithstanding anything to the foregoing, any failure by a Subcontractor to meet its obligations to Contractor, or any delay due to labor shortages, defective tooling, transportation difficulties, equipment failure or breakdowns, lockouts, or inability to obtain materials shall not constitute a Force Majeure event (except where such circumstance is itself caused by a Force Majeure event), and shall not relieve Contractor from meeting any of its obligations under this Contract. Contractor shall use Reasonable Efforts to minimize the effect of any Force Majeure event. In the event Contractor claims a Force Majeure event. Contractor's written notice called for above shall include a detailed description of the portion of the Work known to be affected by such delay, as well as a proposed work-around plan reasonably satisfactory to Purchaser. The work-around plan shall: (1) set forth Contractor's Reasonable Efforts to mitigate the effect of any such Force Majeure event and include a schedule for such mitigation; and (2) contain sufficient detail for Purchaser to be able to evaluate such plan. If appropriate, such work-around plan shall use work-around schedules, payment of expedited fees, twenty four (24) hour



SS/L-TP20701 ViaSat Contract

operations, and the use of alternate Subcontractors (to be approved by Purchaser if required under this Contract). In the event of a Force Majeure event, the Delivery requirement shall be extended for such period as is supported by the evidence provided; provided, however, the occurrence of a Force Majeure event shall in no event entitle Contractor to an increase in the price for the Work. Any adjustment of time appropriate under this Article shall be formalized promptly by the execution of a mutually acceptable amendment to this Contract.

17.2 Termination for Force Majeure

Purchaser may, upon written notice to Contractor, terminate this Contract immediately in whole or in part if and when the cumulative aggregate of Force Majeure events exceeds twelve (12) months. In the event of such a termination, Contractor shall submit a final invoice to Purchaser within sixty (60) days after the termination date, which shall specify the amount due to Contractor from Purchaser pursuant to this Article 17.2. Purchaser shall pay such invoice, to the extent undisputed by Purchaser, within sixty (60) days after receipt. Notwithstanding the foregoing, Purchaser may dispute the amount of such invoice by giving Contractor notice of such dispute within sixty (60) days after receipt of the invoice. Contractor's invoice shall be in an amount equal to the sum of: (i) for each Milestone accepted, the respective Milestone payment; and (ii) for each Milestone not accepted, a percentage of the Milestone payment, where such percentage equals the percentage of the Milestone completed, both less the sum of all amounts received by Contractor under this Contract. Concurrently with such payment, Contractor shall promptly deliver in the manner and to the extent directed by Purchaser all completed Work and work-in-progress to Purchaser or Purchaser's designee (including all applicable licenses and warranties). Alternatively, Purchaser may request Contractor to purchase any such Work or work-in-progress that Contractor believes can be re-used or use Reasonable Efforts to sell any such items, in which case Contractor shall remit any sales proceeds to Purchaser, less, where Contractor sells such items on behalf of Purchaser, a deduction for the Actual Costs of disposition (without mark-up, margin or administrative charges) reasonably incurred by Contractor for such efforts. Payment of the amount payable by Purchaser under this Article shall constitute a total discharge of Purchaser's liabilities to Contractor under this Contract. If this Contract is terminated as provided in this Article, Contractor shall protect and preserve property in the possession



SS/L-TP20701 ViaSat Contract

of Contractor in which Purchaser has an interest. Nothing in this paragraph shall limit Purchaser's rights to terminate this contract for convenience pursuant to Article 21.



SS/L-TP20701 ViaSat Contract

ARTICLE 18 — PURCHASER DELAY OF WORK

If the performance of all or any part of the Work required of Contractor under this Contract is materially delayed or interrupted by Purchaser's failure to perform its contractual obligations set forth in paragraph 2.6 of the SOW within the time specified in this Contract or within a reasonable time if no time is specified, or by an act or failure to act of Purchaser or its agents, consultants or subcontractors (including, without limitation, the Launch Agency) that unreasonably interferes with and/or materially delays Contractor's performance of its obligations under this Contract (each a "Purchaser Delay"), Purchaser shall be responsible for any additional Actual Costs incurred by Contractor as a result thereof, and the Firm Fixed Price shall be modified accordingly. In addition, Contractor shall be entitled to an equitable adjustment in performance requirements, Delivery schedules, and any other terms of this Contract affected by such Purchaser Delay. The Parties will formalize all such adjustments by executing an appropriate amendment to this Contract. To qualify as a Purchaser Delay hereunder, (a) Contractor must notify Purchaser in writing of any applicable Purchaser Delay no later than five (5) Business Days after Contractor becomes aware of a possible impact due to the Purchaser Delay; and (b) Contractor must use Reasonable Efforts to avoid and/or mitigate the effect of such Purchaser Delay.



SS/L-TP20701 ViaSat Contract

ARTICLE 19 — INTELLECTUAL PROPERTY INDEMNITY

19.1 Indemnification

Contractor, at its own expense, hereby agrees to defend or, at Contractor's sole option, to settle, and to indemnify and hold harmless Purchaser, and its respective shareholders, officers, directors and employees from and against any claim or suit based on an allegation that Contractor's performance of the Work hereunder or the manufacture of any Deliverable Item or any part thereof or the normal intended use, lease or sale of any Deliverable Item or any part thereof infringes any third party's Intellectual Property Right ("Intellectual Property Claim"), and shall pay any royalties and other liabilities adjudicated (or provided in settlement of the matter) to be owing to the third party claimant as well as costs and expenses incurred in defending or settling such Intellectual Property Claim. Contractor's obligations under this Article 19.1 shall be subject to the conditions to indemnification set forth in Article 20.3.

19.2 Infringing Equipment

If Contractor's performance of the Work or the manufacture of any Deliverable Item or any part thereof or the normal intended use, manufacture, lease sale, or other disposition of any Deliverable Item or any part thereof under this Contract is enjoined or otherwise prohibited as a result of an Intellectual Property Claim, Contractor shall, at its option and expense (i) resolve the matter so that the injunction or prohibition no longer pertains, (ii) procure for Purchaser the right to use the infringing item, and/or (iii) modify the infringing item so that it becomes non-infringing while remaining in compliance with the Performance Specification (as such may be modified or waived pursuant to Article 9.1.4 or Article 11.8, as applicable). Purchaser shall reasonably cooperate with Contractor to mitigate or remove any infringement. If Contractor is unable to accomplish (i), (ii) or (iii) as stated above, Purchaser shall have the right to terminate this Contract, in whole or in part, and receive [***].

19.3 Combinations and Modifications

Contractor shall have no liability under this Article 19 for any Intellectual Property Claim to the extent arising from (i) use of any Deliverable Item in combination with other items not provided, recommended, or approved by Contractor, or (ii) modifications of any

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Deliverable Item after Delivery by a person or entity other than Contractor unless authorized by written directive or instructions furnished by Contractor to Purchaser under this Contract or (iii) the manufacture, delivery or use of any Deliverable Item in compliance with the design, specification or instructions of Purchaser.

19.4 Sole Remedies

The remedies set forth in this Article 19 are Purchaser's sole and exclusive remedies for, or related to, any Intellectual Property Claim.



SS/L-TP20701 ViaSat Contract

ARTICLE 20 — INDEMNIFICATION

20.1 Contractor's Indemnities

- a. Contractor, at its own expense, shall defend, indemnify and hold harmless Purchaser and its respective shareholders, directors, officers and employees, from and against any losses, damages, and other liabilities adjudicated (or provided for in settlement of the matter) to be owing to a third party claimant as well as costs and expenses, including court costs and reasonable attorneys' fees (collectively, "Losses") incurred in connection with any third party claim or suit alleging personal injury, death, or damage to the property of a third party, but only if such Losses were caused by, or resulted from, a negligent act or omission or willful misconduct of Contractor or its employees or representatives. For the avoidance of doubt, the Satellite in any stage of manufacture or operation shall not be considered as property subject to coverage under this Article 20.1, and the CondoSat Associate is not considered a third party subject to coverage under this Article 20.1.
- b. Contractor, at its own expense, shall defend, indemnify and hold harmless Purchaser and its respective shareholders, directors, officers and employees, from and against all range support and de-stacking charges to the extent not otherwise covered under a Purchaser policy of insurance, if any (including without limitation charges assessed to de-erect, place and maintain in storage, reerect and re-process the Launch Vehicle and Satellite and to re-integrate the Satellite), levied against Purchaser by the Launch Agency for the Satellite due to Launch delays caused by Contractor and/or its Subcontractors, agents or representatives at any tier, or any of them, up to a cumulative maximum of [***] U.S. dollars (U.S. \$[***]). Purchaser shall use Reasonable Efforts to cause the Launch Agency not to incur range support costs or de-stack a Launch Vehicle after Purchaser has been notified by Contractor that such a delay may occur.
- c. Contractor, at its own expense, shall defend, indemnify and hold harmless Purchaser and its respective shareholders, directors, officers and employees, from and against all Losses in connection with claims arising out of, or relating to, Contractor's breaches of its obligations in Article 4.5.

^{**} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

20.2 Purchaser's Indemnity

Purchaser, at its own expense, shall defend, indemnify and hold harmless Contractor, and its respective directors, officers and employees from and against any Losses incurred in connection with any third party claim or suit alleging personal injury, death, or damage to the property of a third party claimant, but only if such Losses were caused by, or resulted from, negligent acts or omissions or willful misconduct of Purchaser or its employees or representatives.

20.3 Conditions to Indemnification

The right to any indemnity specified in Article 19, Article 20.1 and Article 20.2 shall be subject to the following conditions:

- A. The Party seeking indemnification shall promptly advise the other Party in writing of the filing of any suit or of any written claim upon receipt thereof and shall provide the other Party, at its request and at the indemnifying Party's expense, with copies of all documentation and reasonable assistance relevant to such suit or claim. Notwithstanding anything to the foregoing, a delay in providing written notice of a suit or written claim shall not relieve a Party from its indemnity obligations unless such delay materially prejudices a Party's ability to defend the suit or claim.
- B. The Party seeking indemnification shall not reach a compromise or settlement without the prior written approval of the other Party, which approval shall not be unreasonably withheld or delayed, provided that the indemnifying Party has agreed in writing and without limitation to defend the applicable claim.
- C. The indemnifying Party shall assume the defense of any claim or suit thereof and shall satisfy any judgments rendered by a court of competent jurisdiction in such suits and shall make all settlement payments. The Party seeking indemnification may participate in any defense at its own expense, using counsel reasonably acceptable to the indemnifying Party.



SS/L-TP20701 ViaSat Contract

20.4 Waiver of Subrogation

Each Party shall use Reasonable Efforts to obtain a waiver of subrogation and release of any right of recovery against the other Party and its Affiliates, contractors and subcontractors at any tier (including suppliers of any kind) and their respective directors, officers, employees, shareholders and agents, that are involved in the performance of this Contract and from any insurer providing coverage for the risks subject to indemnification by the insured Party under this Article 20.



SS/L-TP20701 ViaSat Contract

ARTICLE 21 — TERMINATION FOR CONVENIENCE

21.1 Reimbursement of Contractor

- 21.1.1 Right to Terminate. Purchaser may terminate this Contract without cause, in whole or in part, upon giving Contractor written notice; provided, however, Purchaser may not terminate this Contract after Delivery of the Satellite in accordance with Article 3.2. Upon receipt of Purchaser's notice of termination for convenience, Contractor shall: (i) stop the terminated Work under this Contract; (ii) place no further orders or subcontracts for materials, services or facilities to the extent that they relate to the performance of the terminated Work; (iii) terminate orders and subcontracts to the extent that they relate to the performance of the terminated Work; and (iv) settle all outstanding liabilities and all claims arising out of such termination of orders and subcontracts.
- 21.1.2 Termination Liability. If Purchaser terminates Work pursuant to this Article 21, Contractor shall be entitled to the Actual Costs reasonably incurred by Contractor with respect to: (a) terminated Work performed hereunder, plus a reasonable profit not to exceed [***] percent on such Actual Costs; and (b) Actual Costs of termination and settlement with all vendors and subcontractors (provided that Contractor shall undertake Reasonable Efforts to minimize any such costs). In the event of termination by the Purchaser of any of the Work, it is agreed that the termination charges shall be negotiated pursuant to Article 21.1.3 below, but shall not exceed the lesser of: (i) the Firm Fixed Price; or (ii) the termination liability amount (in the "ViaSat TL" column) corresponding to the amount of the most recently completed Milestone on Exhibit E, Payment Plan and Termination Liability Schedule, and in the case of both (i) and (ii) less the amount of any payments made by Purchaser to Contractor pursuant to this Contract. Notwithstanding the above, in the event of a termination in part as provided above, if such termination causes an increase in the Actual Costs of performing the remaining non-terminated Work, Contractor shall be entitled to a corresponding increase (including a [***] percent profit) in the price of the non-terminated Work.
- 21.1.3. Determination of Termination Liability. Contractor shall submit a proposal to

^{**} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Purchaser within sixty (60) days after the effectiveness of the termination, which proposal shall specify the amount due to Contractor from Purchaser pursuant to this Article (the "Final Statement"). The Final Statement shall be a complete statement of all amounts that are due to Contractor hereunder as of the termination date for the terminated Work. In the event that Purchaser is not in agreement with the Final Statement, the Parties shall negotiate an agreed upon Final Statement. After agreeing on the Final Statement, Contractor shall invoice Purchaser shall pay such invoice within thirty (30) days after Purchaser's receipt of the invoice.

21.2 Disposition of Work

At the time of payment by Purchaser to Contractor of the termination liability amounts due under this Article, subject to applicable U.S. Government export laws and license conditions, Purchaser may direct Contractor to transfer to Purchaser, in the manner and to the extent directed by Purchaser, title to and risk of loss and possession of any items comprising the Work terminated (including all Work-in-progress, parts and materials, all inventories, licenses, and associated warranties but not including any portion of the Work to which Contractor would not have otherwise been obligated to transfer title hereunder had the Contract been completed). Contractor shall, upon direction of Purchaser, protect and preserve property at Purchaser's expense in the possession of Contractor or its Subcontractors in which Purchaser has an interest and shall facilitate access to and possession by Purchaser of items comprising all or part of the Work terminated. Alternatively, Purchaser may request Contractor to make Reasonable Efforts to re-use or sell such items and, in the case of Work Contractor can re-use, Contractor shall remit the mutually agreed cost of all such items to Purchaser, and, in the case of sold items, remit any sales proceeds to Purchaser less a deduction for Actual Costs of disposition reasonably incurred.

21.3 Sole Remedy

The remedy set forth in this Article shall be the sole remedy to which Contractor is entitled for termination of Work under this Article. In the event of a termination for convenience of the entire Contract, Payment of the amount payable by Purchaser to Contractor pursuant to this Article shall constitute a total discharge of Purchaser's liabilities to Contractor under this Contract, except for those liabilities and/or obligations that survive pursuant to Article 34.13.



SS/L-TP20701 ViaSat Contract

ARTICLE 22 — LIQUIDATED DAMAGES

22.1 Liquidated Damages - Schedule/Delivery

The Parties acknowledge and agree that failure to complete the Milestones listed below in this Article by the corresponding indicated dates may cause substantial financial loss to Purchaser (the "Key Milestone Dates"). The Parties further acknowledge and agree that the following liquidated damages represent a genuine and reasonable estimate of all losses that would be suffered by Purchaser by reason of any such delay (which losses would be difficult or impossible to calculate with certainty) in meeting the Key Milestone Dates.

For the purposes of this Article only, the Satellite will not be deemed "Delivered" if the Satellite is placed into Ground Storage for reasons primarily due to the fault of Contractor. In such case, the time period of such Ground Storage will be included for purposes of calculation of liquidated damages for delay under this Article 22; provided that Contractor shall not be liable for, or assessed any liquidated damages for, any period of such Ground Storage associated with any failure by Purchaser to schedule with the Launch Agency the next available Launch opportunity that is reasonably satisfactory to Purchaser and which is consistent with the Satellite's availability for Launch. If, as of the Satellite Delivery dates specified in Article 3.1 (as such date may be extended as provided in this Contract), the Launch Vehicle is unavailable for reasons that are not attributable to Contractor, or if the Satellite is delivered for purposes of Ground Storage for reasons that are not primarily due to the fault of Contractor, such that Purchaser is not materially harmed or prejudiced by a delay in Satellite Delivery, then the assessment of liquidated damages for failure to meet Key Milestone 4 (Satellite Delivery) shall be suspended/forgiven on a day-to day basis commensurate with the period of unavailability of the Launch Vehicle or storage after the Satellite Delivery date set forth in Article 3.1, as the case may be, and no payment by Contractor will be required for such period.

In the event Contractor fails to meet the Key Milestones Dates set forth below, as such Dates may be adjusted in accordance with this Contract, Contractor shall pay Purchaser, as liquidated damages and not as a penalty, up to a cumulative maximum of [***] U.S. dollars (U.S. \$[***]), the following incremental amounts with the damages period

*** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

beginning on the day after the applicable Key Milestone Date (as such Date(s) may be adjusted in accordance with this Contract):

1. Key Milestone 1: [***]

			Daily Amount of		
	Da	ays Late	Liquidated Damages	Period Total	Cumulative Total
1-30			[***]	[***]	[***]
31-60			[***]	[***]	[***]
61-90			[***]	[***]	[***]
2.	Key Milestone 2: [***]				

Days I	_ate	Liquidated Damages	Period Total	Cumulative Total
1-30		[***]	[***]	[***]
31-60		[***]	[***]	[***]
61-90		[***]	[***]	[***]

Daily Amount of

3. Key Milestone 3: [***]

	Daily Amount of		
Days Late	Liquidated Damages	Period Total	Cumulative Total
1-30	[***]	[***]	[***]
31-60	[***]	[***]	[***]
61-90	[***]	[***]	[***]

4. Key Milestone 4: [***]

	Daily Amount of		
Days Late	<u>Liquidated Damages</u>	Period Total	Cumulative Total
1-30	[***]	[***]	[***]
31-60	[***]	[***]	[***]
61-90	[***]	[***]	[***]
91-120	[***]	[***]	[***]
121-180	[***]	[***]	[***]

Any amounts due in accordance with this Article 22 shall be, at Purchaser's election, either (i) credited to Purchaser against any outstanding or future invoices hereunder or (ii) paid by Contractor to Purchaser within thirty (30) days of issuance of an invoice from Purchaser. Purchaser's failure to invoice or offset shall not constitute a waiver by Purchaser with respect to any amount of liquidated damages due and owing hereunder.

Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Notwithstanding the foregoing, in the event that liquidated damages are assessed against Contractor pursuant to the foregoing for delays in meeting Key Milestone Date 1 ([***]) and/or Key Milestone Date 2 ([***]) and/or Key Milestone Date 3 ([***]), but nevertheless Contractor Delivers the Satellite on or before the date set forth in Article 3.1 (as such date may be adjusted in accordance with this Contract), [***] percent ([***]%) of the liquidated damages assessed and paid pursuant to the foregoing shall be waived and forgiven, and Purchaser shall reimburse Contractor for any such liquidated damages actually paid.

22.2 <u>Liquidated Damages/Incentives Based Upon [***]</u>

22.2.1 [***]. Upon the earlier to occur of: (a) thirty-six (36) months after EDC, or (b) the conclusion of SPSR, the [***] of the U.S. Beams plus the Additional Beams (but not including the Canadian Beams) on the Satellite shall again be calculated using the data from SPSR and [***] (and the resulting calculated [***] being the "[***]") that has been placed in the escrow described in Article 22.2.2, and Contractor shall be entitled to an incentive payment or shall incur liquidated damamages as provided in the [***] table set forth below. In the event that the [***] is measured at thirty-six (36) months rather than at the conclusion of SPSR, the calculation shall be done based upon all available data and an engineering anaylsis performed in accordance with good engineering practices applying standards generally applicable in the aerospace industry. In determining [***], if Mexico service beams are included as any of the Additional Beams, then [***] contributed by such beam(s) will be considered in determining the [***]; provided that upon Purchaser obtaining all necessary landing rights and any other rights necessary for Purchaser to offer its intended service in Mexico, the [***] shall be recalculated and Contractor shall receive payment for the remaining [***] of the Mexican [***] within thirty (30) days of Purchaser's receipt of an invoice therefor. Purchaser shall have a good faith obligation to exert Reasonable Efforts to obtain such landing and other rights to provide its intended service in Mexico.

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Contractor acknowledges that the Satellite [***], not including the Canadian Beams, will be calculated assuming the use of the [***] frequencies and such [***] calculation shall not assume the use of the [***] frequencies. If the [***] is different than the then-current [***] Contractor shall be subject to liquidated damages or incentive payments as follows (with such amounts to be pro-rated):

[***]	Liquidated Damages or Incentive Payment		
Greater than [***], and greater than [***]	Incentive payment of [***] dollars (U.S. \$[***]) for each [***] (or portion thereof) greater than [***] (up to [***]), adjusted as follows: (i) [***] then the incentive payment shall be decreased by the amount of [***]; or (ii) [***] then the incentive payment shall be increased by the amount of [***]		
Greater than [***], but less than [***]	Liquidated damages of [***] U.S. dollars (U.S. $$[***]$) for each [***] (or portion thereof) difference between the [***] and the [***]		
Greater than [***], and the same as [***]	No incentive payments and no liquidated damages		

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Less than [***], but greater than [***]

Incentive payment of [***] dollars (U.S. [***]) for each [***] (or portion thereof) difference between the [***] and [***]

Less than [***], and less than [***]

Liquidated damages of [***] dollars (U.S. \$[***]) for each [***] (or portion thereof) less than [***], adjusted as follows:

(i) [***] then the liquidated damages shall be increased by the amount of [***]; or (ii) [***] then the liquidated damages shall be reduced by the amount of [***]

Less than [***], and the same as [***]

No liquidated damages and no incentive payments

Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Less than [***] but greater than [***], but [***] is greater than [***] or less than [***]

(i)[***] then the liquidated damages shall be assessed in the amount of [***] or (ii) [***] then an incentive payment shall be made in the amount of [***]

For example, if [***] is [***] and [***] is [***], Contractor would be obligated to pay Purchaser [***]. For another example, if [***] is [***] and [***] is [***], Purchaser would be obligated to pay Contractor [***]. For another example, if [***] is [***] and [***] is [***], Contractor would be obligated to pay Purchaser liquidated damages equal to [***]. For another example if [***] is [***] and [***] is [***], Purchaser would be obligated to pay Contractor an incentive payment equal to [***]. For another example [***] is [***] and [***] is [***], Contractor would be obligated to pay Purchaser liquidated damages equal to [***].

In the event that Contractor owes liquidated damages to Purchaser pursuant to the operation of this Article 22.2.1, at Purchaser's election, such liquidated damages shall either be (i) credited to Purchaser against any outstanding or future invoices hereunder or (ii) paid by Contractor to Purchaser within thirty (30) days of issuance of an invoice from Purchaser. In the event that Purchaser owes Contractor an incentive payment pursuant to operation of this Article 22.2.1, Purchaser shall make such payment to Contractor within thirty (30) days of issuance of an invoice from Contractor.

22.2.2 <u>Escrow of [***].</u> For the purpose of ensuring that all [***] made hereunder are done with the version of the [***] utilized to establish the [***], Purchaser shall, no later than fifteen (15) days after establishing the [***] pursuant to Article 4.6,

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

place the [***] used to establish the [***] into escrow pursuant to the escrow agreement attached hereto as Exhibit J. The version of the [***] placed into escrow shall not be updated, unless otherwise agreed by the Parties. The Parties shall share equally all costs associated with establishing and maintaining the escrow.

- 22.2.3. Contractor's Access to [***] and License. Subject to the terms of the license set forth below, Contractor shall have reasonable access to Purchaser's [***] until the determination of [***] as provided in Article 22.2.2 above (at which time the license described in this Paragraph shall also automatically terminate). Purchaser hereby grants to Contractor a non-exclusive, non-transferable, non-assignable, non-sublicensable, revocable, limited license to possess and use a single copy of the [***] during the time period described above solely for purposes of Contractor's measurement of the [***] of the Satellite to be provided under this Contract. Contractor agrees that, upon termination of the license, Contractor will either destroy (or permanently erase) all copies of the [***], or return the original [***] to Purchaser. Purchaser makes no warranties, express or implied, with respect to the [****].
- 22.2.4. <u>Limitations on Use.</u> Contractor acknowledges that Purchaser retains exclusive ownership of all Intellectual Property Rights in Purchaser's [***]. Contractor is not granted any rights in Purchaser's [***] other than the license rights expressly set forth above. Contractor agrees that it will not modify, adapt, create a derivative work of, merge, translate, decompile, disassemble, or otherwise reverse engineer Purchaser's [***].

22.3 Remedy

Contractor and Purchaser agree that the actual damages that Purchaser would suffer as a result of the late Delivery of the Satellite [***] are, in the nature of this transaction, difficult and impracticable to fix, that the liquidated damages set forth herein are reasonable amounts to compensate Purchaser for such failures and delays, and that the payment of liquidated damages is in lieu of payment of actual damages for such failures and delays. Such liquidated damages shall be Purchaser's sole and exclusive remedy and compensation for Contractor delays with respect to late Delivery of Deliverable

*** Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Items [***], provided that the [***] does not drop below [***]; and further provided, however, Purchaser retains all rights and remedies under Article 17.2 regarding termination for Force Majeure, Article 21 regarding termination for convenience and Article 23 regarding termination for default. Contractor hereby waives, to the extent permitted by applicable law, any defense as to the validity of any liquidated damages in this Contract on the grounds that such liquidated damages are void as penalties.

22.4 Incentives for Early Delivery

If Delivery of a Satellite occurs before the date specified in Article 3.1 (as such date may be extended as provided in this Contract), then Purchaser shall pay Contractor, within thirty (30) days of receipt of invoice, an amount not to exceed [***] U.S. dollars (U.S. \$[***]), as specified below, as an early delivery incentive for Contractor's early Delivery:

	Daily Amount of		
Days Early	Incentives	Period Total	Cumulative Total
1-30	[***]	[***]	[***]
31-60	[***]	[***]	[***]
61-90	[***]	[***]	[***]
91-120	[***]	[***]	[***]
121-180	[***]	[***]	[***]

^{***} Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 23 — TERMINATION FOR DEFAULT

23.1 Purchaser Rights of Termination

- 23.1.1 Right to Terminate. Subject to Article 17, Article 18 and Article 23.1.4 below, Purchaser may terminate this Contract in whole or in part by written notice to Contractor if: (i) Contractor fails to Deliver the Satellite within the time specified therefor in Article 3.1 (as such date may be extended in accordance with the terms of this Contract) plus [***] days (or such longer time as may be agreed to in writing by Purchaser); (ii) Contractor fails to make progress (including due to failure to meet specification requirements) so as to endanger performance of this Contract that it is reasonably likely Contractor will fail to Deliver the Satellite within the time period set forth in subclause (i) immediately above, and fails, within [***] days (or such longer period as may be agreed to in writing by Purchaser) after receipt from Purchaser of written notice thereof, to cure such breach or correct such failure; or (iii) except for failures covered by items (i) and (ii) above, Contractor fails to perform any material provision of this Contract and does not correct such failure within a period of [****] days (or such longer period as Purchaser may authorize in writing) after receipt of notice from Purchaser specifying such failure.
- 23.1.2 Termination Liability. In the event of termination pursuant to this Article 23.1, Purchaser shall be entitled to: (i) a refund by Contractor of all payments made by Purchaser to Contractor for the terminated Work plus interest thereon at the interest rate stipulated in Article 5.3.1 hereof from the date payment was received by Contractor to the date the refund is made to Purchaser; and/or (ii) to retain and/or obtain Work as set forth in Article 23.1.3, provided that Purchaser shall remain liable for payment of all amounts for such Work as set forth in Article 23.1.3. In addition, Purchaser shall be entitled to any liquidated damages for delay accrued pursuant to Article 22 prior to the effective date of termination. In addition, Purchaser shall be entitled to [***]. Contractor shall pay the foregoing amounts no later than thirty (30) days after Contractor's receipt of Purchaser's written notice requesting such amounts pursuant to subclause (i) above.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Payment of such amounts, or transfer of Work as set forth in Article 23.1.3, shall be Purchaser's sole and exclusive remedy in the event of a termination for default.

- 23.1.3 Contractor's Reimbursement for Terminated Work. In the event Purchaser terminates the Contract pursuant to this Article 23.1, Purchaser shall have the option to obtain title to the Deliverable Items provided for under this Contract or associated Work-in-progress (but not including Deliverable Data) or any other portion of the Work to which Contractor would not have otherwise been obligated to transfer title hereunder had the Contract been completed, including, if requested by Purchaser in writing (and subject to any approvals required by the associated Subcontractors and receipt by Contractor of associated reasonable releases and indemnifications in favor of Contractor), assignment of Subcontracts, subject to payment: (i) at the price set forth in this Contract for delivered items for which an itemized price is set forth herein and (ii) at the Actual Costs incurred by Contractor for (a) such items for which no itemized price is set forth herein and (b) partially completed items or services and Work-in-progress. This sum shall be offset by amounts paid by Purchaser and received by Contractor for Deliverable Items and Work-in-progress retained by Purchaser, and Contractor shall protect and preserve property in the possession of Contractor in which Purchaser has an interest. Purchaser shall pay amounts due hereunder no later than sixty (60) days after issuance of its written notice pursuant to Article 23.1.2. Upon settlement and payment of amounts due hereunder, Contractor, subject to applicable U.S. Government export control laws, shall promptly, at Contractor's or Subcontractor's plant, transfer title and risk of loss to Purchaser for the applicable Deliverable Items and Work-in-progress.
- 23.1.4 <u>Special Provision Limiting Purchaser's Remedies</u>. Purchaser's sole and exclusive remedy with respect to delays in Delivery shall be as specified in Article 22 and this Article 23.1. Purchaser shall have no right to terminate this Contract pursuant to this Article 23.1 above on or after Intentional Ignition (unless and to the extent of a Terminated Ignition pursuant to Article 12.1).
- 23.1.5 <u>Disposition of the Work</u>. Except for items obtained by Purchaser as set forth in Article 23.1.3, Contractor shall retain title to any and all Work, Work-in-progress,



SS/L-TP20701 ViaSat Contract

- parts or other material, inventories, and any associated warranties, and any subcontracted items Contractor has specifically produced, acquired, or subcontracted for in accordance with this Contract.
- 23.1.6 Invalid Default Termination. If, after termination pursuant to this Article 23.1, it is finally determined pursuant to Article 25 or written agreement of Purchaser that Contractor was not in default under Article 23.1.1, or that the default was excusable under the Contract, the rights and obligations of the Parties shall be the same as if the termination had occurred under Article 21.

23.2 Contractor Rights of Termination

- 23.2.1 Right to Terminate. Contractor may terminate this Contract upon written notice to Purchaser if Purchaser fails to make any payment required under this Contract, and Purchaser shall fail to cure such failure within thirty (30) days after receiving written notice thereof from Contractor (or such longer period as may be agreed to in writing by Contractor). For purposes of clarification, Contractor shall not be entitled to terminate this Contract for Purchaser's failure to make payment hereunder to the extent Purchaser has disputed such payment in good faith pursuant to Article 5.6.
- 23.2.2 <u>Termination Liability</u>. In the event of termination pursuant to this Article 23.2, Contractor shall be paid as if such termination were for convenience pursuant to Article 21. Further, if Purchaser's failure to perform is a failure to pay Contractor invoiced amounts when due, Contractor shall be entitled to late payment interest pursuant to Article 5.3. Payment of the total amounts payable by Purchaser pursuant to this Article 23.2.2 shall constitute a total discharge of Purchaser's liabilities to Contractor for termination pursuant to this Article 23.2.
- 23.2.3 <u>Disposition of the Work</u>. Upon completion of all payments to Contractor in accordance with this Article 23.2, Purchaser may, subject to Article 7 hereof, require Contractor to transfer to Purchaser in the manner and to the extent directed by Purchaser, title to and possession of any items comprising all or any part of the Work terminated (including all Work-in-progress, but not including any_Deliverable_Data_or_any_other portion of the Work to which Contractor would not have otherwise been obligated to transfer title hereunder had the Contract been completed) not used or disposed of by Contractor pursuant to the foregoing



SS/L-TP20701 ViaSat Contract

sentence. Contractor shall, upon direction of Purchaser, protect and preserve such items at Purchaser's expense in the possession of Contractor or its Subcontractors and shall facilitate access to and possession by Purchaser of items comprising all or part of the Work terminated. Alternatively, Purchaser may request Contractor to undertake Reasonable Efforts to re-use or sell such items and, in the case of Work Contractor can re-use, remit the agreed-upon cost of such items to Purchaser, and, in the case of sold items, remit any sales proceeds to Purchaser less a deduction for Actual Costs of disposition reasonably incurred by Contractor for such efforts. Contractor shall, if requested by Purchaser in writing, assign to Purchaser or its designee such Subcontracts as requested by Purchaser (subject to any approvals required by the associated Subcontractors and receipt by Contractor of associated reasonable releases and indemnifications in favor of Contractor).

23.2.4 <u>Invalid Default Termination</u>. If, after termination pursuant to this Article 23.2, it is finally determined pursuant to Article 25 or written agreement of Contractor that Purchaser was not in default under Article 23.2.1, Contractor shall be liable to Purchaser for direct damages resulting from such termination of this Contract (in no event exceeding amounts payable to Purchaser pursuant to Article 23.1, and subject to the limitation of liability set forth in Article 27).



SS/L-TP20701 ViaSat Contract

ARTICLE 24 — RESERVED [THIS SPACE INTENTIONALLY LEFT BLANK]



SS/L-TP20701 ViaSat Contract

ARTICLE 25 — DISPUTE RESOLUTION

Any dispute, claim, or controversy between the Parties arising out of or relating to this Contract ("Dispute"), including any Dispute with respect to the interpretation, performance, termination, or breach of this Contract or any provision thereof shall be resolved as provided in this Article 25.

25.1 Informal Dispute Resolution

Prior to the initiation of formal dispute resolution procedures, the Parties shall first attempt to resolve their Dispute informally, in a timely and cost-effective manner, as follows:

- A. If, during the course of the Work, a Party believes it has a Dispute with the other Party, the disputing Party shall give written notice thereof, which notice will describe the Dispute and may recommend corrective action to be taken by the other Party. Contractor's program manager shall promptly consult with Purchaser's contract manager in an effort to reach an agreement to resolve the Dispute.
- B In the event that agreement cannot be reached within ten (10) days of receipt of written notice, either Party may request that the Dispute be escalated, and the respective positions of the Parties shall be forwarded to an executive level higher than that under Paragraph A above for resolution of the Dispute.
- C. In the event agreement cannot be reached within twenty (20) days of receipt of written notice, either Party may request that the Dispute be escalated, and the respective positions of the Parties shall be forwarded to the Chief Executive Officer (CEO) or equivalent of each Party for resolution of the Dispute.
- D. In the event agreement cannot be reached as provided in Paragraphs A, B, or C above within a total of sixty (60) days after receipt of the written notice described in Paragraph A above either Party may proceed in accordance with Article 25.2.

Each phase of the dispute resolution process shall require the Parties to meet in person at least one time.



SS/L-TP20701 ViaSat Contract

25.2 Litigation

If any Dispute arising between the Parties cannot be settled pursuant to Article 25.1 (or, if a Party makes a good faith determination that (i) a breach by the other Party is such that a temporary restraining order or other preliminary injunctive relief to enforce its rights or the other Party's obligations under the provisions of this Contract is necessary or (ii) litigation is appropriate to avoid the expiration of an applicable limitations period or to preserve a superior position with respect to creditors), either Party shall have the right to bring suit.

Any suit brought shall be brought in any court of competent jurisdiction in the state of California, and the Parties hereby waive any objection to that venue and that court's exercise of personal jurisdiction over the case. The Parties hereby irrevocably consent to the exercise of personal jurisdiction by the state and federal courts in the State of California concerning any Dispute between the Parties. If, for any reason, neither the state nor federal courts in California will exercise jurisdiction over the Dispute, then litigation as permitted herein may be brought in any court of competent jurisdiction in the United States of America.

If a dispute arises as to whether or not a Party has committed or acted with gross negligence or willful misconduct, that issue alone shall be resolved by the court without a jury, and the court shall resolve such issue by applying the laws of the State of New York without regard to its conflict of law rules. THE PARTIES EXPRESSLY WAIVE THEIR RIGHT TO A JURY IN CONNECTION WITH SUCH DISPUTE.

Nothing in this Contract precludes a Party that prevails on any claim from initiating litigation in any appropriate forum to enter or enforce a judgment based on the court's award on that claim.

Pending final resolution of any dispute (including the informal dispute resolution process and litigation), Contractor shall, unless otherwise directed by Purchaser in writing, perform all its obligations under this Contract, provided that Purchaser continues to make undisputed payments as they come due. For purposes of clarification, Contractor shall not be entitled to stop work under this Contract for Purchaser's failure to make payment hereunder to the extent Purchaser has disputed such payments in good faith pursuant to Article 5.6. Without prejudice to Purchaser's other rights and remedies set



SS/L-TP20701 ViaSat Contract

forth in this Contract, in the event that Contractor stops work (unless otherwise permitted under this Contract) which results in delays in Delivery, [***].

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 26 — INTER-PARTY WAIVER OF LIABILITY FOR A LAUNCH

26.1 Launch Services Agreement Inter-Party Waiver of Liability

- 26.1.1 Inter-Party Waiver. Each Party hereby agrees to be bound by the no-fault, no-subrogation inter-party waiver of liability and related indemnity provisions required by the Launch Services Agreement with respect to the Launch and to cause their respective contractors and Subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract and any other person having an interest in the Satellite or any Transponder thereon (including customers of Purchaser) to accede to such waiver and indemnity, which in every case shall include claims against the Launch Agency, either Party and their respective contractors and subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract. The Parties shall execute and deliver any instrument that may be reasonably required by the Launch Agency to evidence their respective agreements to be bound by such waivers.
- 26.1.2 Waiver of Subrogation. The Parties also shall use Reasonable Efforts to obtain from their respective insurers, and shall require their respective contractors and subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract and any other person having an interest in the Satellite or any Transponder thereon (including non-consumer customers of Purchaser), to use Reasonable Efforts to obtain from their respective insurers, an express waiver of such insurers' rights of subrogation with respect to any and all claims that have been waived pursuant to this Article 26.

26.2 Indemnity Related to the Inter-Party Waiver of Liability

Each Party shall indemnify against and hold harmless the other Party and/or its contractors and Subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract, from and against any claim made by the indemnifying Party and/or any of its contractors and Subcontractors (including suppliers of any kind) that are involved in the performance of the Contract, or by any person having an interest in the Satellite or Transponder thereon (including customers of Purchaser), or by insurer(s) identified in Article 26.1, resulting from the failure of the



SS/L-TP20701 ViaSat Contract

indemnifying Party to waive any liability against, or to cause any other person the indemnifying Party is obligated to cause to waive any liability against, the Launch Agency, the other Party or either of their contractors and subcontractors at any tier (including suppliers of any kind) involved in the performance of this Contract. The Parties shall execute and deliver any instrument that may be reasonably required by the Launch Agency to evidence their respective agreements to be bound by such indemnifications.

26.3 Survival of Obligations

The waiver, indemnification and hold harmless obligations provided in this Article 26 shall survive and remain in full force and effect, notwithstanding the expiration or termination of this Contract.

26.4 Third Party Claims Coverage

With respect to third party liability for death or bodily injury of for the loss or damage to property that may be sustained, and any consequences thereof, resulting from, or arising in connection with the performance of the Launch Services for the Satellite, Purchaser shall use Reasonable Efforts to require the Launch Agency to include Contractor (and any other party or entity as Contractor may request) as an additional named insured under all policies of third party claims coverage (or any other program of third party claims coverage, including coverage provided by agencies of any government) that are provided or required to be provided by or through the Launch Agency.



SS/L-TP20701 ViaSat Contract

ARTICLE 27 — LIMITATION OF LIABILITY

27.1 Limitation

EXCEPT AS PROVIDED OTHERWISE IN ARTICLES 27.3 AND 27.4, NEITHER PARTY SHALL BE LIABLE DIRECTLY OR INDIRECTLY TO THE OTHER PARTY, TO ITS OFFICERS, DIRECTORS, EMPLOYEES, CONTRACTORS OR SUBCONTRACTORS AT ANY TIER (INCLUDING SUPPLIERS OF ANY KIND), AGENTS OR CUSTOMERS, TO ITS PERMITTED ASSIGNEES OR SUCCESSOR OWNERS OF ANY SATELLITE OR OTHER DELIVERABLE ITEM OR TO ANY OTHER PERSON CLAIMING BY OR THROUGH SUCH OTHER PARTY, FOR ANY AMOUNTS REPRESENTING LOSS OF PROFITS, LOSS OF BUSINESS, OR INDIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, CONSEQUENTIAL OR PUNITIVE DAMAGES, INCLUDING WITHOUT LIMITATION LOST PROFITS, LOST REVENUES OR COSTS OF RECOVERING A SATELLITE POST-LAUNCH, ARISING FROM OR RELATING TO THE PERFORMANCE OR NONPERFORMANCE OF THIS CONTRACT OR ANY ACTS OR OMISSIONS ASSOCIATED THEREWITH OR RELATED TO THE USE OF ANY ITEMS DELIVERED OR SERVICES FURNISHED HEREUNDER, WHETHER THE BASIS OF SUCH LIABILITY IS BREACH OF CONTRACT, TORT (INCLUDING NEGLIGENCE OF ANY TYPE (EXCEPT AS PROVIDED IN ARTICLE 27.3 BELOW) AND STRICT LIABILITY), STATUTE OR OTHER LEGAL OR EQUITABLE THEORY.

27.2 Liability

EXCEPT AS PROVIDED OTHERWISE IN ARTICLES 27.3 AND 27.4, IN NO EVENT WILL CONTRACTOR'S TOTAL LIABILITY ARISING OUT OF OR RELATED TO THIS CONTRACT EXCEED THE SUM OF: (i) ALL PAYMENTS PREVIOUSLY MADE TO CONTRACTOR UNDER THIS CONTRACT; AND (ii) ANY LIQUIDATED DAMAGES ACTUALLY INCURRED PURSUANT TO ARTICLE 22.1; AND (iii) [***]. EXCEPT AS PROVIDED OTHERWISE IN ARTICLE 27.3, IN NO EVENT WILL PURCHASER'S TOTAL LIABILITY ARISING OUT OF OR IN CONNECTION WITH THIS CONTRACT EXCEED [***].

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

27.3 Gross Negligence and Willful Misconduct

ARTICLE 27.1 SHALL NOT APPLY IN THE EVENT OF A PARTY'S GROSS NEGLIGENCE AND/OR WILLFUL MISCONDUCT IN WHICH CASE SUCH PARTY MAY BE LIABLE AND RESPONSIBLE FOR AMOUNTS REPRESENTING LOST PROFITS, LOSS OF BUSINESS AND THE OTHER DAMAGES DESCRIBED IN ARTICLE 27.1 UP TO AN AMOUNT NOT TO EXCEED IN THE CUMULATIVE MAXIMUM, [***] UNITED STATES DOLLARS (US\$[***]). THE FOREGOING AMOUNT SHALL BE IN EXCESS OF ANY LIABILITIES SUBJECT TO ARTICLE 27.2.

27.4 Indemnities

ARTICLES 27.1 AND 27.2 SHALL NOT APPLY TO LIABILITY ARISING UNDER ARTICLES 19 (INTELLECTUAL PROPERTY INDEMNITY), 20 (INDEMNIFICATION) (AND IN BOTH CASES ONLY AS TO LIABILITIES ADJUDICATED (OR PROVIDED IN SETTLEMENT OF THE MATTER) TO BE OWING TO THE THIRD PARTY CLAIMANT AS WELL AS COSTS AND EXPENSES INCURRED IN DEFENDING OR SETTLING SUCH CLAIMS), AND 26.2 (INDEMNITY RELATED TO THE INTER-PARTY WAIVER OF LIABILITY). NOTHING IN THIS ARTICLE 27 SHALL BE DEEMED AS A WAIVER OF ANY RIGHTS OF EITHER PARTY UNDER THE NONDISCLOSURE AGREEMENT DATED MARCH 8, 2006 (AS AMENDED ON APRIL 27, 2007).

27.5 Survival

THIS ARTICLE 27 SHALL SURVIVE THE EXPIRATION OR TERMINATION OF THIS CONTRACT FOR WHATEVER CAUSE.

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 28 — DISCLOSURE AND HANDLING OF PROPRIETARY INFORMATION

28.1 <u>Definition of Proprietary Information</u>

For the purpose of this Contract, "Proprietary Information" means all confidential and proprietary information (other than Deliverable Data, which is subject to the provisions of Article 39) in whatever form transmitted, that is disclosed or made available directly or indirectly by such Party (hereinafter referred to as the "disclosing party") to the other Party hereto (hereinafter referred to as the "receiving party") and: (i) is identified as proprietary by means of a written legend thereon or (ii) if disclosed orally, is identified as proprietary at the time of initial disclosure and then summarized in a written document, with the Proprietary Information specifically identified, that is supplied to the receiving party within ten (10) days of initial disclosure. Notes and memoranda prepared by the receiving party (but not the receiving party's attorneys) that include the disclosing party's Proprietary Information shall be considered the disclosing party's Proprietary Information for all purposes of this Article. Proprietary Information shall not include any information disclosed by a Party that (i) is already known to the receiving party at the time of its disclosure, as evidenced by written records of the receiving party, without an obligation of confidentiality at the time of disclosure; (ii) is or becomes publicly known through no wrongful act of the receiving party; (iii) is independently developed by the receiving party as evidenced by written records of the receiving party; or (iv) is rightfully obtained by the receiving party from any third party without restriction and without breach of any confidentiality obligation by such third party.

28.2 Terms for Handling and Use of Proprietary Information

Subject to Article 28.1, for a period of ten (10) years after receipt of any Proprietary Information, the receiving party shall not disclose Proprietary Information that it obtains from the disclosing party to any person or entity except its employees, Affiliates (who are not direct competitors of the disclosing party), attorneys, agents, Financing Entities, potential and actual joint venture partners, and consultants (who, in all cases, are not direct competitors of the disclosing party) who have a need to know, who have been informed of and have agreed in writing (or are otherwise subject to confidentiality obligations consistent with the obligations set forth herein) to abide by the receiving party's obligations under this Article 28, and who are authorized pursuant to applicable U.S. export control laws and licenses or other approvals to receive such information. The receiving party shall use not less than the same degree of care to avoid disclosure



SS/L-TP20701 ViaSat Contract

of such Proprietary Information as it uses for its own Proprietary Information of like importance; but in no event less than a reasonable degree of care. Proprietary Information shall be used only for the purpose of performing the obligations under this Contract, or as the disclosing party otherwise authorizes in writing.

28.3 Legally Required Disclosures

Notwithstanding the foregoing, in the event that the receiving party becomes legally compelled to disclose Proprietary Information of the disclosing party (including disclosures necessary or in good faith determined to be reasonably necessary under the Securities Act of 1933, as amended, and the Securities Exchange Act of 1934, as amended), the receiving party shall, to the extent practicable under the circumstances, provide the disclosing party with written notice thereof so that the disclosing party may seek a protective order or other appropriate remedy, or to allow the disclosing party to redact such portions of the Proprietary Information as the disclosing party deems appropriate. In any such event, the receiving party will disclose only such information as is legally required, and will cooperate with the disclosing party (at the disclosing party's expense) to obtain proprietary treatment for any Proprietary Information being disclosed.

28.4 Return of Confidential Information

Upon the request of the Party having proprietary rights to Proprietary Information, the other Party in possession of such Proprietary Information shall promptly return such Proprietary Information (and any copies, extracts, and summaries thereof) to the requesting Party, or, with the requesting Party's written consent, shall promptly destroy such materials (and any copies, extracts, and summaries thereof), except for one (1) copy which may be retained for legal archive purposes, and shall further provide the requesting Party with written confirmation of same; provided, however, where both Parties have proprietary rights in the same Proprietary Information, a Party shall not be required to return such information to the other Party. Nothing in this Article 28.4 shall require a Party to return or destroy computer files or records containing Proprietary Information but only if and to the extent such files or records were created in the ordinary course of business pursuant to such Party's automatic archiving and back-up procedures for computerized or word-processed records. The rights and obligations of the Parties under this Article shall survive any return or destruction of Proprietary Information.



SS/L-TP20701 ViaSat Contract

28.5 No License

Except as expressly provided in this Contract, nothing in this Contract shall be construed as granting the receiving party whether by implication, estoppel, or otherwise, any license or any right to use any Proprietary Information received from the disclosing party, or use any patent, trademark, or copyright now or hereafter owned or controlled by the disclosing party.

28.6 Injunctive Relief

The Parties agree that, in addition to any other rights and remedies that exist under this Contract, in the event of a breach or threatened breach of this Article, the disclosing party shall be entitled to seek an injunction prohibiting any such breach. The Parties acknowledge that Proprietary Information is valuable and unique and that disclosure in breach of this Article may result in irreparable injury to the disclosing party.



SS/L-TP20701 ViaSat Contract

ARTICLE 29 — CONTRACT TECHNOLOGY ESCROW

29.1 Segregation of Contract Technology/Escrow.

- 29.1.1. No later than 30 days after completion of PDR, the Parties shall enter into an escrow agreement (the "Escrow Agreement") in the form attached hereto as Exhibit G. Pursuant to the terms of the Escrow Agreement, Contractor shall deliver to the escrow agent copies of all Contract Technology, and shall diligently keep such escrow updated, no less frequently than on a quarterly basis. Purchaser shall pay all costs associated with establishing and maintaining the Escrow Agreement, and Contractor, at no charge to Purchaser, shall be responsible for gathering and depositing the Contract Technology into the escrow. Purchaser shall be responsible for all escrow agent fees incurred in connection with the withdrawal of Contract Technology under the Escrow Agreement, once established.
- 29.1.2 The Escrow Agreement will not permit Purchaser's access to the Contract Technology except if:
 - (i) Contractor becomes insolvent;
 - (ii) Contractor makes a general assignment for the benefit of creditors;
 - (iii) Contractor files a voluntary petition in bankruptcy or an involuntary petition in bankruptcy is filed against Contractor, which petition is not dismissed within 30 days;
 - (iv) Contractor suffers or permits the appointment of a receiver for its business;
 - (v) Contractor becomes subject to any proceeding under any bankruptcy or insolvency law;
 - (vi) Contractor has liquidated its business;
 - (vii) Contractor ceases doing business without providing for a successor; or
 - (viii) Contractor is unwilling or unable to continue to perform its obligations under this Contract.

In the cases of items (i) through (v), Purchaser shall only be able to access the Contract Technology if Contractor is unwilling or unable to continue to perform its obligations under this Contract.



SS/L-TP20701 ViaSat Contract

- 29.1.3 Contractor represents and warrants that the Contract Technology kept at Contractor's facilities and delivered into escrow under the Escrow Agreement:
 - (i) will comprise all of the Contract Technology (including the source code language statement for any and all applicable Software) to the extent its retention is consistent with Contractor's normal retention policies;
 - (i) will be kept current, including all updates needed to maintain compliance with the terms of this Contract; and
 - (ii) will be in a form suitable for reproduction by Purchaser.



SS/L-TP20701 ViaSat Contract

ARTICLE 30 — PUBLIC RELEASE OF INFORMATION

30.1 Generally

Either Party intending to disclose publicly, whether through the issuance of news releases, articles, brochures, advertisements, prepared speeches or other information releases, information concerning the financial details of this Contract or Proprietary Information of the other Party regarding the Work must obtain the prior written approval of the other Party with respect to the content and timing of such issuance, which approval shall not be unreasonably denied, delayed or withheld.

30.2 Exceptions

The obligations set forth in Article 30.1 shall not apply to the following:

- 30.2.1 information that is publicly available from any governmental agency or that is or otherwise becomes publicly available without breach of this Contract;
- 30.2.2 internal publications or releases which are clearly marked or otherwise identifiable as not intended for the public at large; and
- 30.2.3 disclosure required by applicable law or regulation, including without limitation, disclosure required by the Securities and Exchange Commission or the Nasdaq Stock Market or any other securities exchange on which the securities of a Party or its Affiliate is then trading.



SS/L-TP20701 ViaSat Contract

ARTICLE 31 — NOTICES

31.1 Written Notification

Each contractual or legal notice or correspondence required or permitted to be given or made hereunder shall be in writing (except where oral notice is specifically authorized) to the respective addresses, facsimile and telephone numbers and to the attention of the individuals set forth below, and any such notice or correspondence shall be deemed given on the earlier to occur of (i) actual receipt, irrespective of whether sent by post, facsimile transmission, overnight courier or other method, and (ii) seven (7) days after mailing by registered or certified mail, return receipt requested, postage prepaid.

In the case of Purchaser:

ViaSat, Inc. 6155 El Camino Real Carlsbad, California 92009 Attn: David Abrahamian Telephone No.: 760-476-3053 Facsimile No.: 760-795-1045

With a separately delivered copy to: ViaSat, Inc. 6155 El Camino Real Carlsbad, California 92009 Attn: Jared Flinn Telphone No.: 760-476-2672

Facsimile No.: 760-476-2672

In the case of Contractor:

Space Systems/Loral, Inc. 3825 Fabian Way M/S G-44 Palo Alto, CA 94303-4697 Attn.: Contract Manager, Nick Pound Telephone No.: 650-852-6606 Facsimile No.: 650-852-4631

With a separately delivered copy to: Space Systems/Loral, Inc. 3825 Fabian Way M/S G-56 Palo Alto, CA 94303-4697



SS/L-TP20701 ViaSat Contract

Attn.: Program Manager, Greg Harms Telephone No.: 650-852-5575 Facsimile No.: 650-852-6392

31.2 Change of Address

Either Party may from time to time change its notice address or the persons to be notified by giving the other Party written notice (as provided above) of such new information and the date upon which such change shall become effective.

31.3 Authorized Representatives

The only representatives of Purchaser and Contractor authorized to sign contractual documents and to direct Work under this Contract are:

PURCHASER

Name: David Abrahamian Title: Contracts Manager Name: Jared Flinn

Title: Associate General Counsel

Or others as may be authorized in writing by an executive officer of Purchaser from time to time.

CONTRACTOR:

Name: Ronald Haley

Title: Senior Vice President, Finance and Contracts
Or others authorized by written delegation by Contractor.



SS/L-TP20701 ViaSat Contract

ARTICLE 32 — RISK MANAGEMENT SERVICES

32.1 Purchaser Responsibility

Purchaser shall be responsible for procuring any Launch and In-Orbit Insurance Policy (defined below) covering the risks of loss or damage to the Satellite from and after Intentional Ignition.

IN NO EVENT SHALL PURCHASER DISCLOSE OR TRANSFER CONTRACTOR-PROVIDED TECHNICAL INFORMATION OR PROVIDE TECHNICAL/DEFENSE SERVICES BASED ON CONTRACTOR-FURNISHED TECHNICAL INFORMATION TO NON-U.S. PERSONS INCLUDING INSURANCE BROKERS OR UNDERWRITERS OR OTHER NON-U.S. PERSONS OR ENTITIES (AS DEFINED IN 22 CFR SECTION 120.15 AND SECTION 120.16) WITHOUT CONTRACTOR'S PRIOR WRITTEN APPROVAL (NOT TO BE UNREASONABLY WITHHELD OR DELAYED) AND, WHERE REQUIRED, PRIOR APPROVAL OF THE U.S. GOVERNMENT.

32.2 Contractor Support

At Purchaser's request, Contractor shall, at its own expense, use its Reasonable Efforts to obtain quotes for launch and/or in-orbit insurance to cover the risk of loss or damage to the Satellite from Intentional Ignition and continuing up to one year after Launch (provided that such coverage period is then commercially available) on such commercial terms and conditions as are then commercially available, with a maximum sum insured equal to the sum of prices for the Satellite, Launch Services and insured insurance premium, such amount to be confirmed by Purchaser (or, if such coverage is not then available, at the maximum amount commercially available) (the "Launch and In-Orbit Insurance Policy"). Within twenty (20) Business Days of receiving such quotes from Contractor, Purchaser, in its sole discretion, shall either: (i) directly procure such insurance by and on its own behalf; or (ii) notify Contractor in writing that Purchaser rejects such proposal for Launch and In-Orbit Insurance Policy. Failure of Purchaser to provide such written notice within the twenty (20) Business Day period shall be deemed to be a rejection of such proposal for the Policy. In either event, Contractor shall have no further obligation with regard to the procurement of Launch and In-Orbit Insurance Policy or Purchaser's acquisition or non-acquisition thereof, except as provided in the immediately succeeding Paragraph and Article 32.3.



SS/L-TP20701 ViaSat Contract

At Purchaser's request, Contractor shall, regardless of whether Purchaser shall have accepted or rejected Contractor's proposal for the Launch and In-Orbit Insurance Policy, perform the various activities described in this Article 32. Subject to Article 7 and Article 28, Contractor, at no additional charge to Purchaser, shall furnish Purchaser with such information regarding the Satellite as is requested by the insurers and as is customary and normal to support and assist Purchaser in obtaining and maintaining a Launch and In-Orbit Insurance Policy including but not limited to: (i) providing a comprehensive presentation package on the Satellite, suitable for presentation to the space insurance brokers and underwriters; (ii) supporting Purchaser with all necessary presentations (oral, written or otherwise), including attendance and participation in such presentations where requested by Purchaser; (iii) providing on a timely basis all reasonable and appropriate technical information, data and documentation; (iv) providing documentation and answers to insurer and underwriter inquiries; and (v) obtaining any agreements and other approvals that are required (e.g., those agreements and approvals required pursuant to Article 7.3) for Purchaser's potential insurance providers to have access to all information required by such potential providers. Notwithstanding Article 28, Disclosure and Handling of Proprietary Information and Article 30, Public Release of Information, but subject to Article 7, Purchaser may disclose this Contract to its insurers, provided that Purchaser has entered into binding agreements with such insurers that limits the disclosure and use of such Contract on terms comparable to those contained herein and further provided that Contractor shall be allowed to make redactions reasonable to both Parties to Article 19 (Intellectual Property Indemnity), Article 22 (Liquidated Damages for Late Delivery), Article 23 (Termination for Default), Article 27 (Limitation of Liability), Article 29 (Contract Technology Escrow), Article 35 (Ground Storage), Article 36 (Orbit Maneuver Life Payment and Launch Vehicle Guarantee), Article 39 (Intellectual Property), Article 40 (Lender Requirements) (and definitions applicable to the foregoing), interest percentages, and this sentence.

32.3 Insurance Amounts

Purchaser shall be responsible for procuring directly and on its own behalf any Launch and In-Orbit Insurance Policy. Notwithstanding the foregoing, in the event that any loss is paid to Contractor under the Launch and In-Orbit Insurance Policy, then Contractor agrees to hold such amounts in trust for Purchaser and to pay such amounts to Purchaser within three (3) Business Days after Contractor receives the corresponding



SS/L-TP20701 ViaSat Contract

payment from the insurers, except for any amount(s) due to Contractor including earned Orbital Performance Incentives as provided under Article 13, and, for the avoidance of doubt, Contractor shall be obligated to make payment to Purchaser under this Article 32 only to the extent that Contractor receives payment from the insurers. Notwithstanding the foregoing, Contractor shall reasonably cooperate with Purchaser in recovering all amounts due from the insurers under the Launch and In-Orbit Insurance Policy.

32.4 Claims Support

Subject to Article 7 and Article 28, Contractor shall cooperate with and provide reasonable and customary support to Purchaser in making and perfecting claims for insurance recovery and as to any legal proceeding as may be brought by Purchaser associated with any claim for insurance recovery. Contractor shall furnish Purchaser with any information that may be reasonably required to prepare and present any insurance claim.

32.5 Contractor Insurance Requirements

During the period from EDC until Intentional Ignition (and in the event of a Terminated Ignition, until the subsequent Intentional Ignition where Contractor re-acquires title and risk of loss to the Satellite as provided in Article 12.1) Contractor shall obtain and maintain, at its own expense, Ground Insurance coverage against all risks of loss, including earthquake and other natural disasters, and damage to the Satellite and its Components in an amount sufficient to cover the greater of: (i) the Contractor's full replacement value of the Satellite; and (ii) the amounts paid by Purchaser with respect to the Satellite. Such insurance shall provide: (w) coverage for removal of debris, and insuring the structures, machines, equipment, facilities, fixtures and other properties constituting part of the project, (x) transit coverage, including ocean marine coverage (unless insured by the supplier), (y) off-site coverage for any key equipment, and (z) off-site coverage covering any property or equipment not stored on the construction site. The deductible for such insurance shall not exceed [***] Dollars U.S. dollars (U.S. \$[***]). Contractor shall have Purchaser and/or its designees named as an additional named insured on such insurance policy(ies) to the extent of their interest. Prior to commencing the Work, and whenever requested by Purchaser, Contractor agrees to furnish to

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Purchaser certificates of insurance evidencing that insurance required under this Article is in full force and effect. Contractor agrees to notify Purchaser in advance of any change to any material terms and conditions of said policies of insurance which are relevant to this Contract, and, if any such change is made without Purchaser's consent (not to be unreasonably withheld), Contractor shall be obligated to procure supplemental insurance coverage, subject to availability, to comply with the insurance obligations set forth above. Contractor shall provide certificates of insurance which shall contain an endorsement setting forth that the insurer cannot terminate or materially amend the provisions of the insurance without prior written notification to Purchaser at least thirty (30) days before such termination or amendment. If, after being requested in writing by Purchaser to do so, Contractor fails to produce evidence of compliance with Contractor's insurance obligations hereunder within fourteen (14) days, Purchaser may effect and maintain the insurance and pay the premiums, and Contractor, at no charge, shall provide reasonable cooperation as requested by Purchaser. The amount paid shall be a debt due and payable from Contractor to Purchaser, or Purchaser, at its option, may elect to offset payments due Contractor.



SS/L-TP20701 ViaSat Contract

ARTICLE 33 — ORDER OF PRECEDENCE

In the event of conflict among the terms of this Contract (Preamble, Recitals, and Articles 1 to 41) and the Exhibits, the following order of decreasing precedence shall apply:

o Contract terms and conditions (Preamble, Recitals and Articles 1 through 41)

o Exhibit A Statement of Work

Payment Plan and Termination Liability Schedule

Default Configuration o Exhibit K o Exhibit B

Satellite Performance Specification

o Exhibit C Mission Assurance Plan Satellite Program Test Plan o Exhibit D

Satellite Control Facility Requirements Document o Exhibit F

Dynamic Satellite Simulator Statement of Work and Functional Requirements

Document

o Exhibit G **Escrow Agreement** o Exhibit I **Guaranty Agreement** o Exhibit J [***] Escrow Agreement

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



o Exhibit E

o Exhibit H

SS/L-TP20701 ViaSat Contract

ARTICLE 34 — GENERAL

34.1 Assignment

- 34.1.1 <u>General</u>. This Contract may not be assigned, either in whole or in part, by either Party without the express written approval of the other Party, not to be unreasonably withheld or delayed.
- 34.1.2 By Purchaser. Notwithstanding the foregoing, Purchaser may assign or transfer this Contract or all its rights, duties, or obligations hereunder without Contractor's approval (i) to an Affiliate, provided that such Affiliate has sufficient financial resources or funding to fulfill Purchaser's obligations under this Contract; (ii) in connection with obtaining financing for the Satellite under any financing agreement; (iii) to any entity which, by way of merger, consolidation, or any similar transaction involving the acquisition of substantially all the stock or the entire business assets of Purchaser relating to the subject matter of this Contract succeeds to the interests of Purchaser, provided in each case the assignee, transferee, or successor to Purchaser has expressly assumed all the obligations of Purchaser and all terms and conditions applicable to Purchaser under this Contract; or (iv) to any subsidiary or joint venture associate of Purchaser not meeting the requirements of item (i) above, provided that Purchaser executes the Guaranty Agreement attached hereto as Exhibit I.
- 34.1.3 <u>By Contractor</u>. Notwithstanding the foregoing, Contractor may assign or transfer this Contract or all of its rights, duties, or obligations hereunder to: (i) any Affiliate of Contractor that has equivalent or greater financial resources as Contractor; or (ii) any person in connection with the sale, transfer, merger, assignment or other reorganization affecting Contractor or all (or substantially all) of Contractor's assets or capital stock, whether by way of merger, consolidation, or otherwise, provided in each case the assignee, transferee, or successor to Contractor has expressly assumed all the obligations of Contractor and all terms and conditions applicable to Contractor under this Contract.
- 34.1.4 <u>Security Interests</u>. Either Party, upon prior written notice to the other Party, may grant security interests in its rights hereunder to lenders that provide financing for the performance by such Party of its obligations under this Contract or for the subject matter hereof. In the event that either Party is sold to or merged into



SS/L-TP20701 ViaSat Contract

another entity, its responsibilities under this Contract shall not be altered and the successor organization shall be liable for performance of such Party's obligations under this Contract.

34.2 Binding Effect

This Contract shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Assignment of this Contract shall not relieve the assigning Party of any of its obligations nor confer upon the assigning Party any rights except as provided in this Contract.

34.3 Severability

If any provision of this Contract is declared or found to be illegal, unenforceable or void, the Parties shall negotiate in good faith to agree upon a substitute provision that is legal and enforceable and is as nearly as possible consistent with the intentions underlying the original provision. If the remainder of this Contract is not materially affected by such declaration or finding and is capable of substantial performance, then the remainder shall be enforced to the extent permitted by law.

34.4 Waiver of Breach of Contract

A waiver of any provision or any breach of a provision of this Contract shall not be binding upon either Party unless the waiver is in writing, signed by a duly authorized representative of the Party to be bound, as applicable, and such waiver shall not affect the rights of the Party not in breach with respect to any other or future breach. No course of conduct by a Party shall constitute a waiver of any provision or any breach of a provision of this Contract unless a written waiver is executed in accordance with the provisions of this Article 34.4.

34.5 Amendments

This Contract, including any and all its Exhibits, may not be modified except by written instrument of subsequent date signed by an officer of Contractor, or another person designated in writing by any such officer to sign such an instrument, and an authorized officer of Purchaser, or another person designated in writing by any such authorized officer of Purchaser to sign such an instrument.



SS/L-TP20701 ViaSat Contract

34.6 Captions

The captions contained herein are for purposes of convenience only and shall not affect the construction of this Contract.

34.7 Relationships of the Parties

It is expressly understood that Contractor and Purchaser intend by this Contract to establish the relationship of independent contractors only, and do not intend to undertake the relationship of principal and agent or to create a joint venture or partnership or any other relationship, other than that of independent contractors, between them or their respective successors in interests. Neither Contractor nor Purchaser shall have any authority to create or assume, in the name or on behalf of the other Party, any obligation, expressed or implied, or to act or purport to act as the agent or the legally empowered representative of the other Party, for any purpose whatsoever.

34.8 Entire Agreement

This Contract, including all its Exhibits, represents the entire understanding and agreement between the Parties hereto with respect to the subject matter hereof, and supersedes all prior negotiations and agreements with respect to the subject matter hereof, which cease to have any further force or effect.

34.9 Standard of Conduct

Both Parties agree that all their actions in carrying out the provisions of this Contract shall be in compliance with applicable laws and regulations and neither Party will pay or accept bribes, kickbacks or other illegal payments, or engage in unlawful conduct.

34.10 Construction

This Contract, including all its Exhibits, has been drafted jointly by the Parties and in the event of any ambiguities in the language hereof, there shall be no inference drawn in favor of or against either Party.

34.11 Counterparts

This Contract may be signed in any number of counterparts with the same effect as if the signature(s) on each counterpart were upon the same instrument.



SS/L-TP20701 ViaSat Contract

34.12 Applicable Law

This Contract shall be interpreted, construed and governed, and the rights of the Parties shall be determined, in all respects, according to the laws of the State of New York without reference to its conflicts of laws rules.

34.13 Survival

Termination or expiration of this Contract for any reason shall not release either Party from any liabilities or obligations set forth in this Contract remain to be performed or by their nature would be intended to be applicable following any such termination or expiration, including the following: Article 2.2 (Option Satellite) (only if the option is still valid as of the termination or expiration of this Contract); Article 4.4 (Fees and Other Expenses); Article 4.5 (Taxes); Article 7.4 (Compliance with U.S. Laws and Directives); Article 12.1(C) (Satellite), Article 15.2.8 (Disclaimer); Article 19 (Intellectual Property Indemnity); Article 20 (Indemnification); Article 21 (Termination for Convenience); Article 22 (Liquidated Damages); Article 23 (Termination for Default); Article 25 (Dispute Resolution); Article 26 (Inter-Party Waiver of Liability for a Launch); Article 27 (Limitation of Liability); Article 28 (Disclosure and Handling of Proprietary Information); Article 29 (Contract Technology Escrow); Article 30 (Public Release of Information); Article 31 (Notices); Article 33 (Order of Precedence); Article 34 (General); Article 36 (Orbital Maneuver Life Payment and Launch Vehicle Guarantee) and Article 39 (Intellectual Property).

34.14 U.N. Convention on the International Sales of Goods

The U.N. Convention on the International Sales of Goods shall not apply or otherwise have any legal effect with respect to this Contract.

34.15 No Third-Party Beneficiaries

This Contract is entered into solely between, and may be enforced only by, Purchaser and Contractor and their permitted assigns, and this Contract shall not be deemed to create any rights in third parties, including suppliers, customers, the CondoSat Associate, and owners of a Party, or to create any obligations of a Party to any such third parties.



SS/L-TP20701 ViaSat Contract

34.16 Business Ethics

In the event that Contractor has cause to believe that Purchaser has acted improperly or unethically during the term of this Contract, Contractor is requested to report such conduct to Purchaser's ethics hotline at **1-888-475-8376**. Copies of Purchaser's Guide to Business Conduct are available at www.viasat.com under "Investors-Corporate Governance." Any failure by Contractor to act under this clause will not constitute a breach of this Contract.

34.17 Remedies

Where a specific remedy is set forth in this Contract, such remedy shall be the sole and exclusive remedy available to the Party entitled to claim it and shall be in substitution for all other remedies arising at law or in equity in relation to the specific matter that the remedy responds to. Purchaser's right to terminate this Contract for default pursuant to Article 23.1.1(iii) shall not constitute a "specific remedy" hereunder, provided that, where Purchaser elects to proceed pursuant to Article 23.1.1(iii), such election shall be the sole remedy to which Purchaser is entitled. For the avoidance of doubt, in the event of a material breach by either Party where multiple specific remedies are set forth herein and are available, the non-breaching Party shall have right to elect the specific remedy to enforce, provided that, in no event shall the non-breaching Party be entitled to multiple recoveries or remedies unless expressly provided otherwise in this Contract.



SS/L-TP20701 ViaSat Contract

ARTICLE 35 — GROUND STORAGE

35.1 Ground Storage

Purchaser may direct Contractor, in writing, to store the Satellite after completion of SPSR. If so directed, Contractor shall store the Satellite and related equipment at Contractor's own facilities for up to [***] years ("Ground Storage"). Such storage shall be conducted in accordance with Contractor's standard Shipping, Handling and Storage Plan (the "Satellite Storage Plan").

35.2 Costs of Ground Storage

[***] Purchaser shall pay Contractor for Ground Storage at the rate of \$[***] per month, and any post-storage re-verification testing excess of the testing that would have been required if the Satellite were only placed in Ground Storage for six (6) months will be considered a change under Article 16, Changes. The foregoing shall be Purchaser's sole liability for Storage Costs.

35.3 Equitable Allocation of Storage Costs

Notwithstanding the foregoing, in the event that Ground Storage is required primarily due to the fault of Contractor, all Storage Costs shall be borne by Contractor, provided that Purchaser schedules with the Launch Agency the next available Launch opportunity that is reasonably satisfactory to Purchaser and which is consistent with the Satellite's availability for Launch.

In the event the Satellite is initially placed in Ground Storage due to the fault of Contractor, but Purchaser does not schedule with the Launch Agency the next available Launch opportunity reasonably satisfactory to Purchaser and which is consistent with the Satellite's availability for Launch: (A) Contractor shall be responsible for all Storage costs for the period of time up to the next available Launch opportunity reasonably satisfactory to Purchaser and which is consistent with the Satellite's availability for Launch, [***]; and (B) Purchaser shall be responsible for payment of those amounts set forth in paragraph 35.2(b), above, for the period of Ground Storage [***].

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

35.4 Payments of Storage Costs

Payments of Storage Costs required to be made under this Article 35 shall be made on the thirtieth (30th) day of each month for the prior month's storage, provided an invoice is received at least thirty (30) days prior to the payment date.

35.5 Title and Risk of Loss

Title and risk of loss to the Satellite delivered for Ground Storage shall remain with Contractor at the storage site.

35.6 Notification of Intention to Launch a Previously Stored Satellite

In the event a Satellite is placed in Ground Storage, Purchaser shall notify Contractor in writing when the Satellite should be removed from Ground Storage and delivered to the Launch Site. This notification must be received by Contractor not less than three (3) months prior to the scheduled date for Launch of the Satellite.



SS/L-TP20701 ViaSat Contract

ARTICLE 36 — ORBIT MANEUVER [***] AND LAUNCH VEHICLE [***]

36.1 Orbital Maneuver [***]

On the IOT Complete Date, assuming a nominal Launch (as such term is used in the launch industry), the remaining orbital maneuver life (the "Remaining OML") of the Satellite shall be calculated by taking the fuel loaded at Launch, subtracting the fuel consumed to date, and projecting the consumption of the available fuel. Fuel consumed to date shall be calculated by taking the average of: [***]. Projected consumption shall be based on sufficient fuel for the station keeping operations projected by Purchaser (based upon procedures provided by Contractor pursuant to this Contract), assuming full north/south and east/west station keeping, altitude control, momentum management, and a reasonable reserve for retirement maneuvers. For purposes of this calculation, any fuel consumed associated with the Satellite temporarily occupying an orbit location other than the final orbit location, 115°W, pursuant to a "fly-by", or with drifting the Satellite to its final orbit location of 115°W due to the Satellite being tested in an orbit location other than the final orbit location of 115°W, shall not be subtracted for calculation of the Remaining OML.

If, at the time of the IOT Complete Date, the Remaining OML of the Satellite is less than one hundred eighty (180) months due to the fault of Contractor, Contractor shall pay to Purchaser an amount calculated as [***] dollars (\$[***]) per each month by which such Remaining OML is less than one hundred eighty (180) months (with such amount to be pro-rated for any period of less than one (1) month) (the "OML Payment"). For example, if the Remaining OML were 178 months and 10 days (for a total of 5,428 total days in the remaining Orbital Performance Period), the Contractor would be obligated to pay Purchaser [***].

In the event Contractor is required to pay the OML Payment, the Orbital Performance Incentive Period shall be shortened to end on the last day of the Remaining OML period,

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

and the Daily Rate shall be adjusted to account for such shortened period. In the above example, the Orbital Performance Incentive Period would be 5,428 days and the new Daily Rate for purposes of Article 13 would be the Orbital Performance Incentive amount divided by 5,428.

Any amounts due in accordance with this Article shall, at Purchaser's election either be (i) credited to Purchaser against any outstanding or future invoices hereunder or (ii) paid by Contractor to Purchaser within thirty (30) days of issuance of an invoice from Purchaser.

36.2 Launch Vehicle [***]

Contractor shall [***], based upon pre-launch estimates and analysis, [***]. If, as of the completion of the critical design review, [***], following a Launch [***], would be [***], then Contractor shall be [***]. Contractor will perform all necessary Integration Activities [***]. In the event that the Satellite has to be placed in Ground Storage at the conclusion of SPSR as a result of a change of Launch Vehicle due to Contractor failing to meet the [***] as provided above, the liquidated damages provisions of Article 22.1 shall apply during the term of such Ground Storage.

***Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

ARTICLE 37 — CONTRACTOR PERSONNEL

37.1 Contractor Personnel

The Contractor will assign only properly qualified and experienced personnel to the Work contemplated under the Contract, and Contractor shall use Reasonable Efforts to retain such personnel on the Work for its duration. At the reasonable request of Purchaser, Contractor shall not use, and shall not permit any Subcontractor to use, in the performance of the Work any personnel deemed by Purchaser to be abusive, disorderly, incompetent, careless, unqualified to perform the Work assigned, or otherwise unsatisfactory to Purchaser. Without limiting the generality of the foregoing, Contractor shall, within ten (10) Business Days after receiving notice from Purchaser, remove from the performance of the Work, and, as soon as is reasonably practicable, replace, any personnel of Contractor or any Subcontractor who is performing any portion of the Work, if Purchaser reasonably believes that such personnel is creating a risk to the timely or safe completion of the Work in accordance with this Contract.

37.2 Key Personnel as of EDC

Personnel assigned to the following positions shall be considered "Key Personnel," and, as of the date of EDC, shall be filled by the following Contractor employees:

- a) the Contractor's Program Manager: Greg Harms
- b) the Contractor's Contracts Manager: Nick Pound
- c) the Contractor's Systems Engineering Manager: Michel Baylocq
- d) the Contractor's Payload Manager: Melles Tsighe

Purchaser may from time to time change the positions designated as Key Personnel under this Contract on sixty (60) days notice to Contractor and with mutual agreement of the Contractor, not to be unreasonably withheld or delayed.

37.3 Assignment of Key Personnel

Contractor shall assign individuals only from within Contractor's organization to fill the Key Personnel positions. All Key Personnel shall have significant relevant experience and expertise. Before assigning an individual to any Key Personnel position, whether as an initial assignment or a subsequent assignment, Contractor shall notify Purchaser of



SS/L-TP20701 ViaSat Contract

the proposed assignment, shall introduce the individual to Purchaser (and, upon request of Purchaser, provide Purchaser with the opportunity to interview the individual) and shall provide Purchaser with a resume and other information reasonably requested by Purchaser. If Purchaser in good faith objects to the proposed assignment within fifteen (15) Business Days after being notified thereof, Contractor shall not assign the individual to that position and shall propose to Purchaser the assignment of another individual of suitable qualifications and experience, the criteria of which are to be mutually agreed by Purchaser and Contractor acting reasonably. Should the individuals filling the positions of Key Personnel leave such positions for whatever reason, Contractor shall follow the procedure set forth in this Article to assign replacement personnel. Key Personnel shall not be assigned to other duties without the Purchaser giving prior written consent.



SS/L-TP20701 ViaSat Contract

ARTICLE 38 — SUBCONTRACTS

38.1 Major Subcontracts

Contractor shall select Major Subcontractors subject to Purchaser's prior written approval, and Purchaser shall be provided with complete copies of all Major Subcontracts promptly upon execution thereof. For purposes of this Contract, a "Major Subcontract" shall be any Subcontract that has a contract value of at least five million dollars (\$5,000,000.00), or, in the event Contractor issues Subcontracts therefor, for those items set forth below. In the event Contractor or any Subcontractor desires to terminate any Major Subcontractor or to substitute Subcontractors on any Major Subcontract, Contractor shall first notify Purchaser in writing.

To the extent that the Contractor elects to procure the following items from subcontractors, those subcontractors shall be Major Subcontractors: TWTAs, LNAs, and downconverters.

38.2 No Privity of Contract

Subject to the provisions of this Article, Contractor shall have the right to use such Subcontractors as may be necessary to perform the Work under this Contract. Nothing in this Contract shall be construed as creating any contractual relationship between Purchaser and any Subcontractor. Contractor is fully responsible to Purchaser for the acts or omissions of Subcontractors and of any other parties used by Contractor or a Subcontractor in connection with the performance of the Work. Any failure by a Subcontractor to meet its obligations to Contractor shall not constitute a basis for Force Majeure (except where such failure is itself a Force Majeure event), and shall not relieve Contractor from meeting any of its obligations under this Contract. Notwithstanding anything to the contrary herein, Purchaser's acknowledgment or approval of any Major Subcontractor or Subcontractor shall not relieve Contractor from any of its obligations or responsibilities under this Contract.

38.3 Thales Agreement

Purchaser has previously entered into an agreement with Thales Components Corporation ("TCC") to reserve production capacity for forty (40) high-powered traveling wave tubes ("TWTs"), as well as mission unique long lead parts for use on the Satellite. Promptly upon execution of this Contract, but in no event later than January 30, 2008, Purchaser agrees to assign and relinquish to Contractor all its rights under its contract



SS/L-TP20701 ViaSat Contract

with TCC for the reservation of the necessary TWTs and associated mission unique long lead parts in favor of Contractor. In the event that the terms of Purchaser's contract with TCC prejudices or adversely affects Contractor's rights and ability to negotiate terms and conditions with TCC for the TWT's that would otherwise be available to Contractor, such differing terms shall be considered as changes subject to the provisions of Article 16.



SS/L-TP20701 ViaSat Contract

ARTICLE 39 — INTELLECTUAL PROPERTY

39.1 Ownership

- A. Reserved.
- B. Subject to the licenses granted to Purchaser in this Article 39 of this Contract, Contractor shall own all Intellectual Property Rights, title and interest in and to all Contractor Intellectual Property. For avoidance of doubt, Contractor shall retain title to all Deliverable Data (including Software in Deliverable Items of hardware and Deliverable Items of Software) developed by or on behalf of Contractor and the Intellectual Property Rights contained therein, excluding Purchaser Intellectual Property.
- C. Contractor agrees that the obligations set forth in Article 28.2, subject to the terms and exceptions to confidentiality obligations set forth in Article 28, shall apply to Contractor with respect to the handling and use of Purchaser-unique information contained in the Satellite Payload Specifications; provided that to the extent such Purchaser-unique information constitutes a trade secret under applicable law, such Purchaser-unique information shall be held in confidence by Contractor in accordance with Article 28 for as long as such information remains a trade secret under applicable law

39.2 Licenses and Restrictions

- A. Contractor grants to Purchaser a nonexclusive, perpetual, irrevocable, royalty-free, license, with the right to sublicense, except to Competitors, subject to the same limitations applicable to Purchaser (including but not limited to the provisions of Articles 7.2, 7.4, 15.1, 15.2.8, 17.1, 26.2, 27, 28, 30 and 39.1), to Exploit Contractor Intellectual Property. Any sublicense granted under this Article 39.2(A) shall be in writing and Contractor shall be provided with complete copies of any such sublicense promptly upon grant thereof. Purchaser shall be liable for the acts of its sublicensees to the same extent as if such acts of sublicensees had been committed by Purchaser.
- B. Subject to the terms of this Contract, Purchaser grants to Contractor a nonexclusive, nontransferable, royalty-free license during the term of this



SS/L-TP20701 ViaSat Contract

Contract, with no right to sublicense (except to Subcontractors), under Purchaser Intellectual Property only to the extent necessary to provide the Work under this Contract. Any sublicense granted under this Article 39.2(C) shall be in writing. Contractor shall be liable for the acts of its sublicensees to the same extent as if such acts of sublicensees had been committed by Contractor.

- C. Neither Party shall have a license to use the trademarks or service marks of the other Party without such other Party's express written consent.
- D. There are no implied licenses under this Contract, and any rights not expressly granted by a Party to the other Party hereunder shall be reserved by such Party. Each Party agrees not to reverse engineer the Intellectual Property of the other Party provided to such Party in connection herewith (whether provided prior to or after EDC). Each Party agrees not to file for patents covering the Intellectual Property Rights owned by the other Party hereto.
- E. Contractor shall make Reasonable Efforts to secure for Purchaser from each Major Subcontractor entering into a Subcontract on or after EDC at least the same rights with respect to such Subcontractor's Intellectual Property as are provided to Purchaser in this Article 39 with respect to Contractor Intellectual Property (such Reasonable Efforts not to include payment of additional amounts for such rights).



SS/L-TP20701 ViaSat Contract

ARTICLE 40 — LENDER REQUIREMENTS

The Parties recognize this Contract may be financed through external sources. Notwithstanding anything to the contrary in this Contract, except for restrictions and conditions set forth in Article 7 and Article 28, Contractor shall provide to any Financing Entity any information (including, without limitation, this Contract) that such Financing Entity reasonably requires. Contractor agrees to negotiate in good faith and issue such documents as may be reasonably required by any Financing Entity to implement such financing, including a contingent assignment of this Contract to such Financing Entity, under terms reasonably acceptable to Contractor, but in no event shall Contractor be obligated to agree to anything (including agreement to make modifications to this Contract) that would impair, create a risk to, or otherwise prejudice its rights and benefits hereunder or increase its liabilities or obligations hereunder.



SS/L-TP20701 ViaSat Contract

ARTICLE 41 — SECURITY INTEREST

41.1 Security Interest

- 41.1.1 Collateral. As collateral security for payment and performance by Contractor of the Secured Obligations, whether now existing or hereafter arising, Contractor hereby grants to Purchaser a first priority security interest (the "Security Interest") in all of Contractor's right, title and interest in, to and under the following property, whether owned on EDC or thereafter acquired by Contractor and whether existing on EDC or thereafter coming into existence (collectively, the "Collateral"):
 - a. all [***]; and
 - b. all [***] of and to any of the foregoing.
- 41.1.2 Contractor Actions in Support of Grant of Security Interest. In furtherance of the grant of the Security Interest, Contractor, as of EDC:
 - a. shall give, execute, deliver, file, record, obtain, and authorize all financing statements, intellectual property security agreements, and any other notices, instruments, agreements and documents, and Contractor shall take such other actions, as reasonably requested by Purchaser to (1) create, perfect, validate and preserve the Security Interest and the priority thereof or (2) to enable Purchaser to exercise and enforce its rights hereunder with respect to such pledge, grant and Security Interest, and, in the event the Protocol of Space Assets to the Cape Town Convention on Mobile Goods is ratified by the United States and becomes effective, Contractor shall take all action required thereunder to protect and preserve the security interest of Purchaser hereunder:
 - b. shall indicate in its corporate records the Security Interest that Purchaser has in the Collateral; and
 - c. authorizes Purchaser to file UCC-1 financing statements with respect to the Security Interest and the Collateral.

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

- 41.1.3 Contractor Representations and Warranties in Support of Grant of Security Interest. Contractor represents and warrants that as of EDC the Security Interest granted to Purchaser shall constitute a legal, valid and enforceable security interest therein and upon the filing of UCC-1 financing statements in the office of the Secretary of State of Delaware will be a first priority perfected security interest in those items that can be perfected by filing under the UCC subject to no other liens or security interests other than Junior Liens (as defined below) [***]. Contractor further represents and warrants that it is duly organized as a corporation solely under the laws of the State of Delaware, and that its full legal name is set forth is as set forth on the signature pages hereof. Contractor further represents and warrants that no authorization, approval, or other action by, and no notice to or filing with, any U.S. governmental authority or regulatory body is required for either (i) the pledge or grant created in favor of Purchaser hereunder, or (ii) the exercise by Purchaser of any rights or remedies in respect of any Collateral (whether specifically granted or created hereunder or provided for by applicable law), except for the UCC-1 financing statement in the State of Delaware, any filings under federal statutes in respect of Intellectual Property Rights, any filings to perfect security interests under laws outside the United States, any filings to perfect security interests under provisions other than the UCC, and any actions in connection with enforcement rights required under applicable export restrictions and security regulations.
- 41.1.4 <u>Junior Liens</u>. On and after EDC, Contractor shall not sell, lend, rent, lease, transfer or otherwise dispose of any portion of the Collateral or any right, title or interest therein except to Purchaser pursuant to the Contract and Contractor shall keep such Collateral free from any security interest, lien, encumbrance or claim. Notwithstanding the foregoing, Contractor shall be permitted to grant [***] ("Junior Liens") on the Collateral to secure indebtedness of Contractor and its Affiliates from time to time, including reimbursement obligations in respect of letters of credit (collectively, "Permitted Debt"), provided that the terms of such Junior Liens shall be expressly subordinated to the Security Interest and otherwise acceptable to

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Purchaser. Prior to granting any Junior Lien on the Collateral, Contractor shall provide written notice to Purchaser describing the circumstances of such Junior Lien (including the identity of the proposed holder of such Junior Lien and the nature of the Permitted Debt). Within ten (10) Business Days after receipt of such notice, Purchaser shall notify Contractor whether such proposed Junior Lien is acceptable to Purchaser in its reasonable discretion and, if acceptable, provide the terms for such Junior Lien and the subordination thereof that would be reasonably acceptable to Purchaser. Contractor agrees and acknowledges that such terms may prohibit the holder of such Junior Lien from seizing, foreclosing on, selling or otherwise disposing of the Collateral so long as the Security Interest in favor of Purchaser in regard to such Collateral remains in effect.

- 41.1.5 <u>Periodic Inventory</u>. Upon the reasonable request of Purchaser (not to occur more often than [***] absent a default under Article 23.1.1, and in the event of a default under Article 23.1.1 on a [***] basis) following EDC, Contractor shall provide Purchaser a written inventory describing the [***] as of the time of such request.
- 41.1.6 Exercise of Rights. On and after EDC, Purchaser shall have and be entitled to exercise all the rights and remedies with respect to the Collateral of a secured party under the UCC (whether or not the UCC is in effect in the jurisdiction where Purchaser asserts the rights and remedies), such additional rights and remedies as provided in this Contract, and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where Purchaser may assert its rights and remedies. Effective at the moment at which Purchaser exercises its right to terminate the Contract pursuant to Article 23.1.1, Purchaser shall have the right, subject to applicable export control restrictions and security regulations, to take possession of the Collateral or any part thereof upon ten (10) days notice to Contractor, and Contractor shall assemble the Collateral in the location or locations specified by Purchaser and give Purchaser access to Contractor's premises for purposes of inspecting and/or removing any or all of the Collateral. Notwithstanding any provision of this Contract to the contrary,

^{***}Certain information on this page has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions



SS/L-TP20701 ViaSat Contract

Purchaser shall not be permitted to exercise its remedies as a secured party in respect of the Security Interests pursuant to this Article 41.1.6 unless Purchaser has the right to terminate the Contract as to the Work pursuant to Article 23.1.1 hereof, and Purchaser exercises such rights of termination.

- 41.1.7 Sale of Collateral. If the Proceeds of sale, collection or other realization of or upon the Collateral are insufficient to cover the Secured Obligatins, Contractor shall remain liable for any deficiency. Purchaser shall not incur any liability as a result of the sale of the Collateral, or any part thereof, at any private sale conducted in a commercially reasonable manner and otherwise in compliance with the UCC and applicable export control restrictions. Contractor hereby waives any claims against Purchaser arising by reason of the fact that the price at which the Collateral may have been sold at such a private sale was less than the price that might have been obtained at a public sale or was less than the aggregate amount of the Secured Obligations, even if Purchaser accepts the first offer received and does not offer the Collateral to more than one offeree, so long as the sale was conducted in a commercially reasonable manner. Purchaser may be the purchaser of any or all of the Collateral at any public or private sale (to the extent any portion of the Collateral being privately sold is of a kind that is customarily sold on a recognized market or subject of widely distributed standard price quotations) in accordance with the UCC, at a price as determined in accordance with Article 23.1.3, and Purchaser shall be entitled, for the purpose of bidding and making settlement or payment of the purchase price for all or any part of the Collateral sold at any such sale made in accordance with the UCC, to use and apply any of the Secured Obligations as a credit on account of the purchase price for any Collateral payable by Purchaser at such sale. Purchaser may sell the Collateral without giving any warranties as to the Collateral. Purchaser may specifically disclaim or modify any warranties of title or the like. This procedure will not be considered to adversely affect the commercial reasonableness of any sale of the Collateral. Notwithstanding any Junior Liens, Purchaser shall have no obligation to marshal any of the Collateral.
- 41.1.8 <u>Application of Proceeds</u>. Except as otherwise herein expressly provided and except as provided below in this Article 41.1.8, the Proceeds of any collection, sale or other realization of all or any part of the Collateral pursuant hereto: (i) first, to the



SS/L-TP20701 ViaSat Contract

payment of the costs and expenses of such collection, sale or other realization; (ii) second, to the payment in full of the amounts due to Purchaser as set forth in Article 23.1.2, (iii) third, to satisfy any other Secured Obligations, (iv) fourth, to the payment of any Junior Liens secured by such Collateral, and (v) fifth, to the payment to the Contractor, or its respective successors or assigns, or as a court of competent jurisdiction may direct, of any surplus then remaining. Notwithstanding anything to the contrary in this Contract, the Proceeds from the sale, disposition or other realization of the Collateral may be applied to the payment of Secured Obligations only to the extent expressly set forth in this Article 41.1.8.

- 41.1.9 Appointment as Attorney in Fact. On and after EDC, without limiting any rights or powers granted by Article 41 to Purchaser while the Contract has not been terminated in whole for Contractor's default in accordance with Article 23.1.1, upon the occurrence and during the continuance of any termination of the Contract for Contractor's default pursuant to Article 23.1.1, Purchaser is hereby appointed the attorney in fact of the Contractor for the purpose of carrying out the provisions of the Security Interests and taking any action and executing any instruments that Purchaser may deem necessary or advisable to accomplish the purposes hereof, which appointment as attorney in fact is irrevocable and coupled with an interest.
- 41.1.10 Reasonable Supporting Actions. Contractor agrees that, from time to time upon the written request of Purchaser on and after EDC, Contractor will execute and deliver such further documents and do such other acts and things as Purchaser may reasonably request in order to fully effectuate the purposes of this Article 41. Contractor shall not change its name, address, or jurisdiction of incorporation unless it shall have given Purchaser thirty (30) day's prior written notice thereof and shall have taken all action reasonably requested by Purchaser to preserve the validity, perfection and priority of the security interests hereunder.
- 41.1.11 <u>Termination of Security Interest</u>. After a valid termination of this Contract pursuant to Article 21 or Article 23.2, or in the event of a termination of this Contract pursuant to Article 23.1.1, each of the foregoing in accordance with the respective terms of such provisions, and subject to receiving Purchaser's consent, which will not be unreasonably withheld, conditioned or delayed, Contractor may, on behalf of



SS/L-TP20701 ViaSat Contract

Purchaser, file any documents (including UCC termination statements) to effect a termination of the Security Interests and any document filed or recorded to perfect such Security Interests. Purchaser shall execute and deliver and, if appropriate, file with the applicable filing offices, such documents and instruments as may be necessary or desirable to effect such termination.



SS/L-TP20701 ViaSat Contract

IN WITNESS WHEREOF, the Parties have executed	d this Contract by their duly authorized officers as of the date	set forth in the Preambl
Space Systems/Loral, Inc.	ViaSat, Inc.	

Ву:	/s/ RON O. HALEY	By:	/s/ KEVEN LIPPERT
Name:	Ron Haley	Name:	Keven Lippert
Title:	CFO	Title:	General Counsel

BEAM SHARING AGREEMENT

THIS BEAM SHARING AGREEMENT is dated as of the 11th day of January, 2008 by and between ViaSat, Inc., a corporation organized and existing under the laws of the State of Delaware ("<u>ViaSat</u>") and Loral Space & Communications, Inc., a corporation organized and existing under the laws of the State of Delaware ("<u>Loral</u>").

WHEREAS, ViaSat and Space Systems/Loral, Inc., a Delaware corporation and an indirect, wholly-owned subsidiary of Loral ("SS/L") have entered into a Contract, dated as of January 7, 2008, for the construction, testing and purchase of the ViaSat-1 satellite (such contract, as may be amended, modified or supplemented from time to time, the "Satellite Contract"), and Loral and SS/L have entered into a Loral Payload Agreement, of even date herewith (such agreement, as may be amended, modified or supplemented from time to time, the "Loral Payload Agreement"), with respect to the Loral Payload, as hereinafter defined.

WHEREAS, the Parties desire to create an arrangement between ViaSat and Loral for such ViaSat-1 satellite (the "Satellite");

WHEREAS, SS/L has consented to the arrangement contemplated hereby, and the Satellite Contract and the Loral Payload Agreement contemplate the simultaneous transfer of title to and risk of loss of the Satellite to ViaSat and Loral, as their respective interests therein shall appear;

WHEREAS, Loral and Telesat Canada, a Canadian corporation ("<u>Telesat</u>") are entering into a separate agreement, dated today's date (such agreement, as may be amended, modified or supplemented from time to time, the "<u>Option Agreement</u>"), pursuant to which Telesat will have the option to assume and succeed to the rights and obligations of Loral hereunder, and for and in consideration of which Telesat is entering into the TT&C Agreement, dated today, with ViaSat

and Loral pursuant to which Telesat will provide TT&C services with respect to the Satellite and payments will be made to Loral, and the Cooperation Agreement, dated today, with ViaSat and Loral pursuant to which Telesat will make the orbital location at 115 degrees W.L. available to the Satellite and its replacement satellites.

WHEREAS, SS/L and ViaSat will allow Loral and its assignees reasonable access to the Satellite during the construction and testing thereof; and

WHEREAS, ViaSat and Loral desire to set forth in this Agreement the detailed terms and conditions of their agreement with regard to the arrangement for the Loral Payload and such other matters as set forth herein.

NOW, THEREFORE, in consideration of the premises and of the mutual representations, warranties, covenants and agreements hereinafter contained, the Parties hereby agree as follows:

ARTICLE 1

INTERPRETATION

- 1.1. Words and expressions used in this Agreement shall have the meanings set out in Schedule 1, unless the context requires otherwise.
- 1.2. The Schedules and Annexes to this Agreement shall form part of this Agreement.

ARTICLE 2

OWNERSHIP OF THE SATELLITE

2.1. Ownership, Title, and Operation. Loral shall exclusively own and operate the Canadian Satellite payload (defined in Section 2.1B of the Satellite Contract as the "Canadian Beams") which will have, as a minimum, a capacity equal to 15% of the satellite's Baseline Capacity (excluding Additional Beams) (subject to ViaSat's right to complete the design and configuration of the Satellite in accordance with Section 2.1(B) of the Satellite Contract and Section 2.2.1

below), and described in the Loral Payload Agreement (the "Loral Payload") and ViaSat shall exclusively own and operate the remainder of the Satellite payload (defined as the US Beams and Additional Beams, if any, in Section 2.1B of the Satellite Contract) (the "ViaSat Payload") and all the elements of the Satellite that are common to and/or shared by the ViaSat Payload and the Loral Payload (the "Common Elements"). Loral shall maintain a license to the Deliverable Items related to the Loral Payload in accordance with the Loral Payload Agreement. ViaSat shall maintain a license to the Deliverable Items related to the ViaSat Payload and Common Elements in accordance with the Satellite Contract; provided, that, ViaSat hereby grants to Loral a sublicense (and will provide a copy to Loral) with respect to the Deliverable Items related to the Common Elements in accordance with its rights to provide sublicenses under the Satellite Contract (provided that this sublicense shall not be considered a license to Loral to Purchaser Intellectual Property as such term is defined in the Satellite Contract). Subject to Sections 2.2, 2.3.3 and 2.3.4, ViaSat will make all operational decisions affecting the Common Elements in a fair and evenhanded manner, having due regard to its interests and the interests of Loral in the health and functionality of the Satellite and with prior notice to and consultation with Loral in the event of any such decision that is likely to affect the Loral Payload. Each Party may grant security interests in and otherwise encumber its portion of the Satellite and the Deliverable Items only in accordance with Section 13.1. As ViaSat is the owner of the Common Elements, in order to provide Loral with access to, and use of, the Loral Payload in the event ViaSat or any permitted successor fails to operate or contract for the operation of the Satellite bus, ViaSat grants to Loral with effect only upon such failure a non-exclusive license to access and operate the Satellite bus. Upon the written request of Lo

2.2. Satellite Oversight

2.2.1 ViaSat may make changes in the design of the Common Elements (subject to this Section 2.2.1 and Section 2.2.2 below) and either Party may at any time and from time to time make changes in the design of its payload, provided that in each case such changes do not negatively impact: (i) the scheduled launch date for the Satellite, (ii) the performance of the other Party's payload, (iii) the designed operational life of the other Party's payload or the designed orbital maneuver life of the Satellite below 15 years, (iv) the designed operation margins available for the Common Elements or the other Party's payload or (v) the designed reliability of the Common Elements or the other Party's payload in a material manner; provided, that Loral acknowledges and agrees that the Satellite configuration and design is not complete as of the date hereof and that ViaSat shall complete the design and configuration of the Satellite in accordance with Section 2.1(B) of the Satellite Contract in a fair and evenhanded manner, having due regard to its interests and the interests of Loral in the operation, capacity and functionality of the Satellite. For purposes of clarification Loral shall not have any authority under the prior sentence to make changes (directly or indirectly) to any of the Common Elements of the Satellite without ViaSat's consent, which consent shall not be unreasonably withheld or delayed (for purposes of clarification only, it shall be reasonable for ViaSat to withhold consent from Loral making a change to a Common Element of the Satellite if in ViaSat's good faith opinion such proposed change would adversely affect the ViaSat Payload or the Common Elements). All other design changes (including all changes to the Common Elements) shall be made only in accordance with Section 2.2.2. The Party making a design change to its payload shall be solely responsible for paying any increases to the Firm Fixed Price (defined in the Satellite Contract) that result from such change (including changes to the Common Element

design change made to such Party's payload). Similarly, the Party making a design change to its payload(s) shall receive the full benefit of any reductions in the Firm Fixed Price (as defined in the Satellite Contract) resulting from such change to its payload.

- 2.2.2 Each Party shall have final decision-making authority with respect to all matters regarding the construction and operation of the Satellite that solely and exclusively impact its respective payload, <u>provided</u> that the Parties shall nonetheless keep one another reasonably informed about such actual and contemplated decisions and provided further that ViaSat shall give the directions to SS/L under the Satellite Contract contemplated by this Agreement in its capacity as Purchaser under the Satellite Contract. The Parties shall consult with one another with respect to all matters regarding the launch of the Satellite and all matters regarding the construction and operation of the Satellite that involve the Common Elements or otherwise impact the payloads of both Parties, <u>provided</u> that in the event the Parties fail to reach agreement, ViaSat shall make the final decision in a fair and evenhanded manner, having due regard to its interests and the interests of Loral in the health and functionality of the Satellite and a timely launch of the Satellite. Neither Party shall be liable to the other as a result of delays in the construction of its respective payload(s) or the Common Elements that occur without breach of this Agreement.
- 2.2.3 ViaSat agrees to permit SS/L to provide Loral and its designees with the same level of access to the Work (as defined in the Satellite Contract) as provided to ViaSat under the Satellite Contract as consistent with the conditions and restrictions as set forth herein; provided, that ViaSat shall have the right to deny access to any Loral designee who is not a bona fide employee of Loral, Telesat or one of their subsidiaries to the extent ViaSat has a reasonable, good faith competitive concern with respect to such designee; provided, further that, prior to any

assumption by Telesat of Loral's rights and obligations hereunder and under the Loral Payload Agreement, ViaSat shall have the right to remove employees or designees of Telesat or any of its subsidiaries from meetings or portions of meetings in the event ViaSat has a reasonable, good faith competitive concern with respect to such employee or designee with respect to such meetings or portions of meetings. ViaSat will support Loral's operation of the Loral Payload by periodically providing (or providing Loral with access to) such technical information concerning transponder performance, interference considerations and operation, including, without limitation, expected End of Life, as Loral may reasonably require, including without limitation (i) real-time processed telemetry for the Loral Payload, (ii) trend and performance analysis for the Loral Payload, (iii) quarterly health and status reports for the Loral Payload and all Common Elements of the Satellite, (iv) information which is customarily contained in health reports provided to insurance underwriters and (v) reasonable responses to questions posed by insurance underwriters should Loral procure its own insurance on the Loral Payload. Each Party shall be responsible to the extent of its operational control over the Satellite for ensuring that its payload is operated in accordance with all applicable governmental (including licensing) requirements.

2.3. Responsibilities.

2.3.1 (a) RESERVED.

(b) In the event of a termination of the Satellite Contract by ViaSat for convenience, or by SS/L for ViaSat's default, (i) Loral shall have the right to terminate this Agreement, and (ii) ViaSat shall promptly reimburse Loral for all amounts Loral has theretofore paid hereunder and the Loral Payload Agreement (under the terms of such contract as of the date hereof and not incorporating any liabilities from future amendments), less any such amounts theretofore refunded to Loral, whereupon ViaSat shall be subrogated to any rights Loral may have to receive

such refunds in the future, and ViaSat shall be responsible for all termination costs arising under the Satellite Contract and the Loral Payload Agreement (under the terms of such contract as of the date hereof and not incorporating any liabilities from future amendments).

- (c) In the event ViaSat terminates the Satellite Contract for cause, (i) ViaSat shall have the right to terminate this Agreement, and (ii) Loral shall be entitled to only such reimbursement as may be provided under the Loral Payload Agreement and its pro rata share (based on payments it has theretofore made in respect thereof compared with the total payments made by ViaSat and Loral) of any amounts actually refunded by the launch services provider and insurance broker or carrier or any other third party in connection with any termination of Satellite launch and associated services and Initial Insurance.
- 2.3.2 ViaSat shall obtain launch and in-orbit operations insurance for the Satellite through the first anniversary of the launch date (the "Initial Insurance"), and in orbit operations insurance (at its discretion) for the Satellite after the Initial Insurance and through the End Of Life (together with the Initial Insurance, "Satellite Insurance"). Loral's interest in the Initial Insurance shall be in an amount equal to the Loral Price for the Loral Payload as defined in Article 3.1; provided that in the event ViaSat is not able to obtain insurance for the entire Loral Price for the Loral Payload then Loral shall be entitled to obtain supplemental insurance, and further provided that, in the event that there is not sufficient insurance capacity available to obtain such supplemental insurance on commercially reasonable terms and conditions, Loral shall be entitled to participate in any insurance obtained by ViaSat in such proportion so that the relative insurance for the Loral Payload shall be 15 percent of the amount of the Initial Insurance and the relative insurance for the ViaSat Payload shall be 85 percent of the amount of the Initial Insurance. ViaSat shall use commercially reasonable efforts to obtain the Initial Insurance for

the Satellite (including, launch and associated services and insurance premium) on commercially reasonable terms and conditions, and providing for coverage customary in the satellite insurance market, provided that the Initial Insurance shall be placed with insurance companies that are in ViaSat's good faith opinion financially sound and responsible at the time the relevant coverage is placed or renewed, and attach at launch and continue in effect until no sooner than one year after launch, shall be for partial loss, constructive total loss and total loss, including customary exclusions and shall, with respect to the Loral Payload, include Loral (or its assignee) and its designee (including any lender and/or collateral agent, with waivers of subrogation therefor if commercially reasonably available) as named insured and additional insured (and/or loss payee, if required), respectively, and provided further, that for periods following the first anniversary of the launch of the Satellite, in the event that ViaSat decides, in its sole discretion, to procure in-orbit insurance for the Satellite, ViaSat shall offer to obtain in-orbit insurance on the Loral Payload and timely disclose the available terms and conditions thereof to Loral, but shall obtain such insurance only if so requested by Loral, and be in an amount confirmed by Loral, in which event Loral will pay Loral's proportionate share of the cost thereof. ViaSat agrees to consult with Loral during the procurement process for the Satellite Insurance. In the event ViaSat decides not to procure in-orbit insurance on the Satellite or Loral declines ViaSat's offer to obtain in-orbit insurance on the Loral Payload, Loral may purchase independent in-orbit insurance for the Loral Payload; provided, that Loral will remain liable for its applicable portion of the Initial Insurance, and any portion of the in-orbit insurance purchased for the Loral Payload at Loral's request and provided further that Loral shall, in all events, but not prior to the earlier of (i) forty

to do so in a timely manner) from insurers for the Satellite Insurance it shall have procured and so long as ViaSat is complying with its obligations under this Section 2.3.2, be entitled to approach the insurance market and purchase, at its own sole cost and expense, supplemental insurance on the Loral Payload, in addition to any insurance procured (or to be procured) by ViaSat. In the event Loral purchases independent or supplemental insurance hereunder, ViaSat shall provide all commercially reasonable assistance and support to Loral with respect to such insurance procurement. In the event the Satellite suffers a total loss during the period starting at Intentional Ignition and ending at End of Life and Loral has elected to participate in such insurance, then ViaSat shall provide to Loral 15% of the net proceeds actually received from the Satellite Insurance, which shall represent ViaSat's sole obligation to Loral for such total loss. In the event the Satellite suffers a partial loss, ViaSat shall provide to Loral its proportionate share of the net insurance proceeds actually received from the Initial Insurance and any in-orbit insurance associated with such partial loss, if Loral shall have elected to participate in such insurance, calculated in accordance with the terms of the insurance as follows: (1) the Parties will share with respect to any loss that does not disproportionately affect either the Loral Payload or the ViaSat Payload in the ratio of 15 percent Loral/85 percent ViaSat; and (2) for losses that disproportionately affect one Party's payload, the proceeds shall be adjusted to reflect such disproportionate impact; this shall represent ViaSat's sole obligation to Loral for such partial loss, other than the obligations set forth in Sections 2.3.3 and 2.3.4. Loral shall cooperate in good faith with ViaSat in connection with the procurement by ViaSat of the Satellite Insurance, including providing any necessary technical or operational information regarding the Loral Payload. If there shall be a dis

not settle any such dispute without first obtaining Loral's prior written consent, which consent will not be unreasonably withheld or delayed. For avoidance of doubt, Loral shall be deemed to have elected to participate in the Initial Insurance.

- 2.3.3 The following procedure shall be adhered to by ViaSat in the event of a loss of primary power that impacts the payloads of both Parties:
 - (i) First, components of the Satellite will be powered down as necessary to protect the overall health of the Satellite;
- (ii) Second, bus components of the Satellite will be powered down to the extent possible without affecting either Party's payload or the overall health of the Satellite;
 - (iii) Third, the Parties will enter into good faith discussions to determine the required changes to the Satellite's operation or configuration; and
- (iv) In the event that the Parties are unable to mutually agree upon such changes within forty-eight (48) hours after the relevant loss of primary power, then the residual amount of primary power required to be shed will be allocated between the payloads of ViaSat and Loral based on the ratio of the nominal full-power operation of the Parties' respective payloads. Such allocation will initially be computed based on the final design specification for each Parties' respective payload(s), and shall be subject to adjustment when actual saturated power and efficiency data is determined during IOT (as defined in the Satellite Contract). Each Party will determine in its sole judgment the appropriate actions to be taken with respect to its payload(s) to shed the power allocated to it pursuant to this Subsection (iv).

- 2.3.4 The following procedure shall be adhered to by ViaSat in the event of a failure of a component part on the Satellite for which the redundant spare components are drawn from a pool that could serve either the Loral or ViaSat payloads:
 - (i) First, redundant or spare components of the Satellite will be used taking into consideration the protection of the overall health of the Satellite;
 - (ii) Second, subject to clause (iii) below, restoration of a failed component part on the Satellite by a redundant or spare component on the Satellite shall be done on a first failed, first restored basis on condition that the redundant or spare component is technically capable of restoring the failed component and is not already deployed to restore a failure, provided that, in the event of failures occurring simultaneously or within 24 hours of each other, the redundant or spare component shall first be used to restore the component on the Satellite which restores the highest amount of the Satellite's Designed Payload Capacity; and
 - (iii) Third, to the extent all redundant or spare components that are technically capable of restoring a failed component have already been deployed to restore previous failures, then the redundant and spare components that have been previously assigned shall be redeployed on terms as shall be agreed by ViaSat and Loral (using their commercially reasonable efforts within 120 days after the Effective Date); provided that in the event the Parties are unable to reach agreement on such terms then ViaSat shall redeploy such spare and redundant components in a fair and evenhanded manner, having due regard to the interests of ViaSat and of Loral in the health and functionality of the Satellite.

- 2.4. Ownership of the Common Elements. The Common Elements of the Satellite shall be owned exclusively by ViaSat.
- 2.5. <u>Cooperation</u>. Each Party shall cooperate with and assist the other Party from time to time as required to evidence such Party's title to its applicable portion of the Satellite, including executing a certificate of ownership or other documents as may be reasonably requested by such Party to evidence the title and rights to the Satellite set forth herein.
- 2.6. <u>Satellite Construction and Operational Decisions</u>. ViaSat shall also be entitled to, with the consent of Loral, which consent shall not be unreasonably withheld or delayed, decommission the Satellite without further liability to Loral in the event of the destruction, total loss or constructive total loss of the Satellite or when the on-board fuel on the Satellite reaches a level, or any circumstance develops, which in ViaSat's good faith opinion requires that the Satellite be de-orbited to safely remove it from geostationary orbit in accordance with applicable law or prudent and customary industry practice.
- 2.7. <u>Satellite Resources for Loral Payload</u>. Subject to the terms set forth in this Section 2, ViaSat shall provide for those satellite resources and functions used for the operation of both the ViaSat Payload and Loral Payload in a fair and evenhanded manner, having due regard to the interests of Loral and ViaSat, to permit the Loral Payload to be operated on the Satellite in accordance with the payload technical specifications for the Loral Payload as provided in the Satellite Contract from time to time. Loral will have the right to instruct the TT&C provider directly with respect to the command of the Loral Payload, provided such directions do not affect the Common Elements or the ViaSat Payload.
- 2.8. <u>Gateway Sharing</u>. If a ViaSat gateway is configured with excess capacity based on the Final Configuration (as defined in the Satellite Contract) ("Extra Gateway"), then ViaSat shall

provide written notice to Loral prior to the procurement of Extra Gateway and thereby Loral shall have the option for a period of 90 days to elect to participate in a sharing arrangement with ViaSat as described herein for the Extra Gateway ("Gateway Share"). In the event Loral elects to Gateway Share in accordance with the terms of this paragraph above, then Loral shall have the right to utilize the excess capacity in the Extra Gateway, upon (i) payment of its pro rata share (based on the ratio of the capacity to be allocated to Loral to the total capacity of the Extra Gateway) of the aggregate cost of constructing and testing the Extra Gateway and placing it in service (including reasonable labor costs and other reasonable expenses of ViaSat in provisioning such Extra Gateway) and (ii) its written binding commitment to pay its pro rata share, based on such ratio, of future capital costs of the Extra Gateway and operating and maintenance expenses of the Extra Gateway from such time forth. In the event Loral elects to Gateway Share in accordance with the terms above, then ViaSat will operate the Extra Gateway in a fair and evenhanded manner, having due regard for its and its customers' interests and the interests of Loral and its customers. Loral shall maintain reasonable audit rights with respect to the costs and expenses paid by Loral to ViaSat under this paragraph.

- 2.9. <u>Branding</u>. For the avoidance of doubt, Loral will not be required to market capacity on the Loral Payload under the ViaSat name and shall have the right to market such capacity under a trade name of its own choosing, which need not mention or include the names "ViaSat" or "Leapfrog".
- 2.10. <u>Restrictions on Competition</u>. During the term of this Agreement, neither ViaSat nor any of its affiliates shall market satellite services provided by the Satellite or any Replacement Satellite ("<u>Competitive Services</u>") or intentionally use the Satellite or any Replacement Satellite to provide Competitive Services to user terminals located in Canada; and none of Loral, Telesat

or any of their respective affiliates shall market Competitive Services or intentionally use the Satellite or any Replacement Satellite to provide Competitive Services to user terminals located outside of Canada. The parties acknowledge and agree that the agreements and covenants contained in this Agreement are reasonable and valid in geographical and temporal scope and in all other respects, and essential to protect the value each party's interest in the Satellite and any Replacement Satellite. Nothing in this Section 2.10 shall constrain either Party from using any other satellite at any other orbital location to provide satellite services that are competitive with the Competitive Services.

ARTICLE 3

PAYMENT PLAN

- 3.1. <u>Loral Price for Loral Payload</u>. Based on a total estimated cost of \$400 million for manufacturing, launching, procuring Initial Insurance, and operating the Satellite for 15 years, Loral would pay an aggregate price of \$60 million, as described in more detail below.
- 3.1.1 Loral shall pay to SS/L \$36.9 million for the Loral Payload portion of the Satellite (such payment, together with Loral's share of expenses payable pursuant to Section 3.1.2, the "Loral Price") in accordance with the payment plan set forth in the Loral Payload Agreement.
- 3.1.2 The Loral Price is based on the following estimated costs: (i) \$246 million for the Satellite ("Satellite Costs"); (ii) \$41 million for the Initial Insurance; (iii) \$97 million for Satellite launch and associated services ("Launch Costs") and (iv) \$14.7 million for Satellite telemetry, tracking and command services for 15 years ("TT&C Costs"). Loral shall pay 15% of the actual Launch Costs directly to ViaSat and 15% of the actual costs for the Initial Insurance directly to ViaSat. ViaSat will be responsible for 85% of the TT&C Costs provided under the TT&C Agreement, and Loral will be responsible for the remainder.

- 3.1.3 There shall be no adjustment to the Loral Price hereunder for a price adjustment under the Satellite Contract due to any capacity adjustment for the ViaSat Payload pursuant to Articles 4.6 and 22.1 of the Satellite Contract. Cost changes under the Satellite Contract resulting from changes requested by a Party or a Party's actions or omissions shall be paid as contemplated by Section 2.2.1. In the event ViaSat makes a change to a Common Element which in its reasonable opinion is necessary or advisable and such change benefits the Loral Payload proportionately, Loral will be responsible for fifteen percent (15%) of any increase in the Satellite Cost owed to SS/L associated with such change to the Common Element(s). Such adjustment will be reflected in an amendment to this Agreement, and the Payment Plan under the Loral Payload Agreement will be correspondingly amended.
- 3.1.4 The Loral Price above does not include in-orbit insurance costs for the Satellite during the period after the Initial Insurance. In the event ViaSat decides to procure in-orbit insurance for the Satellite following expiration of the Initial Insurance and Loral elects to cause the Loral Payload to be included in that coverage, Loral shall pay to ViaSat 15% of the in-orbit insurance costs for the Satellite for the period ending after the Initial Insurance and continuing with any subsequent renewals by ViaSat of in-orbit insurance for the Satellite through the End Of Life.
- 3.1.5 Loral shall also pay 15% of any other actual, reasonable and necessary costs, liabilities or other payments made by ViaSat associated with constructing, transporting, launching, operating or maintaining the Satellite to the extent not included in the estimated costs described above, including in orbit insurance renewals costs (but only if Loral elects to include the Loral Payload in such insurance coverage) ("Other Costs"), provided that such Other Costs benefit the Loral Payload proportionately with the ViaSat Payload. ViaSat agrees to advise and

consult with Loral prior to incurring any Other Costs in excess of \$1,000,000 in the aggregate, to provide Loral with a copy of an invoice or other documentation related to the Other Costs prior to or at the time ViaSat invoices Loral for such costs. For the avoidance of doubt, examples of Other Costs include, without limitation, storage costs for Satellite in the event of a launch delay and attorney's fees required to collect on Satellite Insurance, etc. Examples of expenses that will not constitute Other Costs include attorney's fees in connection with the negotiation of this transaction and advice related hereto, general corporate overhead and executive management of the Satellite, on site supervision of the manufacture of the Satellite and Satellite monitoring activities at and following launch.

3.1.6 The Parties will invoice one another for any amounts under this Section 3. ViaSat shall submit Loral such invoice no earlier than 30 days prior to ViaSat incurring such cost. Loral shall pay ViaSat in full all such amounts within 30 days of receipt of such invoice. Late payments by one Party to the other Party shall be subject to the provisions of Section 20.1 below.

ARTICLE 4

REVENUE FROM BEAMS

4.1. Revenues from Beams. Except as set forth in Section 4.2 below, Loral shall be entitled to all revenues generated from the Loral Payload and ViaSat shall be entitled to all revenues generated from the ViaSat Payload.

4.2. Revenue Share Transactions.

4.2.1 In the event Loral has not on or before October 31, 2009 assigned this Agreement to Telesat or a third party or otherwise sold or leased the Loral Payload to Telesat or a third party, then ViaSat agrees, if so requested by Loral, to contribute to Loral for each of its service gateway facilities the requisite next-generation hub equipment (SMTS equipment) in

consideration for the list price thereof (the "SMTS Cost"), and thereafter (until the End of Life of the Satellite) ViaSat shall be entitled to a percentage of the revenues earned by Loral (or any affiliate of Loral that owns the Loral Payload) from the sale or lease of capacity on the Loral Payload equal to a fraction, the numerator of which is the SMTS Cost, and the denominator of which is the sum of (i) the SMTS Cost, (ii) the Loral Price and (iii) the sum of all other capital expenditures made by Loral and its affiliates in connection with the business in question. In that event, the parties will enter into appropriate, separate agreements to implement the foregoing. ViaSat shall maintain reasonable audit rights with respect to the revenue share transaction described above.

4.3. Exclusivity Provisions. In consideration of ViaSat's obligations set forth in Section 4.4 below, ViaSat shall through the End of Life of the Satellite remain the sole and exclusive supplier of end-users broadband terminals (indoor modems and outdoor transceivers, antenna units and related software) and hub (SMTS) equipment for any two way service operated by a Canadian Operation (defined below) over the Loral Payload; provided that the terms of this paragraph shall continue to apply so long as (i) ViaSat remains in the business of the design, manufacture and distribution of equipment suitable for such applications for Eutelsat or other significant customers in addition to Loral, (ii) such terminals and equipment are of merchantable quality, (iii) and such terminals and equipment perform in accordance with final specifications provided to the Key Distributor (defined below), which shall be generally consistent with the specifications described on Annex 4.3 attached hereto and ViaSat does not materially breach the contractual obligations between ViaSat and Loral with respect to such performance specifications (collectively, the "ViaSat Obligations"). In the event ViaSat is in material breach

of the ViaSat Obligations, the Canadian Operator shall provide notice to ViaSat of such breach and provide ViaSat no less than 60 days to cure such breach (and an opportunity to arbitrate the matter hereunder) prior to the ViaSat Obligations terminating.

4.4. <u>Favored Pricing</u>. ViaSat agrees to provide any Person operating (or selling) a two way broadband service over the Loral Payload ("<u>Canadian Operator</u>") with the same price (based on the then-current total price for terminals (indoor and outdoor units together)) being delivered to the Key Distributor (defined below) for ViaSat's next generation Surfbeam equipment ("Equipment") that is provided to the Key Distributor; <u>provided</u>, that the Canadian Operator must purchase the Equipment on the same terms and conditions as the Key Distributor, including warranty terms, delivery lead times, annual maintenance fees, and support levels (but excluding terms related to the minimum number of aggregate committed terminals). ViaSat agrees to provide commercially reasonable pricing for any additional terms or items requested by the Canadian Operator. For purposes of this Agreement, "<u>Key Distributor</u>" means the distributor (and the pricing and terms of such distributor) selected by the Canadian Operator among the distributors purchasing Equipment at the time the Canadian Operator first enters into an agreement to purchase such Equipment from ViaSat. In the event any distributor is purchasing and taking delivery of user terminals at a lower total terminal price (indoor and outdoor unit together) than the price provided to the Canadian Operator (the "<u>Improved Distributor</u>'), ViaSat shall inform the Canadian Operator of such lower price (and the terms under which the Improved Distributor is purchasing terminals). The Canadian Operator shall then have the right for a reasonable period of time (no longer than 60 days) to elect to substitute the pricing and terms of the Key Distributor, then the Canadian

Operator shall again maintain the rights to substitute terminal pricing and terms described above. The terms of this paragraph shall not apply to limited quantities of terminals provided at discounted (or no cost) pricing that may be sold or loaned to a potential distributor for testing or demonstration purposes. In the event ViaSat enters into an agreement with a Key Distributor or Improved Distributor that provides that such Key Distributor or Improved Distributor is to receive the lowest pricing of any distributor, then the Canadian Operator agrees that ViaSat can charge a per unit price \$1.00 higher to the Canadian Operator than such distributor.

ARTICLE 5

THE SATELLITE

- 5.1. <u>Satellite Specifications</u>. Subject to Section 2.2, the Satellite shall have the design and other specifications as are currently set forth in the Satellite Contract. Loral acknowledges the final specifications of the Satellite will be determined in accordance with Section 2.1B of the Satellite Contract.
- 5.2. <u>Modifications of Loral Payload</u>. Subject to ViaSat's right to designate the Final Configuration (defined in the Satellite Contract), the location of the beams associated with the Loral Payload on the Satellite may be modified by SS/L as requested by Loral prior to January 25, 2008¹ in accordance with the Satellite Contract. Any costs associated with such modification, if any, shall be borne solely by Loral.
- 5.3. <u>ViaSat Actions</u>. Subject to Article 2 hereof, Loral acknowledges that it has reviewed the Satellite Contract (as in effect as of the date hereof) and agrees to all the terms and conditions contained therein and that it shall be bound by all actions taken by ViaSat with respect thereto, and with regard to the selection of the launch services provider (or with regard to the launch

services contract), selection of the broker for insurance for the Satellite (or with regard to the insurance broker contract), selection of the insurance for the Satellite (or with regard to the insurance contracts) and other service providers (and other service provider contracts) associated with the design, construction, testing, launch and operation of the Satellite. ViaSat will, nonetheless, consult with Loral concerning such actions and will make its decisions in a fair and evenhanded manner, having due regard to its interests and the interests of Loral in the health and functionality of the Satellite and the timely launch of the Satellite.

5.4. Limited Representations. Loral acknowledges that:

VIASAT HAS NOT MADE, NOR DOES IT MAKE, ANY REPRESENTATION OR WARRANTY, WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION, ANY WARRANTY OF DESIGN, OPERATION, CONDITION, QUALITY, SUITABILITY OR MERCHANTABILITY OR FITNESS FOR USE OR FOR A PARTICULAR PURPOSE, ABSENCE OF LATENT OR OTHER DEFECTS, WHETHER OR NOT DISCOVERABLE, WITH REGARD TO THE SATELLITE OR ANY OTHER DELIVERABLE ITEM UNDER THE SATELLITE CONTRACT, AND VIASAT HAS NOT MADE ANY WARRANTY WITH RESPECT TO THE PERFORMANCE OF ANY LAUNCH VEHICLE.

5.5. No Fault. Loral agrees to be bound by the no-fault, no-subrogation inter-party waiver of liability and related indemnity provisions provided in the Launch Services Agreement and to cause its contractors and subcontractors at any tier (including suppliers of any kind) that are involved in any performance of this Agreement and any other person who through Loral has an interest in the Satellite or any transponder thereon, as required by the Launch Services Agreement, to accede to such waiver. Loral shall execute and deliver any instrument that may be

reasonably required by the Launch Agency to evidence its agreement to be bound by such waiver. In no event shall such no-fault, no-subrogation inter-party waiver and related indemnity provisions have any effect on the rights, obligations, and liabilities of and between Loral and ViaSat under this Agreement or under the other Transaction Documents, and Loral shall cause Telesat or any other purchaser of the Loral Payload to be bound by the provisions of this Section.

5.6. Indemnity. (a) Loral shall indemnify, defend, and hold harmless ViaSat, its officers, directors, shareholders, employees, subcontractors and agents ("ViaSat Indemnitees") from and against all liability, damages, losses, claims, demands, actions, judgments and costs, including legal fees arising from (i) damage to any property, and for injury to or death of any person, including employees or agents of ViaSat and any third parties, resulting from any willful misconduct of Loral relating to this Agreement; (ii) any claim of interference with or disruption of communications caused by any person using, accessing, receiving transmissions or communicating with the Loral Payload; (iii) claims for libel, defamation, passing off, slander, copyright infringement, other intellectual property right infringement or other claims relating to the content of programming, video, data, voice or other material transmitted on, to or from the Loral Payload; (iv) any claims relating to any warranty or representation made by Loral to a third party; (v) to the extent permitted by law, any fines, penalties and forfeitures arising out of or relating to any proceeding against or involving ViaSat Indemnitees arising out of any alleged violation of any law by Loral, its affiliates, agents or customers; (vi) any claims alleging infringement of any letters, patent, copyright, trademark or other intellectual property rights caused by the Satellite or the use or operation thereof.

(b) ViaSat shall indemnify, defend, and hold harmless Loral, its officers, directors, shareholders, employees, subcontractors and agents ("Loral Indemnitees") from and against all

liability, damages, losses, claims, demands, actions, judgments and costs, including legal fees arising from (i) damage to any property, and for injury to or death of any person, including employees or agents of Loral and any third parties, resulting from any willful misconduct of ViaSat relating to this Agreement; (ii) any claim of interference with or disruption of communications caused by any person using, accessing, receiving transmissions or communicating with the ViaSat Payload; (iii) claims for libel, defamation, passing off, slander, copyright infringement, other intellectual property right infringement or other claims relating to the content of programming, video, data, voice or other material transmitted on, to or from the ViaSat Payload; (iv) any claims relating to any warranty or representation made by ViaSat to a third party; (v) to the extent permitted by law, any fines, penalties and forfeitures arising out of or relating to any proceeding against or involving Loral Indemnitees arising out of any alleged violation of any law by ViaSat, its affiliates, agents or customers; (vi) any claims alleging infringement of any letters, patent, copyright, trademark or other intellectual property rights caused by the Satellite or the use or operation thereof.

- (c) NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY OR ANY OTHER PERSON OR ENTITY FOR INDIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL, PUNITIVE, OR EXEMPLARY DAMAGES ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT.
- 5.7. Action to Protect Satellite. ViaSat shall have sole and exclusive control of operation of the Satellite, subject to consultation with Loral, and shall exercise its control in a fair and evenhanded manner, having due regard to its interests and the interests of Loral in the health and functionality of the Satellite. Notwithstanding anything to the contrary herein, if circumstances occur which in ViaSat's reasonable judgment pose a threat to the stable operation or health of the

Satellite, including without limitation any failure of the Satellite to meet its performance specifications (including but not limited to permanent or recurrent shortage of electric power or a material breach in the operating or Satellite payload access procedures by Loral or its designees or customers), ViaSat shall have the right to take any action it reasonably believes necessary to protect the Satellite, including discontinuance or suspension of operation of the Satellite, any transponders (including the Loral Payload) without any liability to Loral, its designees or customers so long as it provides notice to Loral in accordance with the terms of this paragraph below and proceeds in a fair and evenhanded manner, without discrimination against the Loral Payload or the customers it serves. ViaSat shall provide Loral as much notice as practical under the circumstances of any such discontinuance or suspension of operation of the Satellite payload or the Loral Payload (or portion thereof). Under no circumstances shall ViaSat be responsible for any loss or damage to Loral, its designees or customers for any action taken by ViaSat in conformity with this Section 5.7, except to the extent that ViaSat's actions are not taken in good faith and result in a disproportionate damage to the Loral Payload or the customers it serves. Notwithstanding anything to the contrary herein, ViaSat shall operate the Satellite in accordance with applicable regulations and subject to applicable regulatory authorities.

ARTICLE 6

REPRESENTATIONS AND WARRANTIES OF THE PARTIES

Each of the Parties, as of the date hereof, hereby represents and warrants to the other, as follows:

6.1. <u>Organization and Standing</u>. It is a corporation duly organized, validly existing and in good standing under the laws of the place of its incorporation and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its businesses as now being conducted, except where the failure to be so organized, existing and in good standing or to

have such power and authority would neither have a material adverse effect on its financial condition, business or results of operation nor materially impair or delay its ability to consummate the transactions contemplated hereby (a "Material Adverse Effect").

- 6.2. <u>Authority Relative to this Agreement</u>. It has all corporate power and authority to execute and deliver this Agreement and the other Transaction Documents, and to perform all of its obligations hereunder and thereunder. Its execution and delivery of this Agreement and the other Transaction Documents to which it is a party and its performance of its obligations hereunder and thereunder have been duly authorized by all necessary and proper corporate action. This Agreement and the other Transaction Documents to which it is a party have been duly executed and delivered by it and constitute the legal, valid and binding obligations, enforceable against it in accordance with their terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws relating to or affecting creditors' rights and subject, as to enforceability, to general principles of equity (regardless whether enforcement is sought in a proceeding in equity or at law).
- 6.3. Noncontravention. Its execution and delivery of this Agreement and the other Transaction Documents to which it is a party, its performance of its obligations to be performed hereunder and thereunder and the consummation of the transactions contemplated hereby and thereby will not (i) contravene or conflict with its charter, by-laws or other organizational documents; (ii) contravene or conflict with or constitute a violation of any provision of any laws or license to which it or any of its properties or assets is subject; or (iii) conflict with, result in a breach of, constitute a default under, result in the acceleration of, cause it to make an offer to purchase under, create in any Person the right to accelerate, terminate, modify or cancel, require any notice or give rise to a loss of any benefit under, any contract, lease, lien or other

arrangement to which it is a party or by which is bound or to which any of its properties or assets is subject or result in the creation or imposition of any liens on any of its assets, other than any loss of benefit, lien or any other such event which would not have a Material Adverse Effect.

- 6.4. <u>Governmental Proceeding; Litigation</u>. There is not in effect any judgment, order, writ, decree, stipulation or injunction by or with any governmental entity to which it or any of its Affiliates is party or by which it or any of its Affiliates or any properties or assets of any of the foregoing is bound, and which relates to or affects this Agreement or the transactions contemplated hereby, and neither it nor any of its Affiliates is party to, engaged in or, to its Knowledge, threatened with any Action which relates to or affects this Agreement or the transactions contemplated hereby, and, to its Knowledge, no event has occurred and no condition exists which could reasonably be expected to result in any such Action. Neither it nor any of its Affiliates is in default under or with respect to any judgment, ruling, order, writ, decree, stipulation or injunction of the type described in this Paragraph.
- 6.5. <u>Lawful Purposes</u>. Each of ViaSat and Loral agrees, and each will require any lessee or user of the Satellite, including the Loral Payload, to agree to restrict its use of the Satellite (including the Loral Payload) transmission capacity only for any lawful purpose and agrees to comply in all material respects with all applicable laws and government regulations. Each party shall be solely responsible for obtaining all governmental authorizations, and complying with all applicable governmental regulations, relating to the construction and operation of its respective payload. Each party acknowledges that governmental licensing for its payload will need to be coordinated with the governmental licensing of the other Party's payload, including potentially requiring an exchange of letters between the licensing administrations, and each Party hereby agrees to consult and cooperate with the other Party in all reasonable respects in such regard.

ARTICLE 7

REPLACEMENT RIGHTS

7.1. End of Life. In the period beginning fifty (50) months and ending forty (40) months prior the then-expected End of Life of the Satellite, Loral shall send written notice in good faith (the date of such notice, the "Replacement Satellite Notice Date") to ViaSat stating whether or not it desires to enter into a beam sharing agreement (consistent with the terms hereof) for any ViaSat replacement satellite ("Replacement Satellite") to be placed in the 115 WL or a nearby (within one degree) orbital location (any such location, the "Subject Slot"). In the event that Loral provides such notice to ViaSat by the Replacement Satellite Notice Date and ViaSat contracts for a Replacement Satellite in the Subject Slot, ViaSat shall, so long as it does not (after ViaSat using commercially reasonable efforts to design a payload similar to the Loral Payload onto such Replacement Satellite) substantively adversely affect the Replacement Satellite (from a technical, reliability, regulatory and risk perspective), offer to Loral the opportunity to enter into an arrangement consistent with the terms herein with respect to the Replacement Satellite, providing Loral with substantially similar rights and benefits to those provided by the arrangement contemplated hereby, including, without limitation, provisions substantially similar to this Section 7.1 with respect to Loral's right to an arrangement consistent with the terms herein with respect to replacement satellites. In the event ViaSat contracts for the Replacement Satellite (or replaces a replacement satellite) and ViaSat was unable to place a Loral payload on such Replacement Satellite because it substantively adversely affected the Replacement. In the event ViaSat

contracts for the Replacement Satellite (or replaces a replacement satellite) and Loral has not expressed its desire to enter into an arrangement with ViaSat for the Replacement Satellite by the Replacement Satellite Notice Date, then Loral shall not be entitled to place a satellite in the Subject Slot for the life of such satellite and Loral will (at ViaSat's expense) cooperate as ViaSat shall reasonably request, in all necessary or appropriate regulatory actions and proceedings to effectuate the foregoing, including, without limitation, granting ViaSat a sublicense for the life of such replacement satellite to Loral's rights to the Subject Slot, as applicable. ViaSat shall have the initial right to replace the Satellite with the Replacement Satellite (and any additional replacement satellites) at the Subject Slot but in the event ViaSat elects not contract for the Replacement Satellite (or any other replacement satellite) at its End of Life, then Loral will have the right thereafter to place a satellite of its own² in the Subject Slot, and ViaSat will (at Loral's expense) cooperate as Loral shall reasonably request, in all necessary or appropriate regulatory actions and proceedings to effectuate the foregoing, including, without limitation, granting Loral a sublicense for the life of such replacement satellite to ViaSat's rights to the Subject Slot, as applicable. For the avoidance of doubt, the Parties agree that the Parties' rights with respect to replacement satellites in the Subject Slot are not limited to the first replacement satellite, but shall apply to all replacement satellites, and that similarly, Loral's right to place its own satellite in the Subject Slot shall apply not only to ViaSat's election not to contract for a Replacement Satellite, but to all follow-on replacement satellites that are thereafter placed in the Subject Slot.

7.2. <u>Replacement Right</u>. If, after reasonable consultation, ViaSat and Loral are unable to reach agreement concerning the terms and conditions of their joint financial participation in a satellite to replace the Satellite before the end of its useful life, ViaSat shall have the right (but

not the obligation) to purchase the Loral Payload from Loral (at no cost) and replace the Loral Payload with an alternative payload on an alternate satellite ("Alternate Satellite") with equal or better capacity and coverage as the Loral Payload ("Substitute Payload") with reasonably minimal interruption in service or changeover costs to Loral or its customers. ViaSat shall provide the Substitute Payload for use by Loral through the projected End of Life of the Satellite ("Original Satellite Life") at no additional cost to Loral on terms and conditions that preserve for Loral benefits equivalent to those contemplated by this Agreement, including without limitation its rights under Section 7.1. ViaSat shall provide Loral with access to the Substitute Payload for the life of the Alternate Satellite beginning after the Original Satellite Life (such period referred to herein as the "Follow On Period") on terms and pricing consistent with Article 3 above, with appropriate reduction in price to reflect the shorter duration of the Follow On Period. Loral shall (at ViaSat's expense) cooperate, as ViaSat shall reasonably request, in all necessary or appropriate regulatory actions and proceedings to effectuate the foregoing.

ARTICLE 8 REASONABLE CHANGES

- 8.1. <u>ViaSat Reasonable Changes</u>. ViaSat agrees to negotiate in good faith and use reasonable efforts to amend the terms of this Agreement as may be reasonably requested by Loral to accommodate Loral in its assignment of the terms of this Agreement and/or the sale of the Loral Payload.
- 8.2. <u>Loral Reasonable Changes</u>. Loral agrees to negotiate in good faith and use reasonable efforts to amend the terms of this Agreement as may be reasonably requested by ViaSat to accommodate ViaSat in obtaining investments for the business operated with the ViaSat Payload.

ARTICLE 9

RESERVED

ARTICLE 10

RESERVED

ARTICLE 11 RESERVED

ARTICLE 12

EFFECTIVE DATE OF THIS AGREEMENT

This Agreement shall become effective on the date (the "Effective Date") set forth on the signature page hereof.

ARTICLE 13

PLEDGES

- 13.1. <u>Rights to Pledge</u>. Each Party shall have the right to pledge, mortgage, charge, grant any security interest in, or otherwise encumber all or part of its interest in its respective payload (and, in the case of ViaSat, the Common Elements), Deliverables, work-in-process under the Satellite Contract and such Party's rights under this Agreement and the agreements contemplated hereby, <u>provided that</u>:
 - (a) such Party shall remain liable for all obligations hereunder;
- (b) the encumbrance shall be subject to any necessary approvals or restrictions of any relevant governmental authority or telecommunications administration; and

- (c) any financing by Loral of the Loral Payload shall be structured so as not to interfere with or preclude any financing by ViaSat of the other portions of the Satellite and shall include intercreditor arrangements on terms reasonably satisfactory to ViaSat and its creditors that shall facilitate such financing by ViaSat, provided that, subject to the foregoing, ViaSat shall work cooperatively with Loral and Telesat and use reasonable efforts to accommodate their current and planned financing requirements with respect to the Loral Payload; and
- (d) satisfactory arrangements as agreed between the Parties (including a Party's lenders) shall have been made to recognize and protect the rights of the other Party and its lenders under this Agreement and the Transaction Documents.

Each of the Parties acknowledges and agrees that the other Party, in the course of the negotiation of this Agreement, used all commercially reasonable efforts to avoid any restriction on its ability to pledge its assets for any purpose, and that those negotiations resulted in the foregoing provisions.

ARTICLE 14 RESERVED

ARTICLE 15

RELATIONSHIP OF THE PARTIES

15.1. <u>No Partnership</u>. The rights and obligations of the Parties hereunder shall be individual, not joint or collective. It is not the intention of the Parties to create, nor shall this be deemed or construed to create a partnership, joint venture, association or trust, or as authorizing any Party to act as an agent, servant or employee for any other party for any purpose except as explicitly set forth herein.

ARTICLE 16

TAXES

ViaSat shall be responsible for the payment of any and all sales, use, gross receipts, excise and other taxes (collectively "<u>Taxes</u>") assessed solely and exclusively on the construction, use and operation or addition of value to its payload on the Satellite, and ViaSat shall indemnify Loral from any such Taxes. Loral shall be responsible for the payment of any and all Taxes assessed solely and exclusively on the construction, use and operation or addition of value to the payload on the Satellite, and Loral shall indemnify ViaSat from any such Taxes. The Parties shall share responsibility (ViaSat 85% and Loral 15%) for any and all Taxes assessed on the construction, use and operation or addition of value to any and all Common Elements, and each Party shall indemnify the other Party from its respective portion of such Taxes.

ARTICLE 17

ENTIRE AGREEMENT, TRANSACTION DOCUMENTS

- 17.1. <u>Transaction Documents</u>. The Parties agree and acknowledge that the following agreements are being entered into contemporaneously herewith (such agreements, the "<u>Transaction Documents</u>"):
 - (a) the Satellite Contract
 - (b) the Loral Payload Agreement
 - (c) the TT&C Agreement
 - (d) the Option Agreement
 - (e) the Cooperation Agreement
- 17.2. Entire Agreement. This Agreement and the other Transaction Documents together constitute the entire agreement and understanding between the Parties in connection with the

transactions hereby contemplated. The Transaction Documents supersede all previous agreements, arrangements and understandings between the Parties with regard to such transaction which shall cease to have any further force or effect. No Party is entering into any of the Transaction Documents or any of the arrangements hereby contemplated in reliance upon any representation, warranty or undertaking which is not expressly set out or referred to in any of the Transaction Documents.

ARTICLE 18

TERMINATION

- 18.1. <u>Termination</u>. Unless terminated sooner pursuant to this Section 18, this Agreement shall continue in full force and effect until the end of life of the last replacement satellite placed in the Subject Slot pursuant to Section 7.1. This Agreement may be terminated as follows:
 - (a) By mutual written agreement of the Parties;
- (b) By either Party by written notice to the other and in accordance with Article 20 hereof in the event of a default by the other Party provided such default meets the requirements stated in said Article 20; and
- (c) By either Party by written notice to the other if the other Party becomes insolvent, enters into a general suspension of payments, bankruptcy, makes a general assignment for the benefit of creditors, admits in writing its inability to pay debts as they mature, suffers or permits the appointment of a receiver for substantially all of its business or assets, or avails itself of or becomes subject to any other judicial or administrative proceeding that relates to insolvency or protection of creditors' rights (each an "Insolvency Event"). In the event Loral has an Insolvency Event, then ViaSat shall have a right of first refusal or option to purchase the Loral

Payload on the same terms and conditions as may be offered to any third party in a sale of such Party's assets, including any auction or liquidation sale.

18.2. <u>Effect of Termination</u>. Termination of this Agreement shall not affect: (i) either Party's obligations under any other Transaction Document; or (ii) either Party's obligations pursuant to Articles 2, 5.6, 5.7, 13, 15, 16 and 22-25 of this Agreement, which obligations shall expressly continue for so long as both Parties possess an economic interest in the Satellite, and Alternate Satellite, or a replacement satellite contemplated by Section 7.1, as the case may be.

ARTICLE 19

FORCE MAJEURE

19.1. <u>Force Majeure</u>. Neither Party shall be liable for nonperformance or delays in performance when caused by acts or events which are beyond the reasonable control of the delayed Party, including but not limited to the following: acts of God, acts of the public enemy, acts government in its sovereign (and not contractual) capacity, strikes or other labor disturbances, hurricanes, earthquakes, fires, floods, epidemics, embargoes, war, and riots ("<u>Force Majeure</u>"). In the event of any such delay, the date of delivery or of performance of the obligation affected by the force majeure event shall be extended for a period equal to the effect of time lost by reason of the delay.

19.2. <u>Notice</u>. A Party claiming delay in delivery or performance due to an event of Force Majeure as set forth herein shall as soon as practicable send written notice thereof and a statement of particulars to the other Party. Upon the cessation of the Force Majeure event, the Party claiming delay in delivery or performance of its obligations hereunder shall promptly notify the other Party of such cessation.

- 19.3. <u>Affected Party</u>. The Party affected shall take appropriate measures to minimize or remove the effects of the event of Force Majeure and, within the shortest time possible, shall attempt to resume performance of the obligations affected by the event of force majeure.
- 19.4. <u>Mitigation Duty</u>. Each Party shall use its reasonable efforts to minimize the losses and damages caused and/or to be caused to the other Party by an event of Force Majeure. Both Parties shall consult as soon as possible to find an appropriate solution.

ARTICLE 20

DEFAULT

- 20.1. Monetary Default. Should either ViaSat or Loral fail to make timely payment of any amount required to be paid to the other Party hereunder in accordance with the provisions defined herein, and such failure to pay shall have continued for a period of 60 days, the Party in breach shall pay interest to the other Party at the 30-day LIBOR rate plus three percent (3%) per annum in respect of the amounts in arrears. Such interest shall be calculated on a daily basis from the date payment was due until the date payment is received by the non-breaching Party. Should the Party in breach continue to fail to make such payment for a period of nine (9) months in the aggregate, then in addition to the interest amount due from the breaching Party, the other Party shall have the right to terminate this Agreement and to claim damages from the Party in breach in accordance with the provisions of Paragraph 20.3 hereof.
- 20.2. <u>Non-Monetary Default</u>. Should either ViaSat or Loral fail to cure a material breach of any provisions of this Agreement (other than provisions regarding payment of monies, which are provided for in Paragraph 20.1 above) within forty-five (45) days after receipt of written notice from the other Party outlining such breach, then the other, non-breaching Party shall have the

right to terminate this Agreement and to claim damages from the Party in breach in accordance with the provisions of Paragraph 20.3 hereof.

20.3. <u>Indemnification</u>. If the other Party suffers any cost, liability or loss as a direct result of a material breach of this Agreement by any Party, and such breach shall not have been cured by such Party within forty-five (45) days from receipt of notice of breach, the Party in breach shall indemnify and hold the non-breaching Party harmless in respect of any such cost, liability or loss; provided always, however, that in no event shall a Party be liable under any theory of tort, contract, strict liability, or other legal or equitable theory, for any indirect, special, incidental, or consequential loss or damage (including without limitation, loss of profit or business opportunity).

ARTICLE 21

CONFIDENTIALITY

- 21.1. Press Release. No press release, announcement or disclosure to a third party concerning the transactions contemplated hereby will be made by any Party hereto without the prior consent of the other Party hereto, except as such release, announcement or disclosure may be: (a) required by law or the rules of any applicable securities exchange; (b) necessary to be made to a Party's lenders for financing purposes provided that such lenders agree to maintain the confidentiality of any such disclosed information on customary and reasonable terms; or (c) is or becomes publicly known, otherwise than as a consequence of a breach of this Agreement.
- 21.2. <u>NDA</u>. The Parties have executed a Confidentiality Agreement dated March 8, 2006 (as amended on April 27, 2007) covering disclosure of information that may be made in connection with the Parties' performance under this Agreement and the Transaction Documents. The rights

and obligations thereunder shall apply to all such proprietary information disclosed in the implementation or performance of this Agreement.

ARTICLE 22

ASSIGNMENT

22.1. General Assignment. This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their respective successors and permitted assigns. Loral may assign its rights and obligations hereunder in part or whole without the consent of the other Party so long as the relevant assignee entity expressly assumes all of the assigning Party's rights and obligations hereunder; provided, that such assignee shall also agree to be bound and act in accordance with the terms of the Cooperation Agreement; provided, further that such assignee shall not be entitled to the benefits and rights set forth in Sections 4.4 or 7 without the prior written consent of ViaSat (such consent not to be unreasonably withheld). ViaSat hereby provides prior written consent for Loral to assign this Agreement to Telesat as long as Telesat expressly assumes all of Loral's rights and obligations hereunder. Neither Party shall provide any technical information related to the Satellite to any third party (including Affiliates) unless and until such third party executes a confidentiality agreement with the other Party with such terms as such other Party reasonably determines are appropriate for such third party, and the provision of such technical information complies with ITAR. Neither Party shall transfer its interest in the Satellite to any party who does not assume such Party's obligations hereunder. No assignment of this Agreement shall relieve the assigning Party of its financial obligations hereunder, except that an assignment of this Agreement by Loral to Telesat or any other entity with a credit rating on senior debt of B2 or better from Moody's (or the equivalent) shall so relieve Loral, and an assignment of this Agreement by ViaSat to a wholly-owned subsidiary of

ViaSat or ViaSat majority-owned or controlled joint venture formed to own and operate the Satellite and conduct its business or any other entity with a credit rating on senior debt of B2 or better from Moody's (or the equivalent), in connection with the transfer of ViaSat's interest in the Satellite to such entity shall so relieve ViaSat. These provisions shall apply to successive assignments of this Agreement. References to a Party following an assignment permitted hereunder shall, following such assignment, refer to such Party's assignee. In the event that Loral assigns its rights and obligations pursuant to this Article 22.1 to Telesat or another non-U.S. entity, ViaSat agrees that it shall use commercially reasonable efforts to obtain all permits, consents, approvals, or licenses that may be required by the United States Federal Communications Commission, Department of State Directorate of Defense Trade Controls, or the Department of Commerce Bureau of Industry and Security, or other relevant United States government or regulatory agency, for the performance of its obligations hereunder to the assignee, and, in particular, to authorize ViaSat to provide such technical data or defense services to the assignee concerning the Satellite and its operations sufficient for ViaSat to satisfy its obligations under this Agreement. It is understood and agreed that ViaSat shall have the right to assign this Agreement as otherwise permitted by this Section 22.1 excluding ViaSat's rights and obligations under Sections 4.2, 4.3 and 4.4, provided that ViaSat shall remain fully responsible for the obligations so excluded from assignment; provided further that in the event ViaSat partially assigns this Agreement in accordance with the foregoing clause, the parties will enter into appropriate, separate agreements reflecting such partial assignment.

22.2. <u>Security Interests</u>. Either Party may assign its rights under this Agreement without the consent of the other Party as security and, subject to Section 13.1, otherwise grant security interests in its rights hereunder to lenders that provide financing to such Party, and the other

Party agrees to consent to such assignment in a form reasonably requested by the first Party and any lenders and/or investors providing financing to ViaSat.

ARTICLE 23

GOVERNING LAW

23.1. <u>Governing Law</u>. This Agreement will be governed by and construed in accordance with the laws of the State of California without giving effect to the choice of law principles therein.

ARTICLE 24

DISPUTE RESOLUTION

24.1. <u>Arbitration</u>. The Parties shall use reasonable efforts to settle all disputes by mutual agreement. Prior to the institution of any arbitration claims, the Parties shall use reasonable efforts to settle all disputes arising out of or in connection with the interpretation, performance, or nonperformance by either Party of its obligations under this Agreement. If a dispute is not settled within thirty (30) days after notification by one Party to the other that a dispute exists, the dispute shall be referred to and finally resolved by arbitration as provided by this Section.

The arbitration shall be administered by the American Arbitration Association (the "AAA"), in accordance with the AAA Arbitration Rules as modified by this Agreement. The cost of arbitration, including the fees and expenses of the arbitrator or arbitrators, shall be shared equally by the Parties unless the award otherwise provides. Each Party shall bear its own attorneys' fees and costs associated with the arbitration. The arbitration award shall be final and binding upon the Parties, not subject to appeal to the extent that said waiver may be validly made, and carried out without delay by them. Judgment on the award may be entered in any court of competent jurisdiction. The arbitration shall take place in Denver, CO and be conducted in the English language. The arbitrators shall determine the matters in dispute in accordance with the laws of California.

Disputes shall be administered by a panel of three (3) arbitrators, one of whom shall be selected by each of the Parties and the third of whom shall be selected jointly by the first two arbitrators. All arbitrators shall: (a) have no relationship to any of the Parties; and (b) be practicing attorneys or judges with experience in international commercial agreements or communications matters. The arbitral panel shall permit the Parties to conduct discovery of each other's books, records and witnesses, including the taking of depositions and the answering of interrogatories.

Notwithstanding the foregoing, a party may obtain preliminary or temporary injunctive relief, including specific performance, at any time from a court of competent jurisdiction where irreparable harm to that party is threatened by another party's acts or omissions; provided, however, that requests for permanent injunctive relief shall be arbitrated pursuant to the process set forth above. Pending final resolution of any dispute (including the informal dispute resolution process and litigation), each party hereunder shall perform all its obligations under this Agreement.

ARTICLE 25

NOTICES

25.1. <u>Notices</u>. All notices regarding technical or operational matters requiring immediate attention will be given by telephone followed within a reasonable period of time by written notification by facsimile or electronic mail transmission and shall be deemed to be received upon actual contact with the specified contact person by telephone or upon confirmation of a facsimile or electronic mail transmission. All other notices and requests will be in writing delivered to the address(es) set forth below and shall be deemed to be received upon actual delivery by a recognized courier service or personal delivery or upon confirmation of a facsimile or electronic mail transmission:

If to Loral, to:

Avi Katz, General Counsel 600 Third Avenue New York, New York 10016

If to ViaSat, to: Keven Lippert Vice President, General Counsel 6155 El Camino Real Carlsbad, CA 92009

Either Party may by notice in accordance with this Section change the person or address to which such notices, requests or other communications are to be given.

ARTICLE 26

MISCELLANEOUS

- 26.1. <u>Headings</u>. The headings in this Agreement and the annexes hereto are inserted for convenience of reference only and shall not constitute a part hereof.
- 26.2. <u>Severability</u>. Any provision of this Agreement which is invalid or unenforceable in any jurisdiction shall be ineffective to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining provisions hereof, and any such invalidity or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 26.3. <u>Expenses</u>. Except as specifically provided otherwise in this Agreement, the Parties hereto shall pay all of their own expenses relating to the transactions contemplated by this Agreement, including, without limitation, the fees and expenses of their respective counsel, accountants and financial advisors.
- 26.4. <u>Amendment</u>. No variation or amendment of this Agreement shall be valid unless it is in writing and signed by or on behalf of both Parties to this Agreement.
- 26.5. <u>Waivers</u>. No failure or delay by any Party in exercising any right or remedy provided by law under or pursuant to this Agreement shall impair such right or remedy or be construed as a waiver or variation of it or preclude its exercise at any subsequent time and no single or partial exercise of any such right or remedy shall preclude any other or further exercise of it or the exercise of any other right or remedy.
- 26.6. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same document.

26.7. <u>Survival of Obligations</u>. The obligations of the Parties under this Agreement which by their nature logically would be expected to survive termination, cancellation, or expiration of this Agreement shall survive termination, cancellation, or expiration of this Agreement for the applicable time period specified in such section or, if no time period is specified, for a reasonable period of time under the circumstances.

VIASAT, INC.

LORAL SPACE & COMMUNICATIONS, INC.

/s/ Keven K. Lippert

By: Keven K. Lippert

By: Janet T. Yeung

By: Janet T. Yeung

Title: Vice President and General Counsel

Title: Vice President and Assistant Secretary

THIS AGREEMENT HAS BEEN SIGNED THIS 11th DAY OF JANUARY, 2008.

SCHEDULE 1

DEFINITIONS

In this Agreement the terms set forth hereinafter shall have the meanings defined in this Article:

- "Action" means any action, suit or proceeding at law or in equity, arbitration, inquiry, investigation or governmental, administrative, regulatory or other proceeding by or before any governmental entity.
- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person. For purposes of the immediately preceding sentence, the term "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise.
- "Agreement" means this agreement (including the Schedules and Annexes hereto), as the same may be amended, modified or supplemented from time to time in accordance with its terms.
 - "Confidentiality Agreement" means the confidentiality agreement entered into by ViaSat and Loral on March 8, 2006 (as amended on April 27, 2007).
 - "Effective Date" shall have the meaning set forth in Article 12 hereof.
- "End of Life" means the earlier of (i) the destruction, total loss or constructive total loss of the Satellite,, and (ii) the date on which the actual orbital maneuver life of the Satellite achieves its de-orbit threshold in accordance with applicable law or prudent and customary industry practice.
 - "Knowledge" means actual knowledge after reasonable inquiry and investigation.

"Launch Agency" means the provider responsible for conducting the launch services for the Satellite pursuant to the Launch Services Agreement.

"<u>Launch Services Agreement</u>" or "<u>LSA</u>" means the contract entered into by ViaSat with a Launch Agency, which contract provides for launch services for the Satellite, as such contract may be amended from time to time in accordance with its terms.

"LIBOR" means the rate of interest per annum, at any relevant time, at which thirty (30) day U.S. dollar deposits are offered at such time in the London interbank market.

"Orbital Slot" means the geostationary orbital slot located at 115°W.L.

"Party" and "Parties" refers to ViaSat and Loral, and their respective permitted successors and assigns hereunder.

"Person" means any individual, partnership, joint venture, trust, corporation, limited liability entity, unincorporated organization or other entity (including a governmental entity).

"Satellite" means the SS/L satellite designated as ViaSat-1 and to be built and delivered by SS/L to ViaSat pursuant to the Satellite Contract.

"SS/L" shall have the meaning set forth in the preamble to this Agreement.

"Subsidiary" of a specified Person means any corporation or other entity of which securities or other ownership interests having ordinary voting power to elect a majority of the Board of Directors or other Persons performing similar functions are directly or indirectly owned by such Person.

"<u>Term</u>" means the term of this Agreement, beginning as of the Effective Date and terminating on the Satellite End of Life date or such other date as mutually agreed by the Parties.

"TT&C" means telemetry, tracking and command.

SECOND AMENDMENT TO SECOND AMENDED AND RESTATED REVOLVING LOAN AGREEMENT

This Amendment to Second Amended and Restated Revolving Loan Agreement (this "Amendment") is entered into as of January 25, 2008, by and between ViaSat, Inc., a Delaware corporation ("Borrower"), each lender from time to time party to the Credit Agreement (as defined below) (collectively, the "Lenders" and individually, a "Lender"), UNION BANK OF CALIFORNIA, N.A., as Administrative Agent (in such capacity, "Administrative Agent") and COMERICA BANK, as Collateral Agent (in such capacity, "Collateral Agent;" collectively with Administrative Agent, the "Agents").

RECITALS

Borrower, Agents and the Lenders are parties to that certain Second Amended and Restated Revolving Loan Agreement dated as of January 31, 2005, as amended from time to time, including by that certain First Amendment to Second Amended and Restated Revolving Loan Agreement dated as of December 22, 2006 (collectively, the "Credit Agreement"). The parties desire to amend the Credit Agreement in accordance with the terms of this Amendment. Unless otherwise defined, all initially capitalized terms in this Amendment shall be as defined in the Credit Agreement.

NOW, THEREFORE, the parties agree as follows:

- 1. The following defined term in Section 1.1 of the Credit Agreement hereby is amended to read as follows:
 - "Revolving Loan Maturity Date" means April 30, 2008.
- 2. No course of dealing on the part of Lenders, Agents or their officers, nor any failure or delay in the exercise of any right by any Agent or any Lender, shall operate as a waiver thereof, and any single or partial exercise of any such right shall not preclude any later exercise of any such right. Agents' or Lenders' failure at any time to require strict performance by Borrower of any provision of any Loan Document shall not affect any right of Lenders or Agents thereafter to demand strict compliance and performance. Any suspension or waiver of a right must be in writing signed by an officer of Administrative Agent, in accordance with the terms of the Credit Agreement.
- 3. The Credit Agreement, as amended hereby, shall be and remain in full force and effect in accordance with its respective terms and hereby is ratified and confirmed in all respects. Except as expressly set forth herein, the execution, delivery, and performance of this Amendment shall not operate as a waiver of, or as an amendment of, any right, power, or remedy of Agents or Lenders under the Credit Agreement, as in effect prior to the date hereof.
- 4. All Representations and Warranties contained in the Credit Agreement or in any other document or documents relating thereto shall survive the execution and delivery of this Amendment. The Borrower is not aware of any events which now constitute, or with the passage of time or the giving of notice, or both, would constitute, an Event of Default under the Credit Agreement.
- 5. As a condition to the effectiveness of this Amendment, Administrative Agent shall have received, in form and substance satisfactory to Administrative Agent, the following:
 - (a) this Amendment, duly executed by Borrower, Collateral Agent and each Lender;
 - (b) Affirmation of Subsidiary Guaranty and Security Agreement, duly executed by Guarantor;

and	(c)	all reasonable attorneys' fees and costs incurred through the date of this Amendment, which may be debited from any of Borrower's accounts;
	(d)	such other documents, and completion of such other matters, as Agent may reasonably deem necessary or appropriate.
6. one ins		Amendment may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute ent.
[Balance of Page Intentionally Left Blank]]		

IN WITNESS WHEREOF, the undersigned have executed this Amendment as of the first date above written.

VIASAT, INC,

/s/ Ronald G. Wangerin

By: Ronald G. Wangerin
Title: Chief Financial Officer

${\bf UNION\ BANK\ OF\ CALIFORNIA,\ N.A.},$ as Administrative

Agent

By: /s/ Douglas S. Lambell

Name: Douglas S. Lambell
Title: Vice President/SCM

COMERICA CALIFORNIA, N.A., as Collateral Agent

By: /s/ Tomas Schmidt

Name: Tomas Schmidt Title: Vice President

[Signature Page to Second Amendment to Second Amended and Restated Revolving Loan Agreement]

[Signatures Continued Next Page]

UNION BANK OF CALIFORNIA, N.A.., as a Lender

By: /s/ Douglas S. Lambell

Name: Douglas S. Lambell
Title: Vice President/SCM

COMERICA BANK, as a Lender

By: /s/ Tomas Schmidt

Name: Tomas Schmidt Title: Vice President

WACHOVIA BANK, N.A.

By: /s/ Robert G. McGill Jr.

Name: ROBERT G. McGILL JR

Title: DIRECTOR

[Signature Page to Second Amendment to Second Amended and Restated Revolving Loan Agreement]

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Mark D. Dankberg, Chief Executive Officer of ViaSat, Inc., certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of ViaSat, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to
 ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those
 entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 6, 2008

/s/ Mark D. Dankberg
Mark D. Dankberg
Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Ronald G. Wangerin, Chief Financial Officer of ViaSat, Inc., certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of ViaSat, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 6, 2008

/s/ Ronald G. Wangerin
Ronald G. Wangerin

Chief Financial Officer

CERTIFICATION OF CHIEF EXECUTIVE OFFICER

Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of ViaSat, Inc. (the "Company") hereby certifies, to such officer's knowledge, that:

- (a) the accompanying Quarterly Report on Form 10-Q of the Company for the quarterly period ended December 28, 2007 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
 - (b) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: February 6, 2008

/s/ Mark D. Dankberg
Mark D. Dankberg
Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER

Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of ViaSat, Inc. (the "Company") hereby certifies, to such officer's knowledge, that:

- (a) the accompanying Quarterly Report on Form 10-Q of the Company for the quarterly period ended December 28, 2007 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
 - (b) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: February 6, 2008

/s/ Ronald G. Wangerin Ronald G. Wangerin Chief Financial Officer